GENDER

# Fathers' household and childcare involvement in New Zealand:

A snapshot, determinants and consequences





# AUTHORS

Juliane Hennecke (PI), Lisa Meehan, Gail Pacheco and Alexandra Turcu<sup>1</sup>

# ACKNOWLEDGEMENTS

We acknowledge the children and the families who are part of the *Growing Up in New Zealand* (GUINZ) study. We thank members of the GUINZ research team for their invaluable work interviewing participants and managing the data used in this analysis. We also extend our gratitude to our external reviewers Deb Potter, Dr. Suzy Morrissey, Maanaima Soa-Lafoai and an anonymous reviewer from Te Puni Kōkiri for providing valuable feedback on draft findings and our analysis. This report was made possible with funding from the Ministry of Social Development, using GUINZ data collected by the University of Auckland, and in accordance with the GUINZ Data Access Protocol.

## DISCLAIMER

The views and interpretations in this report are those of the researchers and not the Ministry of Social Development.

# PUBLISHED

New Zealand Work Research Institute, Auckland, New Zealand

ISBN (PDF): 978-1-927184-96-7

2022

Suggested citation: Hennecke, J., Meehan, L., Pacheco, G. & Turcu, A. (2022) *Fathers' household and childcare involvement in New Zealand: A snapshot, determinants and consequences*. NZ Work Research Institute. Auckland, NZ.

<sup>&</sup>lt;sup>1</sup> Juliane Hennecke – Otto von Guericke University Magdeburg, Germany and NZ Work Research Institute, AUT, New Zealand; Lisa Meehan, Gail Pacheco and Alexandra Turcu – NZ Work Research Institute, Faculty of Business, Economics and Law, AUT, New Zealand.`

# TABLE OF CONTENTS

1.	INTF	RODU	ICTION	1
2.	INTE	RNA	TIONAL LITERATURE REVIEW	3
2	.1	Earr	narked father's leave	3
2	.2	Driv	ers of paternal involvement and the gender care gap	4
	2.2.2	1	Financial and economic considerations	4
	2.2.2	2	Social norms and gender identity	5
	2.2.3	3	Biological and psychological factors	5
3.	GEN	DER	CARE GAP IN THE NZ CONTEXT	6
3	.1	Polic	cy background in NZ	6
	3.1.2	1	Parental leave in NZ	6
	3.1.2	2	Other policy influences: Childcare, welfare and tax	7
3	.2	NZ g	gender care gap in the international context	8
4.	DAT	Α		11
4	.1	Grov	wing Up in New Zealand	11
4	.2	Sam	ple construction	12
4	.3	Desc	criptive statistics	L3
	4.3.2	1	Comparison of fathers who did and did not participate in GUiNZ	L3
	4.3.2	2	Final sample descriptive statistics	L3
5.	RES	EARC	H AIM 1	16
5	.1	Self-	reported childcare and housework involvement	16
5	.2	Qua	lity of care	18
5	.3	Pare	ental leave taking	19
	5.3.2	1	How representative is leave taking of paternal involvement?	22
6.	RES	EARC	H AIM 2	24
6	.1	Vari	ables and empirical approach	24
6	.2	Resu	ılts	25
	6.2.2	1	Participation in GUiNZ	27
	6.2.2	2	Leave taking	28
	6.2.3	3	Childcare involvement	29
	6.2.4	1	Quality of care	30
	6.2.5	5	Psycho-social characteristics as mechanisms	30
	6.2.6	ĵ	Heterogeneity analysis by ethnicity	31
7.	RES	EARC	H AIM 3	33

-	7.1	Variables and empirical approach	33
-	7.2	Results	36
8.	LIMI	TATIONS	39
9.	CON	ICLUSIONS	40
REI	EREN	CES	42
AP	PENDI	x	47

# LIST OF TABLES

Table 1 - Descriptive statistics of quality of care	. 19
Table 2 - Reason for no / shorter leave	. 21
Table 3 - Types of paternal leave	. 21
Table 4 - Direct involvement measures by leave taking	. 22
Table 5 - Determinants of paternal involvement (marginal effects of logit estimation)	. 25
Table 6 - Children's outcome variables	. 34
Table 7 - Estimation results – outcome: child's development	. 37

# LIST OF FIGURES

Figure 1 - Number of weeks of paid maternity and parental leave in OECD countries, 2009 and 2018	37
Figure 2 - Net childcare costs	8
Figure 3 - Time spent on unpaid work (minutes per day) in OECD countries, by gender	9
Figure 4 - Gender differences in paid, unpaid and total work – minutes per day	. 10
Figure 5 - Sample restriction steps (6,853 children)	. 12
Figure 6 - Summary statistics of households (final sample, N=3,098)	. 13
Figure 7 - Summary statistics of mothers and fathers (final sample, N=3,098)	. 14
Figure 8 - Descriptive statistics of childcare involvement	. 17
Figure 9 - Fair shares of childcare and housework	. 18
Figure 10 - Paternal leave taking of employed fathers	. 20
Figure 11 - Distribution of children's developmental outcomes by sample	. 35

# LIST OF APPENDICES

Table A.1 - Descriptive statistics by father availability	47
Table A.2 - Variable overview and summary statistics - children, household and mothers (	(all
observations)	49
Table A.3 - Variable overview and summary statistics - fathers (sample: permanent fathers)	51
Table A.4 - Full estimation results - outcome: GUiNZ participation	54
Table A.5 - Full estimation results - outcome: involvement in day-to-day care	56

Table A.6 - Full estimation results - outcome: leave taking	60
Table A.7 - Full estimation results - outcome: quality of care	64
Table A.8 - Heterogeneity analysis by ethnicity - outcome: high involvement (9m)	68
Table A.9 – Heterogeneity analysis by ethnicity - outcome: quality of care	71
Table A.10 - Heterogeneity analysis by ethnicity - outcome: leave taking	74
Table A.11 – Items list for CBQ variables	77
Table A.12 – Items list for SDQ variables	78

# EXECUTIVE SUMMARY

This report focused on three main research aims using data from the Growing Up in New Zealand (GUINZ) birth cohort:

- 1. To provide a snapshot of fathers' engagement during the early years of their children's lives.
- 2. To analyse the determinants of fathers' engagement.
- 3. To analyse potential consequences of different levels of fathers' engagement on children's outcomes.

The current international literature focuses on the difference between mothers' and fathers' involvement in domestic duties, the so-called 'gender care gap'. The gap has potentially significant consequences for gender equality, as it is likely an underlying driver of gender differences in labour force participation, career advancement, occupational choice and wage rates. Indeed, both internationally and in New Zealand (NZ), these gender gaps in labour market outcomes are higher among parents than non-parents, highlighting the potentially important contribution of the gender care gap to the 'motherhood penalty'. In the NZ context, evidence on this gap is sparse. While parental leave entitlements sit with the mother in NZ, and although she can transfer all or part of her entitlements to her spouse or de facto partner, uptake of paid parental leave (PPL) by fathers is less than 1%. This low uptake of PPL by fathers could indicate the presence of a substantial gender care gap in NZ. However, while many existing studies use paternity leave as a measure of fathers' involvement in childcare, GUINZ also provides more direct measures of involvement. Indeed, comparing these more direct measures available in GUINZ with the amount of paternity leave taken highlights that paternity leave is not a particularly good measure of how involved fathers are.

This also highlights that GUINZ has the distinct advantage of including information that allows us to draw a very detailed picture of the involvement of NZ fathers in their children's upbringing in the first years after birth. The data allow us to measure fathers' involvement in a number of ways. This provides an advantage over many previous studies which often use the amount of paternity leave taken as a proxy for fathers' involvement. Using these data, we first compared indicators of direct involvement relative to the mothers in the sample. We get an overview of different levels and forms of engagement observed with respect to not only parental leave taking, but also direct involvement in day-to-day care, and activity indicators that proxy for quality of care (such as frequency of playing games and reading books). We then analyse the determinants of differing levels of paternal involvement. A novel contribution of this research is to delve into the psycho-social characteristics and personality traits as mechanisms behind NZ fathers' involvement with their children. In the final research aim of this study we undertook exploratory analysis of the association between paternal involvement in the child's early years and children's cognitive, physical and psychological development at later time points. The key findings are as follows:

### Aim 1

- Most fathers take less parental leave than they would like to. Before their child is born, more than half of employed fathers anticipate that they will take less leave than they would prefer to, and the amount they actually end up taking is even less than that anticipated. In addition, before their child is born, fathers overestimate how much they will be involved in their child's care.
- Fathers are distinctly less involved in their child's care than mothers, although their involvement does increase as the child grows from an infant to a toddler. They also spend less time engaged in quality-care activities (i.e. enriching activities such as playing games, reading books etc.).
- It is often assumed that fathers who take more paternity leave are more involved in childcare. However, we do not find evidence to support this. Fathers who take more leave are not more involved in childcare nor do they provide a higher quality of care.

### Aim 2

- Unsurprisingly, fathers with greater work commitments (such as self-employment, full-time employment and overtime work) are less involved in the day-to-day care of their child.
- Similarly, fathers who earn much more than the child's mother are less likely to be involved in childcare. This probably reflects that it makes financial sense for these families to have a higher degree of specialisation, with the father focusing on paid work and the mother focusing on unpaid work, including childcare.
- However, a greater share of fathers (71%) than mothers (56%) perceive this distribution of childcare responsibilities to be fair.
- A range of factors is associated with the amount of time spent engaged in quality childcare activities, such as reading books and playing games. For example, fathers spend less time on these activities if the child has siblings, likely reflecting greater time constraints. Fathers who have more helpful families and good relationships with the child's mother spend more time on these activities.
- In terms of personality traits, conscientious fathers spend more time on their child's day-to-day care, while extraverted and open fathers are more likely to engage in high quality care activities.
- There are clear ethnic differences in the quantity and quality of fathers' involvement, although these should be interpreted with caution due to the small sample size for non-European fathers. Relative to NZ European fathers, Māori and Pasifika fathers are more involved in their child's care and Pasifika fathers are more likely to undertake high quality care activities.

### Aim 3

- There is no clear association between the amount of paternal leave a father takes and their child's developmental outcomes.
- On the other hand, there is a positive relationship between fathers' involvement in day-to-day care and psychological outcomes.
- Higher quality of care is positively associated with language development, motor skills and most psychological outcomes. Thus, both the quantity and quality of care appear to be important for developmental outcomes.

# 1. INTRODUCTION

There is a growing national and international interest among academics and policymakers in paternal family engagement and division of domestic duties in households. Numerous international studies have focused on the difference between mothers' and fathers' involvement in domestic duties, the so-called 'gender care gap'. This gap is evident in nearly all developed countries and persists despite increases in female labour force participation. The gap has potentially significant consequences for gender equality, as it is likely an underlying driver of gender differences in labour force participation, career advancement, occupational choice and wage rates. Furthermore, it could have important consequences for children's development and wellbeing. The New Zealand (NZ) evidence on this front is sparse. This project will contribute to the limited knowledge in this space by delivering an empirical analysis of the involvement of fathers in their children's upbringing and other domestic duties in NZ.

This research uses a contemporary birth cohort dataset from the Growing Up in New Zealand (GUiNZ) survey, which tracks both child and household information from the antenatal period through the early childhood years. It is important to note that in the context of GUiNZ, 'fathers' are defined as being the second parental figure next to the mother, and includes stepfathers, co-mothers, foster and adoptive parents as well as other family members who play a father-figure role. However, the vast majority (97%) are the biological fathers. A clear advantage of this dataset is that in addition to collecting data about the child and mother, the father is also regularly surveyed– this is imperative for our research aims, which are three-fold and detailed next.

### Aim 1 - To provide a snapshot of fathers' engagement during the early years of their children's lives.

This is done by using information on parental leave taking, self-reported day-to-day involvement, hours spent on housework as well as the frequency of child interaction activities such as playing and reading books. This overview is complemented by a comparison with mothers' engagement, as well as a comparison between actual paternal involvement and anticipated antenatal involvement.

### Aim 2 – To analyse the determinants of fathers' engagement.

The second research aim analyses both the external and internal determinants of different levels of paternal involvement. This is done by estimating conditional associations between reported engagement and individual characteristics such as labour market status, education, relationship to the child and the mother, availability/use of external help, as well as inherent norms, values and traits such as parental identity and psychological traits.

# Aim 3 – To analyse potential consequences of different levels of fathers' engagement on children's outcomes.

In the third research aim, we focus on a range of measures of cognitive and non-cognitive development of children surveyed in GUINZ. While this research is exploratory in nature, it provides insights into potential impacts of varying levels of paternal engagement during the early years of a child's life.

The remainder of this paper is organised as follows: Section 2 provides a brief scan of the relevant international literature (particularly with respect to the first two research aims); Section 3 describes the current context in NZ; Section 4 describes the data and variables used in this analysis; Sections 5, 6 and 7 present the identification strategy and results for each research aim consecutively; Section 8 describes

the limitations and caveats that need to be acknowledged; while Section 9 provides an overall conclusion including key policy insights and direction for future research.

# 2. INTERNATIONAL LITERATURE REVIEW

As indicated earlier, there is growing interest in paternal family engagement and division of domestic duties in households. Much of the existing international literature focuses on the difference between mothers' and fathers' involvement in domestic duties, the so-called 'gender care gap'. This gap is evident in nearly all developed countries. Women, and specifically mothers, still perform a significantly higher share of unpaid domestic work, irrespective of their labour force status (Bianchi 2000; Cunningham 2007; Hook 2010; Samtleben 2019; Sanchez and Thomas 1997). This gap is a potentially important determinant of observed differences in the economic outcomes of men and women, including persistent differences in labour force participation, career advancement, occupational choice and wage rates (Blau and Kahn 2007; 2017; Bütikofer, Jensen, and Salvanes 2018; Bertrand, Goldin, and Katz 2010). Both internationally and in NZ, these gender gaps in labour market outcomes are higher among parents than non-parents, highlighting the role of the gender care gap in this observed 'motherhood penalty' (for example, Anderson, Binder, and Krause 2002; Budig and England 2001; Dixon 2000; Gangl and Ziefle 2009; Gough and Noonan 2013; Stats NZ 2017; Wilner 2016; Sin, Dasgupta, and Pacheco 2018).

## 2.1 Earmarked father's leave

Internationally, much of the debate and literature on the gender division of domestic duties has been dominated by consideration of 'daddy months' in national parental leave schemes such as in the Nordic countries (Sweden, Norway, Denmark, Iceland and Finland) as well as in Germany and Canada. 'Daddy months' can broadly be described as schemes where some paid parental leave is exclusively reserved for fathers. Such policies are aimed at increasing paternal leave and thereby fathers' involvement with their children in their early years. These policies are politically motivated by the idea that the active involvement of fathers benefits child development and promotes gender equality, in addition to the notion that a more equal household division of labour could boost fertility (Haas and Rostgaard 2011).

Empirical evidence on the effect of daddy months on fathers' leave taking and childcare involvement are very mixed. Most studies do find that the introduction of daddy months increases fathers' leave taking (Ekberg, Eriksson, and Friebel 2013; Patnaik 2019; Cools, Fiva, and Kirkebøen 2015; Bartel et al. 2018). Moreover, many studies find that the reforms boost fathers' involvement in childcare and promote a more even household division of labour (Tanaka and Waldfogel 2007; Tamm 2019; Kotsadam and Finseraas 2011) as well as mothers' relative income and labour force attachment (Druedahl, Ejrnæs, and Jørgensen 2019; Farré and González 2019). Additionally, in line with the early motivation of these policies, the increase of fathers' involvement due to the introduction of earmarked leave was also observed to have a positive effect on children's development as, for example, measured by school performance (Cools, Fiva, and Kirkebøen 2015; El Nokali, Bachman, and Votruba-Drzal 2010; Mangiavacchi, Piccoli, and Pieroni 2021).

On the other hand, the international literature is still skeptical about whether daddy months are the right way to promote a more equal division of childcare. This is driven by the observation in most countries that fathers' parental leave take-up is still largely restricted to the earmarked time and only very rarely exceeds one or two months (Eriksson 2005 for Sweden; Samtleben, Schaeper, and Wrohlich 2019 for Germany). Additionally, many studies also find that the gender care gap and the persistent traditional allocation of parent's labour supply remained largely unaffected by these reforms (Cools,

Fiva, and Kirkebøen 2015; Ekberg, Eriksson, and Friebel 2013). Although daddy months seemed to have contributed positively towards a more equal division of domestic duties as well as the household allocation of labour supply by reducing the financial constraints of fathers who are willing to take leave, other factors and drivers of paternal and maternal involvement still seem to overshadow the parental decision making.

Additionally, the parental leave literature largely suffers from severe selection issues due to the low take-up rates of parental leave by fathers as well as the selectivity of take-up, especially with respect to education and workplace characteristics. This is shown in Geisler and Kreyenfeld (2019), where highly educated fathers as well as fathers with permanent contracts and fathers working in the public sector are those taking parental leave.

## 2.2 Drivers of paternal involvement and the gender care gap

The existing knowledge on drivers of the gender care gap can be categorised into three main groups: 1) financial and economic considerations, 2) social norms and gender identity, and 3) biological reasons.

### 2.2.1 Financial and economic considerations

The economic literature has for a long time concentrated on economic considerations as the main driver of the division of labour within households. This stream of research is largely based on the theoretical framework developed by Gary Becker's model of intra-household specialisation (Becker 1981). Becker's theory is that household members will always specialise according to their comparative advantages determined by relative human capital levels. The relative wage income should thus determine the choice about leave taking, predicting that the secondary earner will take over the larger share of domestic duties. International statistics do show that large parts of household decision making can potentially still be explained using this theory, some 40 years after Becker's seminal work (Heim 2007; Juhn and Murphy 1997; Blau and Kahn 2017; 2007). As has been prominently shown by Bertrand et al. (2015) and is discussed in detail in section 3.2, female breadwinning remains a rather rare phenomenon with married women having substantially lower employment rates, higher part-time employment rates and lower average earnings than their partners.

Due to low replacement rates of parental leave payments as well as low maximum payment thresholds, high-income fathers in particular face strong financial disincentives when considering taking parental leave (Sigurdardottir and Garðarsdóttir 2018). This is also supported by the research of Hennecke and Pape (2021), who find that after losing their jobs fathers increase their time investment with children and housework, and the outsourcing of tasks (e.g. external childcare facilities) decreases. Nevertheless, the authors also found that the increased time investment is only observed for the time of unemployment and time spent with children or housework decreases below the original level after re-employment.

There is also an important role played by workplace characteristics. This is supported by findings from economic, sociological and management literature about the importance of the role of company policies, workplace culture and workplace support among employers and coworkers (Birkett and Forbes 2019; Samtleben et al. 2019; Brandth and Kvande 2019; Bygren and Duvander 2006; T. D. Allen 2012).

### 2.2.2 Social norms and gender identity

Existing research on workplace support among employers and coworkers shows that it is often inextricably linked with the prevailing social norms within a firm. Researchers such as Dahl, Løken and Mogstad (2014) have found very strong peer effects in the take-up rates of parental leave among male coworkers in addition to those observed in the family network.

The dominance of traditional gender role norms has often been named the most important cause of the gender care gap in the sociological literature in recent years. An important seminal work in this respect was the study by Bertrand, Kamenica and Pan (2015) who find that women who out-earn their partners are more likely to spend more time on household chores. The authors take this as evidence in line with the identity concepts by Akerlof and Kranton (2000) where deviating from the behavioural prescription for the category you belong to (whether men or women) is costly.

### 2.2.3 Biological and psychological factors

One of the biggest challenges of empirical research which attempts to identify the role of social norms and gender identity to understand gender gaps in the division of childcare and housework is the difficulty of separating these social preferences from individual preferences. The gender care gap could also be driven by individual preferences if mothers have different preferences for childcare responsibilities than fathers purely because mothers are the birthing parent and/or because of the biological sex differences between them.

An often-given explanation from opponents to the social identity theory is that the hormone levels of mothers lead to them having a stronger attachment to their newborn babies (Barrett and Fleming 2011). The so-called phenomenon of maternal gatekeeping leads to mothers wanting to take over the major part of the childcare responsibilities and having a hard time granting responsibilities to their partners (Allen and Hawkins 1999). Furthermore, the biological aspects around giving birth, such as puerperium and breastfeeding, are factors which cannot be transferred between birthing and non-birthing parents, making it harder for fathers to bond with their children (Brady et al. 2017). The literature has begun to delve further into this space by comparing adoptive and biological parents, which enables them to compare birthing and non-birthing mothers with respect to their parental leave take-up (Moberg and Van der Vleuten 2021).

# 3. GENDER CARE GAP IN THE NZ CONTEXT

This section outlines the context in NZ and how this compares with the international situation. It first discusses parental leave policies and briefly describes additional relevant policies with respect to childcare, welfare and tax. This provides background for the subsequent discussion on labour market outcomes and time spent in paid versus unpaid work of men and women in NZ, and where information is available, of mothers and fathers.

# 3.1 Policy background in NZ

### 3.1.1 Parental leave in NZ

The provision of unpaid and government paid parental leave (PPL) is set out in the Parental Leave and Employment Protection Act 1987 and its subsequent amendments. At the time GUINZ children were born in 2009-2010, eligible mothers were entitled to up to 12 months of unpaid job-protection leave and 14 weeks of government-funded PPL. Mothers must have been employed with the same employer or self-employed for the preceding six or 12 months and have worked for an average of 10 hours or more a week over this period to qualify for PPL, and 12 months to qualify for up to 12 months of unpaid leave. Since then, the length of PPL has been gradually extended, with the most recent increase to 26 weeks effective from July 2020. In addition, the eligibility criteria have been reduced over time. For example, in 2016, the requirement for employment to have been with the same employer was removed to extend coverage to mothers with less stable employment histories. The payment amount for PPL has not changed in relative terms since paid parental leave was introduced in 2002. It is set at the maximum of the mother's pre-parental leave pay or an amount approximately equivalent to the full-time minimum wage (\$429.74 per week in July 2009).

While parental leave entitlements sit with the mother,<sup>2</sup> she can transfer all or part of her entitlements to her spouse or de facto partner. However, this occurs very infrequently, with the uptake of PPL by fathers being less than 1% (Morrissey 2020). Under the Act, NZ fathers who have been employed continuously for 12 months are entitled to two weeks' unpaid partner leave upon the birth of a child. However, it appears that uptake of this is also very low, with only about 4% of fathers taking this unpaid leave (Reilly and Morrissey 2017).

OECD data shows that in 2009, around the time the GUINZ children were born, the length of NZ's PPL was one of the shortest in the OECD. By 2018, the latest year of data available, the situation has changed little with respect to NZ's ranking in this statistic. Not shown in Figure 1 is the fact that NZ does not have a fathers' quota for PPL. While this situation was not unusual amongst OECD countries in 2009, an increasing number of countries have introduced earmarked paternal PPL quotas and by 2018, NZ was one of just six OECD countries with PPL that does not have dedicated paid paternal leave.

<sup>&</sup>lt;sup>2</sup> More gender-neutral language was adopted as part of the 2016 amendment to this Act, by replacing the term 'mother' with 'primary caregiver'. However, the entitlement to parental leave still sits with the mother since the Act defines the primary caregiver as "a female (the biological mother)", who is entitled to transfer her entitlement to her spouse or de facto partner, who then becomes the primary caregiver for the purpose of the Act.





Source: Own calculations based on data from OECD (2021).

The payment level is also low in NZ. Data from the OECD Family Database (2021) show that, on average across OECD economies, maternity leave benefits replace approximately 77% of previous earnings for a mother with average full-time earnings; whereas NZ replaces less than 50% of gross earnings. Note that this is similar to several other anglophone countries – such as Australia, Canada, Ireland and the United Kingdom. Furthermore, NZ also does not differentiate between PPL (home care leave) available to the mother and maternity leave (pregnancy leave), i.e. leave covering the weeks immediately before and after birth. The latter has higher payment rates (up to 100% in countries like Germany, Spain and many South American countries) and is often covered by the health insurance system (OECD Family Database 2021).

### 3.1.2 Other policy influences: Childcare, welfare and tax

While parental leave generally and dedicated paternal leave specifically are important policy considerations, it is only one of a set of factors that could influence how parents choose to divide their time between work within and outside the home. Indeed, there is a wide range of policy settings that can influence fathers' involvement, not to mention a wider set of non-policy factors such as cultural norms.

One policy setting that is particularly relevant in the NZ context is childcare costs. Lack of affordable, quality childcare could result in more household specialisation and mean that more mothers take on the primary caregiving role while more fathers specialise in paid employment and have less involvement in child rearing. NZ has one of the highest out-of-pocket childcare costs in the OECD (Figure 2) and the high effective marginal tax rate of returning to work that high childcare costs imply, may reinforce traditional gender specialisation within households.

#### Figure 2 - Net childcare costs



Source: Own calculations based on 2018 data from OECD (2021).

Welfare and tax settings also influence household division of labour and choices about whether to work inside or outside the home. It is beyond the scope of this report to discuss these in any detail. However, one example of this is evidence that changes to in-work incentives and financial support for families with dependent children implemented between 2004 and 2007 increased the labour force participation and hours of solo parents but reduced participation and hours of married men and women, with particularly large reductions among married women with children (Mercante and Mok 2014).

### 3.2 NZ gender care gap in the international context

As mentioned, few NZ fathers use their statutory entitlement of two weeks' unpaid leave, and it is rare that part or all of their partner's PPL entitlement is transferred to them. On average, NZ mothers take about the same length of time off work as the job-protection period of one year, which is much longer than the PPL period. Using GUINZ data, Noy & Sin (2021) find that mothers take 53 weeks of leave on average. Mothers who were working antenatally would prefer to take 69 weeks of parental leave, with 70% taking less leave than they would have liked.

Among other factors, differences in parenting responsibilities may be due to, as well as potentially contribute to, gender differences in employment and earnings. The employment rate in 2020 among NZ men was 72.8% versus 62.8% for women. Additionally, and similar to other developed countries, gender gaps in labour market outcomes are higher among parents than non-parents (Dixon 2000; Stats NZ 2017). Recent estimates of the so-called 'motherhood penalty' for NZ in Sin et al. (2018) confirms that parenthood increases employment gaps between men and women. This reflects that women across the income distribution are less likely to be employed after becoming parents, although women with higher antenatal earnings return to employment more quickly. In contrast, men do not decrease their employment after parenthood regardless of prior earnings. Mothers also reduce their hours worked, with almost all of the greater propensity to work part-time among women being driven by mothers being more likely to work part-time than fathers. Moreover, Sin et al. (2018) find that the average woman earns 4.4% lower hourly wages as a parent than if she hadn't had children, whereas there is no significant effect on fathers' pay.

These gender differences are also apparent in the time spent in unpaid work by men and women. NZ is in the mid-range among OECD countries, with a difference between men and women of 123 minutes per day, which is close to the OECD average (126.6 minutes/day) (Figure 3). Scandinavian countries and Canada are among the countries with the lowest differences (between 50 and 75 minutes/day) and countries like Mexico, Portugal and Turkey have the highest gaps (up to 238 minutes/day).





Source: Own presentation based on data from OECD (2021).

Figure 4 depicts the gender difference in unpaid work (blue bars), paid work (orange bars), and the cumulative difference in total minutes per day spent in both paid and unpaid work (grey bars). NZ is among only five OECD countries with a negative gender difference in total work. Kiwi men, on average, spend 10 more minutes per day engaged in any form of work (paid or unpaid) than women.



Figure 4 - Gender differences in paid, unpaid and total work - minutes per day

Source: Own calculations and presentation based on data from OECD (2021).

In NZ, this negative gender work gap is driven by a relatively large difference in the time spent in paid work (133 minutes/day), which is higher in only six other OECD countries, including those with more traditional gender norms such as Turkey, Mexico and Portugal. This pattern potentially signals an interesting dynamic behind the observed gender care gap in NZ, in which the unequal labour market background (opportunities and constraints) of men and women plays a more important role in explaining the unequal distribution of domestic duties than underlying intra-household gender norms which have been observed to be the main driver in many other OECD countries. This is potentially driven, at least in part, by policy settings such as NZ's high childcare costs (discussed above).

This suggests that NZ has some unique characteristics that may limit the applicability of overseas research in this area. Despite this, NZ literature on fathers' involvement in domestic duties is scant. One relevant study from the Families Commission, based on a representative survey of fathers, provides a general overview of the involvement of fathers as well as paternalistic styles and role models (Luketina, Davidson, and Palmer 2009). In line with the OECD data, it finds that work commitments and time pressure are the most common barrier to fathers' involvement in their children's care among Kiwi fathers. Callister (2005) also argues that there is a 'double burden' of paid and unpaid work carried by fathers in NZ.

# 4. DATA

## 4.1 Growing Up in New Zealand

The data used for our analysis is obtained from the Growing Up in NZ survey (GUINZ).<sup>3</sup> This is NZ's largest contemporary longitudinal study of child development, funded by the Ministry of Social Development and conducted by a team at the University of Auckland (Morton et al. 2010; 2013). It is a longitudinal survey of more than 6,000 children born in the Auckland, Waikato, and Counties-Manukau regions in 2009-2010 and their families. Pregnant women were recruited based on their expected delivery date (April 2009 – April 2010) such that they are roughly representative of all families in the NZ population at that time in terms of ethnic diversity and socio-economic status (Morton et al. 2010; 2013).<sup>4</sup>

The survey follows the children and their families from the time of pregnancy onwards. This has resulted in interviews at a total of 14 points in time in the past 11 years with 8 major data collection waves (DCW's): antenatal (DCW0), 9 months (DCW1), 2 years (DCW2), 31-months (DCW3), 45-months (DCW4), 54-months (DCW5), 6 years (DCW6) and 8 years (DCW8) (Morton et al. 2020). A key focus of the survey is always the child. For instance, the researchers follow the child when the household attachment changes. In most cases the mother is the primary respondent especially in the first years, in which the children were not able to respond to the interviewers' questions themselves. Mothers thus answer their own questionnaire as well as in DCW2, a proxy questionnaire for their child. In data collection waves 0, 1 and 2, the partner of the mother answers an additional questionnaire. According to Growing Up in New Zealand (2017), 97% of these partners are the biological fathers of the child and 93% live in the same household. It nevertheless has to be noted that these partners, who we will label 'fathers' in the following, include co-mothers (due to a lack of observed gender for the mothers' partners), stepfathers, foster and adoptive parents as well as other family members who have a father role. In line with Growing Up in New Zealand (2017), the focus is less on the gender aspect of the paternal role and more on the availability of a family member who takes on a fathering role. We define 'fathers' as being the second parental figure in the household next to the mother. The possibility of linking fathers to their children is one of the distinct features of GUINZ. Longitudinal studies of children have typically not included fathers in the past (Pryor et al, 2014).

The questionnaire answered by the fathers was equivalent to the mothers' questionnaire, in which they give personal information about themselves, their relationship with the child and the mother of the child as well as answer questions on the child. Thus, for much information on the children in the early DCWs, two versions (one from the mother and one from the father) are available. If not indicated otherwise, we will always draw on the child's information given by the mother in order to 1) avoid missing information for families in which no partner is observed and 2) reduce endogeneity of the information with respect to the father's involvement.

It is important to acknowledge that the GUiNZ survey, and therefore our analysis, uses the nuclear family as the unit of analysis. However, other family structures and modes of child-rearing are

<sup>&</sup>lt;sup>3</sup> GUiNZ questionnaires and data dictionaries can be found on survey website https://www.growingup.co.nz/available-data-2 (accessed 1 April 2022).

<sup>&</sup>lt;sup>4</sup> While Asian and Pasifika families are slightly oversampled in the survey, Māori mothers are undersampled overall (Morton et al. 2013).

important, particularly in non-western countries and cultures. Within NZ, for example, communal and extended family approaches are more commonplace among non-European families, including Māori, Pasifika and Asian communities. While it is beyond the scope of this report to examine these approaches, it would be interesting for future research to explore the contribution of these and how they could be supported in NZ.

## 4.2 Sample construction

We construct our estimation sample in a way which ensures the use of a homogeneous sample throughout the later analysis steps. The stepwise reduction of the sample is documented in Figure 5.





*Source:* GUINZ 2020, own calculations and illustrations.

We start with the full GUINZ dataset of 6,853 children included in the database. First, we drop 714 children (and their mothers) from the sample who were not observed in at least one of the first three observation waves (DCW0, DCW1 or DCW2). Secondly, we drop all families with missing information on key demographics (including the father's demographics in the cases where fathers are observed), which reduces the sample by around a quarter (1,604 observations). Thirdly, we drop 92 single mothers as defined by their partnership status, i.e. mothers who indicate that they do not have a partner.<sup>5</sup>

For the remaining 4,443 children in the full sample, roughly three-quarters (3,369) of the partners in the same household agreed to participate in GUINZ and were thus surveyed in the antenatal interview. In a final data-cleaning step, we drop 271 children whose fathers do not participate in GUINZ at a later interview (either DCW1 or DCW2), leaving us with a final sample of 3,098 children with the same father-

<sup>&</sup>lt;sup>5</sup> See Pryor et al. (2014) for a detailed overview of the characteristics of the families in the unrestricted full GUiNZ sample. The share of single mothers in our sample is already reduced since we have already dropped observations with missing information. Pryor et al. (2014) report that "just over 5% of pregnant women stated that they were not currently in any relationship" (p.7).

figure in their first two years of life who is permanently present in the GUINZ data.<sup>6</sup> The resulting sample is utilised for research aims 1 and  $2.^{7}$ 

## 4.3 Descriptive statistics

### 4.3.1 Comparison of fathers who did and did not participate in GUINZ

We first compare a range of characteristics of families in which the partners did versus did not participate in GUiNZ in the first three DCWs. These descriptions, as well as the t-test results of whether means between the samples are significantly different from each other, are provided in Table A.1 in the Appendix. We find that lower household income and unplanned pregnancy are more prominent in households where the father did not participate in GUiNZ. Furthermore, there is overrepresentation of younger mothers and mothers who are not employed in this group. These differences result in household incomes and mother's education being higher in the final sample (as compared to the full sample of families) due to the need to drop observations where the father did not participate in GUiNZ.

### 4.3.2 Final sample descriptive statistics

Figures 5 and 6 give an overview of some key statistics of the households, mothers, and fathers in our final sample. The full set of descriptive statistics can be found in Table A.2 (for mothers and households) and Table A. 3 (for fathers) in the Appendix.



### Figure 6 - Summary statistics of households (final sample, N=3,098)

Source: GUINZ DCW0 (Household income, HH type) and DCW2 (external childcare) as well as the 16-month interview (siblings), own calculations.

<sup>&</sup>lt;sup>6</sup> Approximately 42% of fathers who drop out between the antenatal phase and the 2-year interview do so because of a separation with the mother (i.e. the mother indicates not having a partner in DCW2).

<sup>&</sup>lt;sup>7</sup> As research aim 3 includes the child's outcomes at a later stage (most at 8 years), we reduce the sample further for this final aim (details provided later).

Figure 6 shows that 40% of the children in our sample have no other siblings, 37% have one and 23% have more than one older sibling at the time of birth. In 81% of the households the parents live alone while 19% of households consist of the parents and other adult persons such as the extended family. At the age of two, 41% of the GUINZ children are not in any external care while 38% are in a type of early childhood education (ECE) facility and another 7% are in a home-based care facility. The other 14% are cared for in the private space, for example by grandparents, nannies or other relatives and friends. As has been discussed above, the average household income of the families in our sample is relatively high with 71% of families having NZ\$70,000 or more per year and only 4% have less than NZ\$30,000.





*Source:* GUINZ DCW0, own calculations and illustrations.

Figure 7 indicates that the majority of both mothers and fathers are in the prime child-bearing age with 61% of mothers and 52% of fathers aged between 25 and 34 years. More than two-thirds of parents

were born in NZ (69% of fathers and 68% of mothers). In terms of their prioritised ethnicity,<sup>8</sup> 65% (66%) of fathers (mothers) are NZ European, 13% (12%) Māori, 8% (7%) Pacific Peoples, 11% (13%) Asian and another 2% (3%) MELAA or any other ethnicity. Figure 7 also reveals that mothers are on average better educated than fathers in the sample, with 53% of mothers but only 40% of fathers having a bachelors or higher degree. We also find that 84% of fathers are employed during the antenatal interview while only 66% of mothers are, and the share of mothers who are not in the labor force is much higher (23% as opposed to only 1% for fathers).

<sup>&</sup>lt;sup>8</sup> The prioritised ethnicity assigns one ethnicity to a person if multiple ethnicities are reported by prioritising Māori, Pacific Peoples, Asian, MELAA/ Others and NZ European in this order. MELAA is Middle Eastern, Latin American or African.

# 5. RESEARCH AIM 1

The first research aim is to provide a snapshot of fathers' engagement during the early years of their children's lives. This includes a comparison between fathers' actual engagement and anticipated engagement prior to birth, relative to mothers' engagement. We will do so by looking into self-reported childcare and housework involvement (section 5.1), quality of care as measured by child interaction activities such as playing and reading books (section 5.2) and parental leave taking (section 5.3).

## 5.1 Self-reported childcare and housework involvement

Figure 8 gives an overview of the available measures of self-reported direct involvement in childcare. Our two main outcome variables are the self-reported frequency of involvement (upper graph) in the child's day-to-day care as well as the self-reported frequency of direct responsibility for the child (lower graph). Both are measured on a scale from 1 (Not much of the time) to 4 (All the time). For both outcomes, information on both anticipated and actual involvement (at 9 months and 2 years) are reported for both fathers and mothers.

The figure shows that fathers are distinctly less involved than mothers, with more fathers being involved 'some of the time' rather than 'most of the time' during the first two years of the child's life. The differences are even stronger if direct responsibility for the child is considered, with only 5% (6%) of fathers reporting to be directly responsible for the child all the time in the 9-month (2-year) interview. As opposed to this, while 32% (21%) of fathers report to be directly responsible not much of the time for their 9 months (2 year) old child, hardly any mothers (1% and 2% respectively) report no direct responsibility.

Figure 8 also shows that fathers strongly overestimate their anticipated involvement in day-to-day care and direct responsibility in childcare with respect to (at least) the first 2 years after birth. In the antenatal interview 78% of fathers anticipated they would be involved in the day-to-day care of their child at least most of the time while only 35% (47%) actually did at 9 months (2 years). Interestingly, mothers also overestimate their anticipated involvement in day-to-day care and direct responsibility, although the degree of overestimation among mothers is much less. Given that both parents overestimate their involvement, it is possible that part of this finding is due to the wording of the questions. Mothers and fathers were asked about "the plans you and your partner may have about being involved with your baby after they are born". No specific time period after birth was given. Parents may have interpreted this to mean immediately after the child is born when they are a newborn, rather than when the child is 9-months or 2-years old, which may, at least partly, explain this overestimation.





### FREQUENCY OF INVOLVEMENT IN DAY-TO-DAY CARE

### FREQUENCY OF DIRECT RESPONSIBILITY



Source: GUINZ DCW0, DCW1 and DCW2, own calculations and illustrations.

Furthermore, and not shown in the figure, we find that 61% of fathers report a higher involvement category in DCW0 (the antenatal interview), when anticipating involvement, relative to the actual involvement reported in DCW1 (at 9 months). The comparable figure for mothers is 33%. Nevertheless, as can be seen in Figure 8, fathers' involvement and direct responsibility increases from DCW1 to DCW2, indicating higher involvement of fathers with children over time in the first two years.

Gender differences are also evident, although not shown in the figure, for hours spent on domestic housework. The weekly hours spent on domestic housework is 10.24 hours for men and 17.49 hours for women in DCW2.

Based on these strong differences in childcare involvement between mothers and fathers, we take a closer look at the satisfaction with these shares. Figure 9 reports the responses to a question in DCW2 on whether the father (mother) perceives the current division of care and housework as fair. Although fathers are observed to be involved less, 71% (62%) of fathers still perceive this distribution of responsibilities in childcare (housework) as fair as opposed to only 56% (48%) of women. Not shown in Figure 9, only 53% (49%) of partners agree on this with only 42% (32%) agreeing on a fair distribution of tasks while 11% (16%) agree that the mother does more than her fair share. Thus, it appears that fathers not only take on fewer domestic tasks, they are generally more satisfied with this arrangement than mothers are.





Source: GUINZ DCW2, own calculations and illustrations.

### 5.2 Quality of care

Table 1 provides an overview of the available measures for activities undertaken by the parent with the child. These include playing games, playing with toys, telling stories / singing songs and reading books. Fathers (and mothers) report the frequency in which they engage in these activities. Responses are categorised into 'seldom' or 'never', 'at least once a week', 'once a day' and 'several times a day'. We expect that higher frequency of these activities proxy as a higher 'quality of care'. While the activities

playing games and playing with toys are only reported in DCW1, telling stories / singing songs as well as reading books are reported in DCW1 and DCW2, allowing for a dynamic perspective.

	Father		Ma	ther
	DCW1	DCW2	DCW1	DCW2
Playing games				
Seldom or Never	0.01		0.01	
At least once a week	0.19		0.12	
Once a day	0.26		0.19	
Several times a day	0.54		0.68	
Playing with toys				
Seldom or Never	0.01		0.00	
At least once a week	0.16		0.04	
Once a day	0.24		0.11	
Several times a day	0.58		0.84	
Telling stories / singing songs				
Seldom or Never	0.14	0.38	0.03	0.34
At least once a week	0.36	0.50	0.15	0.46
Once a day	0.25	0.10	0.28	0.14
Several times a day	0.25	0.02	0.54	0.05
Reading books				
Seldom or Never	0.36	0.10	0.13	0.03
At least once a week	0.40	0.40	0.31	0.22
Once a day	0.19	0.31	0.37	0.28
Several times a day	0.05	0.20	0.20	0.47

### Table 1 - Descriptive statistics of quality of care

Source: GUINZ DCW1 and DCW2, own calculations and illustrations.

In general, fathers are less engaged in the activities reported on in Table 1, compared to mothers. Fathers are less likely to report daily participation in these activities than mothers especially in the 9-month interview. Interestingly, engaging once a day is relatively more common among fathers while the majority of mothers engage in most activities several times a day, supporting the idea of the fathers being the storytellers and entertainers in the evenings after work.

In line with the findings on involvement in the previous section, the gap in involvement in these activities between mothers and fathers decreases as the children age. In particular, the share of fathers who read books to their children once a day or more increases. It is useful to note that only a very small fraction of fathers respond saying they 'seldom or never' play with toys or play games with their children. Nearly all fathers report playing with their children at least once a week.

## 5.3 Parental leave taking

As a last group of variables to measure paternal involvement, we analyse paternal leave taking. In the literature, parental leave taking is the most widely used measure of fathers' involvement. This is because studies often lack good measures of the within-household division of unpaid work and care responsibilities, so parental leave taking is often the only observed proxy for paternal involvement, especially when administrative data sources are used. Contrasting the observed leave taking of fathers with the other measures of involvement will thus form an important part of the later analysis to capture

how representative leave taking is for the informal division of care responsibilities in families. Of course, the amount of parental leave taken will be influenced by government policy. Mothers who had children during the GUINZ window of March 2009-April 2010 were eligible for a year of unpaid right-to-return-to-work leave and 14 weeks of government-funded paid parental leave provided they met certain employment criteria. Fathers were entitled to two weeks unpaid leave if they met certain employment criteria. In addition, the mother's leave entitlement could be transferred to the father – while data on how frequently this occurred is scant, indicators suggests that this was not common.

Figure 10 gives an overview of parental leave taken by employed fathers in GUINZ. It includes the actual leave taking as reported in the 9-month interview (DCW1) as well as the reported preferred and anticipated leave taking from the antenatal interview (DCW0). The reported preferred, anticipated, and actual leave is categorised into the following seven categories: No Leave; 1 week; 2 weeks; between 2 weeks and 1 month; between 1 month and 2 months; between 2 months and 3 months; more than 3 months (including still on leave at 9 months).

As opposed to the two earlier involvement variables, parental leave taking is only measured for employed fathers, reducing the sample to 2,539 households.



### Figure 10 - Paternal leave taking of employed fathers

*Source:* GUINZ DCW0 and DCW1, own calculations and illustrations.

As the results in the left bar show, all fathers would prefer to take some leave with 66% having a preference for more than two weeks and 19% wanting to take more than 3 months. In comparison, 9% anticipate they will not be taking any leave and we find that for 57% of the employed sample, anticipated leave is shorter than the preferred amount. Turning to the right bar, the actual leave is again shorter than the anticipated leave in 30% of the cases with 14% of fathers ending up not taking any leave.

In addition, Table 2 reports the answers of fathers to the question about why their anticipated leave is shorter than their preferred leave as well as why they ended up taking no leave. The most common reason for taking no / shorter leave is professional or work commitments (36% and 47%) followed by

financial reasons (16% and 25%). Parenting preferences are the reason in only 4% and 10% of the sample.

It is noteworthy that most fathers would like to take more leave than they do, and that work commitments and financial considerations are the main reasons for taking less leave than they would like, while parenting preferences are given as a reason by only a small share of fathers. These results suggest that the low levels of paternity leave are not just being driven by traditional gender norms, but also reflect financial realities. This is particularly relevant in the NZ policy context where no parental leave is specifically set aside for fathers, unlike in the majority of other OECD countries. However, it should also be kept in mind that the international evidence on daddy months is mixed. Daddy months are found to contribute positively to a more equal division of domestic duties and labour supply by reducing the financial constraints of fathers who are willing to take leave, but other factors seem to be overshadowing families' decisions about how to divide paid and unpaid work.

Reason why	anticipated shorter than prefered	no actual leave	
Observations	1,840	470	
Financial Reasons	25%	16%	
Government Regulations	4%	2%	
Company or Employer Regulations	13%	4%	
Professional or Work Commitments	47%	36%	
Parenting Preferences	4%	10%	
Resigned or Redundancy		5%	
Flexible work arranged		3%	
Self employed		13%	
Other		12%	

### Table 2 - Reason for no / shorter leave

Source: GUINZ DCW0 and DCW1, own calculations and illustrations.

Finally, Table 3 gives an overview of the types of leave fathers take. Categories include paid, unpaid, annual, sick and / or other types of leaves, with the groups not being mutually exclusive. Most fathers take annual leave (54%) with an average of 2.22 weeks taken. About 28% of fathers indicate taking paid leave and 22% take unpaid leave. Unfortunately, we cannot split out paid leave into governmental paid and employer paid. However, we know from other sources that the take up of governmental paid leave by fathers is extremely low (less than 1%) – see Morrissey (2020) and Meehan (forthcoming).<sup>9</sup> We can therefore assume that the majority of the 28% is likely to be employer paid parental leave.

Table 3 -	Types	of paternal	leave
-----------	-------	-------------	-------

Leave types	share	mean length
Paid leave	0.28	2.37
Unpaid leave	0.22	2.78
Annual leave	0.54	2.22
Other types of leave		4.54
Sick leave	0.02	
Other leave	0.03	

Source: GUINZ DCW1, own calculations and illustrations. Length of leave is in weeks.

<sup>9</sup> Based on the share of fathers who received any IRD paid parental leave payments within one year of having a child.

### 5.3.1 How representative is leave taking of paternal involvement?

One issue, which has been discussed extensively in prior literature, is whether leave taking is a good proxy for paternal involvement in childcare in general, i.e. is the observed paternal leave taking representative of paternal involvement in the day-to-day care as well as the quality of care and engagement provided by fathers. This is an important issue because, due to the lack of available measures of direct involvement, leave taking is frequently used as a proxy measure for paternal engagement. Drawing conclusions on drivers and consequences of paternal involvement from findings on leave taking thus crucially depends on the assumption those that take longer leave are more involved relative to those that take shorter leave.

The advantage of the GUINZ data is that we are able to observe both measures of involvement and leave taking and derive correlations as well as mean equality tests. Table 4 summarises the results of a mean equality test of the direct involvement measures by leave taking (i.e., whether the father took 2 weeks or less or more than 2 weeks of leave).

, ,	Leave Taking				
Direct involvement measure	2 weeks or less	More than 2 weeks			
Frequency of involvement in day-to-day-care at 9 months					
Not much of the time	0.07	0.06			
Some of the time	0.61	0.56	**		
Most of the time	0.26	0.30			
All the time	0.06	0.08	**		
Frequency of direct responsibility at 9 months					
Not much of the time	0.32	0.32			
Some of the time	0.53	0.52			
Most of the time	0.11	0.12			
All the time	0.04	0.04			
Frequency of involvement in day-to-day-care at 2	years				
Not much of the time	0.05	0.05			
Some of the time	0.49	0.45	*		
Most of the time	0.34	0.35			
All the time	0.12	0.14	*		
Frequency of direct responsibility at 2 years					
Not much of the time	0.22	0.22			
Some of the time	0.58	0.55			
Most of the time	0.15	0.18			
All the time	0.05	0.06			
Quality of care					
Total score	12.27	12.79	***		
Total score > median	0.49	0.56	* * *		
Observations	2,081	574			

### Table 4 - Direct involvement measures by leave taking

*Source:* GUINZ DCW1 and DCW2, own calculations.

*Notes:* Total quality of care score is based on the sum of all activity categories in DCW1 and DCW2. See section 6.1 for details. Significant stats in column 3 refer to the significance levels of a t-test for mean equality between columns 2 and 3 with \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

Interestingly, the results show that in the GUINZ survey, leave taking is not a good proxy for paternal involvement. Average involvement in day-to-day care as well as direct responsibility at the ages 9 months and 2 years only weakly differ between fathers who took 2 weeks or less and fathers who took more than 2 weeks leave. Only the quality of care index seems to be positively linked to leave taking.

Two reasons can be responsible for this finding. First, paternal leave taking could be a bad proxy for paternal involvement in general as each could be caused by very different determinants. It also highlights that parental leave taking is often the only observed proxy for paternal involvement, especially in the absence of time-use data that provide measures of the within-household division of paid work and unpaid work, including childcare responsibilities. It suggests that time-use data can provide valuable insights into how cultural gender norms at work are shifting, and that these factors are difficult to proxy with administrative data sources on parental-leave taking. Second, these observations could be a result of generally low levels of leave taking in NZ and thus a less accurate description of intrinsic leave taking based on the external restrictions imposed on fathers. It may also somewhat limit the applicability of this finding to international literature since the leave period is generally shorter in NZ, although even in countries with earmarked paternity leave, the amount of leave taken rarely exceeds one or two months (e.g., Eriksson 2005 for Sweden; Samtleben, Schaeper, and Wrohlich 2019 for Germany).

# 6. RESEARCH AIM 2

The second research aim is to analyse the external and internal determinants of the different levels of paternal involvement discussed above. The focus of this section is on the empirical identification of the relationship between different involvement variables and a number of key demographic, socio-economic and psychological characteristics of the father, mother and child.

## 6.1 Variables and empirical approach

For the purpose of this analysis, we concentrated on five main outcome variables based on the different involvement measures discussed above. To simplify the interpretation of estimated associations, all outcome measures will be analysed in a binary form:<sup>10</sup>

- 1. Father in GUINZ Dummy variable for whether a father participates in GUINZ.
- 2. **High childcare involvement at 9 months** Dummy variable for whether a father is involved in the day-to-day care of the child at least most of the time at 9 months.
- 3. **High childcare involvement at 2 years** Dummy variable for whether a father is involved in the day-to-day care of the child at least most of the time at 2 years.
- 4. Long leave Dummy variable for whether a father takes more than 2 weeks of leave.
- 5. **High quality care** Dummy variable for whether a father provides a higher-than-median amount of activities based on the sum of all activity categories in DCW1 and DCW2 (1 if father has higher than median value of total quality care, 0 otherwise).<sup>11</sup>

We use **multivariate logit analysis** as our key estimation method:

$$P(Y_{i} = 1) = P(\beta_{1} + \beta_{2}C_{i} + \beta_{3}HH_{i} + \beta_{4}M_{i} + \beta_{5}F_{i} + \varepsilon_{i} > 0)$$
(1)

With  $Y_i$  being the binary outcome variable from the list above for child i. The explanatory variables cover the following domains:

- **Child's characteristics (C<sub>i</sub>)**: twin birth; gender; planned pregnancy; number of siblings; indicators for young siblings (under 5 and under 2); subjective health; developmental problems.
- Household characteristics (HH<sub>i</sub>): household type; household income; perceived helpfulness of family; perceived relationship quality.<sup>12</sup>

<sup>&</sup>lt;sup>10</sup> Estimation results of continuous or ordinal versions of the outcome variables (in the case of leave taking and quality care) are additionally provided in the Appendix and discussed in Section 6.2.

<sup>&</sup>lt;sup>11</sup> A summarised continuous measure of quality care is generated by assigning point-values to the four different possible responses to the quality care questions: seldom or never (=0), at least once a week (=1), once a day (=2), several times a day (=3). These scores are summed up for the seven different activities: playing games, playing with toys, talking, telling stories at DCW1 and DCW2 and reading books at DCW1 and DCW2. This results in a continuous scale ranging from 0 to 21.

<sup>&</sup>lt;sup>12</sup> Perceived helpfulness of the families as well as relationship quality are used from the father's survey responses in order to capture the subjective reasoning of fathers most closely.

- Mother's socio-economic and demographic characteristics (M<sub>i</sub>): marital status; age, migration status; education; ethnicity; employment status/occupation).<sup>13</sup>
- Father's socio-economic, demographic and psychological characteristics (F<sub>i</sub>): biological father; age; criminal history (DCW1); migration status; ethnicity; education; employment status/occupation; income differences between mother and father; shiftwork (DCW1); self-employment (DCW2); working hours (DCW1); mental health diagnosis; physical health diagnosis; subjective health; Big Five personality traits; perceived stress scale.

If not indicated otherwise, all explanatory variables (except the child characteristics) are observed in the antenatal interview to avoid endogeneity in the later estimation models due to reverse causality. A detailed list of the explanatory variables including a definition and summary statistics are provided in Tables A.2 and A.3 in the Appendix.

We also analyse the role of additional psycho-social characteristics of the fathers in the parental context such as perceived work-life balance, parental influence, parental identity, parental confidence, subjective parental quality, and parental satisfaction. Nevertheless, these variables are highly endogenous and can potentially explain important relationships between the explanatory and outcome variables and should not be treated as confounders. These variables are thus not included in the main estimation models but are discussed separately at the end of Section 6.2.

## 6.2 Results

Table 5 summarises the estimation results of the full estimation model in line with equation (1) for all five outcome variables. Average marginal effects are reported. Results for the stepwise inclusion of explanatory variables in which we add 1) children's + household characteristics, 2) mother's characteristics and 3) father's characteristics in a stepwise manner are reported in Table A.4, Table A.5, Table A.6 and Table A.7 in the Appendix.

	Eather in		High CC	High CC	High quality
	GUINZ	Long leave	(9 month)	(2 years)	care
CHILD AND PREGNANCY CHARACTERISTICS					
Is twin	0.015	0.174***	-0.019	0.179***	-0.031
ls girl	-0.006	-0.009	-0.013	0.003	-0.017
Pregnancy planned	0.048***	0.005	-0.027	-0.007	0.028
Number of siblings (Ref: None)					
One	-0.062***	-0.066**	-0.001	0.010	-0.112***
Two	-0.081***	-0.107***	-0.024	-0.004	-0.193***
Three	-0.123***	-0.094**	0.051	-0.038	-0.190***
Four or more	-0.141***	-0.133***	-0.052	0.065	-0.153**
Youngest sibling under 5	-0.005	0.032	-0.030	-0.026	-0.060**
Youngest sibling under 2	0.014	0.020	0.009	0.008	0.003

### Table 5 - Determinants of paternal involvement (marginal effects of logit estimation)

<sup>&</sup>lt;sup>13</sup> Employment status/occupation and ethnicity of the mother are only used as explanatory variables in the estimation of father's participation in GUINZ, in which father's characteristics are not available. In all other estimation models, including mother's detailed information, this causes issues of multicollinearity and perfect prediction due to the strong overlap with the paternal information.

Child's subjective health (Ref: Excellent)								
Very good	-0.018	0.010	0.009	-0.026	-0.013			
Good	-0.022	-0.005	-0.017	0.034	-0.090***			
Fair	0.038	-0.114***	0.025	-0.011	0.003			
Child has developmental problems	-0.013	0.026	0.004	-0.021	-0.007			
HOUSEHOLD CHARACTERISTCS								
Household type = Multiple adults	-0.040**	-0.015	-0.017	-0.006	-0.040*			
Household income (Ref: Less than 20K)								
20-30K	0.046	0.016	0.011	-0.049	-0.017			
30-50K	0.111**	0.016	0.000	-0.095	0.017			
50-70K	0.136***	0.073	-0.000	-0.123	0.028			
70-100K	0.143***	0.065	0.016	-0.045	0.043			
100-150K	0.173***	0.106	0.031	-0.062	0.065			
More than 150K	0.180***	0.066	0.055	-0.059	0.069			
Relationship quality (rated by father)		0.019**	0.018*	0.031***	0.045***			
Average helpfulness of family (rated by father)		-0.020**	0.039***	0.022**	0.025***			
MOTHER'S CHARACTERISTICS								
Mother is married	0.085***	-0.027	-0.020	-0.012	0.029			
Mother is cohabiting with partner	0.222***	0.018	-0.163*	-0.200*	-0.029			
Mother's age (Ref: <25 years)								
25-29 years	0.070***	-0.020	0.006	-0.072*	0.040			
30-34 years	0.068***	0.006	0.029	-0.052	0.059			
35-39 years	0.094***	0.044	0.033	-0.050	0.008			
40+ years	0.089**	0.081	0.015	-0.100*	0.031			
Mother born in NZ	0.013	-0.001	-0.033	-0.073***	0.067***			
Mother's prioritised ethnicity (Ref: NZ European)								
Māori	-0.067***							
Pacific Peoples	-0.173***							
Asian	-0.101***							
MELAA and Other	-0.004							
Mother's education (Ref: No sec qualification)								
Secondary school/NCEA 1-4	0.064*	0.058	0.060	0.134**	0.093			
Diploma/Trade cert/NCEA 5-6	0.031	0.038	0.041	0.119**	0.081			
Bachelors degree	0.085**	0.055	0.024	0.094*	0.061			
Higher degree	0.099**	0.029	0.048	0.089	0.139**			
Mother's occupation (Ref: Professional or admin. worker)								
Not employed	-0.036**							
Managers	-0.027							
Technic./ trades workers/ mach. operators	-0.097**							
Other	-0.002							
FATHER'S CHARACTERISTICS								
Partner is biological father		-0.188**	-0.062	0.132	0.014			
Fathers' age (Ref: <25 years)								
25-29 years		0.013	0.001	0.062	0.021			
30-34 years		0.029	-0.009	0.062	-0.006			
35-39 years		0.022	-0.017	0.084	-0.030			
40+ years		0.044	0.066	0.131**	-0.008			
Father has criminal history		-0.163	-0.029	0.005	-0.096			
Father born in NZ		0.021	-0.062***	-U.UI/	0.004			

Observations	4,443	2,539	3,098	3,098	3,098
Perceived stress scale		-0.011	0.009	0.008	0.018
Openness		0.014	0.026*	0.021	0.072***
Neuroticism		-0.020	0.008	0.035**	0.020
Conscientiousness		-0.040***	0.041***	0.024	0.007
Agreeableness		-0.014	0.024	0.005	0.028
Extraversion		-0.011	-0.016	0.026*	0.036***
Subjective health		-0.061**	0.009	0.012	0.045*
Physical health		0.011	-0.017	0.004	0.011
Mental health		0.024	-0.023	-0.070**	0.027
Father overtime (DCW1)		0.013	-0.079***	-0.083***	0.005
Father fulltime (DCW1)		-0.083***	-0.131***	-0.032	-0.131***
Father self-employed (DCW2)		-0.077***	-0.028	-0.058**	0.017
Father shiftwork (DCW1)		0.091***	0.040*	0.006	-0.018
Income difference in \$10K		0.001	-0.009***	-0.016***	-0.006***
Other		0.013	0.057*	0.037	0.026
Technic./ trades workers/ mach. operators		0.002	0.065**	0.006	-0.029
Managers		-0.071***	-0.025	-0.031	-0.037
Not employed			0.025	-0.011	-0.002
Fathers' occupation (Ref: Professional or admin	. worker)				
Higher degree		0.098**	-0.052	0.012	0.127***
Bachelors degree		0.055	-0.091**	-0.038	0.096**
Diploma/Trade cert/NCEA 5-6		0.077**	-0.037	-0.027	0.080*
Sec school/NCEA 1-4		0.026	-0.040	-0.011	0.106**
Fathers' education (Ref: No sec qualification)					
MELAA and other		0.007	0.037	0.058	-0.111*
Asian		0.060*	-0.050	-0.006	-0.047
Pasifika		0.019	0.081**	0.106***	0.071**
Māori		0.006	0.087***	0.040	0.041
Fathers' prioritised ethnicity (Ref: NZ European)	)				

Source: GUINZ DCW0, DCW1 and DCW2, own calculations and illustrations.

Notes: Explanatory variables are sourced from DCWO unless specified otherwise. Standard errors as well as models in which explanatory variables are included in a stepwise fashion are reported in Table A.4, Table A.5, Table A.6 and Table A.7 in the Appendix.

#### 6.2.1 Participation in GUINZ

Results of the estimation using the binary indicator for father's participation in GUINZ as the outcome variable for the full sample (N=4,443) are shown in the first column of Table 4. The estimated effects are very much in line with the descriptive comparison and mean equality test between the final sample and the excluded sample of families in which the father does not participate in GUINZ (discussed earlier in Section 4.3 and detailed in Table A.1 in the Appendix). We can see that a planned pregnancy as well as marriage and cohabitation and a mother's age above 25 are positively associated with the father participating in GUiNZ, pointing towards the role of a stable family structure encouraging the father's GUINZ participation. For example, in households in which the mother is cohabiting with her partner, the partner is 22.2 percentage points more likely to participate in GUINZ.

An increasing number of siblings as well as additional adults in the household decreases the probability of the father's GUINZ participation. In a family with four or more children, the father is on average 14.1 percentage points less likely to participate than in a family with no children other than the GUINZ child. This is true even through the estimation in Table 5 for many aspects of family socio-economic status, indicating the especially important role of the size of the childcare and housework workload.

Another significant factor is socio-economic status. Fathers in households with an annual household income of above NZ\$150,000 are 18 percentage points more likely to participate in GUINZ than households with less than NZ\$20,000. Lastly, ethnicity was also strongly associated with GUINZ participation even when socio-economic differences are controlled for. Using mother's ethnicity (which is the sole ethnicity indicator available for the household irrespective of whether the father participated in GUINZ), we find that fathers in Māori, Pasifika as well as Asian households are less likely to participate in GUINZ as compared to NZ European households. For example, fathers in Pasifika households are on average 17.3 percentage points less likely to participate, relative to NZ European households.

### 6.2.2 Leave taking

Column 2 of Table 5 summarises the estimation results of the full model (including all explanatory variables in line with equation 1) for the binary indicator of taking leave for more than 2 weeks. Recall that Figure 10 earlier (in Section 5.3) indicated that 64% of employed fathers in the GUINZ sample took either one or two weeks leave; with a further 14% taking no leave. Therefore, understanding the determinants of having a leave length greater than the two-week threshold is useful for further investigating determinants of greater paternal involvement. Given that leave taking is only observed of fathers who indicate they are employed prior to birth, equation 1 is estimated for the subsample of 2,539 employed fathers.

In addition, estimation results for the stepwise inclusion of explanatory variables are presented in columns (1) to (4) of Table A.6 in the Appendix, while columns (5) to (8) report the results for the linear estimation using the continuous measure of weeks of leave as the outcome variable. As results from the linear estimation are very much in line with the findings using the binary indicator, we concentrate on the discussion of the results of the binary model that are presented in column 2 of Table 4.

We find that the results on number of siblings are very similar to the findings on GUINZ participation, i.e. the more children already in the household, the lower the probability that the father is taking more than two weeks leave. For instance, fathers of four or more children are 13.3 percentage points less likely to take more than two weeks leave than first-time fathers. Having twins increases the likelihood of father's taking more than two-weeks leave by 17.4 percentage points, again relative to first-time fathers.

Holding everything else constant, the father's rating of the relationship quality with the mother is on average positively associated with taking more than two weeks leave, indicating two distinct potential mechanisms which we are not able to distinguish between: 1) fathers with healthy relationships with the mothers are more likely to be involved, and/or 2) fathers being involved is good for the relationship between the mother and father.

Biological fathers are observed to be less likely to take more than two weeks leave than non-biological fathers. However, this result should be interpreted with caution due to the very small number of non-biological fathers in our sample. This particular finding might be at risk of being driven by a small group of outliers.
With respect to fathers' characteristics, several observations can be made. First, paternal leave taking on average seems to increase with higher levels of education (although not linearly). Second, in line with the known connection between the internal substitutability of leave-takers within firms and parental leave length (Huebener et al. 2021), managers and self-employed fathers as well as full-time employed fathers are less likely to take more than two weeks leave, likely to the consideration of firm-side problems with substitution of own work. In contrast, fathers working in shifts are on average 9 percentage points more likely to take more than two weeks leave, potentially driven by shift work being more common in routine jobs (i.e. again better internal substitutability).

In terms of personality traits, conscientiousness in particular stands out as it is negatively associated with taking more than two weeks leave. This is likely due to an increased sense of responsibility in the workplace.

#### 6.2.3 Childcare involvement

Columns 3 and 4 of Table 5 summarises the estimation results of the full model for the binary indicator of high childcare involvement, separately for the ages 9 months (DCW1) and 2 years (DCW2). High involvement here is defined as the father involved at least most of the time in the day-to-day care of the child. If we compare the determinants of high childcare involvement with the determinants of taking more than two-weeks leave, there are a variety of similarities and differences observed. Many factors such as having twins (positive effect), relationship quality (positive effect) and factors related to workload such as self-employment, full-time employment and overtime (negative effect) have similar effects on paternal childcare involvement as on paternal leave taking. All other factors with significant associations are either new or have reversed signs.

We can, for example, see that cohabitation, which has an insignificant effect on taking more than twoweeks leave, has a negative association with paternal childcare involvement at both the 9 months and 2 years stage. This may be potentially driven by shared custody arrangements in cases where the mother is not cohabiting with partner, and therefore those fathers have to take over the direct responsibility when the child is in their household.

Table 5 also shows that the income difference between fathers and mothers plays a crucial role in explaining fathers' involvement in the day-to-day care, i.e. the division of unpaid labour in the household, even if other work- and education-related factors are controlled for. The higher the income difference (i.e. in terms of the father out-earning the mother) the lower the probability of the father undertaking high childcare involvement, at both 9 months and 2 years.

With respect to the father's ethnicity, we see an interesting pattern. While fathers in Māori and Pasifika households are less likely to participate in GUINZ,<sup>14</sup> and ethnicity had no effect on leave taking behaviour, we also observe fathers with prioritised Māori and Pasifika ethnicity to be more likely, on average, to undertake high childcare involvement, relative to their NZ European counterparts. The lower probability of GUINZ participation might thus indicate a stronger hesitation towards the survey per se, rather than being a proxy indicator for paternal involvement with the child.

<sup>&</sup>lt;sup>14</sup> Note that, as mentioned in section 6.2.1, we cannot directly observe the ethnicity of fathers who did not participate in GUINZ, therefore, we use the mother's ethnicity.

Another interesting observation can be made with respect to conscientiousness. While it was negatively related to taking more than two weeks leave, the association between conscientiousness and high childcare involvement is positive (at age 9 months). Conscientious fathers might face a strong two-edged sense of obligation towards their workplace and their children causing them to simultaneously take over relatively more paid and unpaid work.

### 6.2.4 Quality of care

The last outcome of interest in Table 5 is the binary indicator of high quality care. Estimation results for the full model are shown in column 5 and the outcome variable is defined as more-than-median amount of activities based on the sum of all activity categories (playing games, playing with toys, talking, telling stories in DCW 1 and DCW 2 and reading books in DCW 1 and DCW 2). Many of the findings follow a similar pattern to those discussed above with respect to other outcome variables related to paternal involvement. For example, in line with the likelihood of participating in GUINZ, as well as the likelihood of taking more than two-weeks leave, the probability of high quality care by the father suffers the higher the number of siblings in the household. Closely related to this, the helpfulness rating of the family (as rated by the father) is on average positively related to fathers' quality care, which potentially indicates more time for childcare activities when other domestic duties are taken care of by others in the household.

Also, in line with earlier findings, the relationship quality between the mother and the father has an on average positive association with high quality care of fathers. Again, the data does not enable us to disentangle the direction of causation between these variables, but there are likely effects in both directions. The results also show that in general a higher educational attainment level for the father is more positively associated with their quality care level. For example, fathers with higher than a bachelor's degree are on average 12.7 percentage points more likely to provide more-than-median quality care.

Very much in line with the finding on ethnicity discussed in Section 6.2.3, Pasifika fathers are not only more likely to be involved in the day-to-day care of the child but also more likely to provide more quality care.

Lastly, while conscientiousness was an important determinant of the quantity of childcare (column 2 of Table 5), extraversion and openness of the father are the personality traits that are positively associated with high quality care provision.

### 6.2.5 Psycho-social characteristics as mechanisms

In further results (detailed in the Appendix), we add to our estimation model the father's psycho-social characteristics. Additional input variables include perceived influence, parental identity, parental confidence, perception of being a good parent, parental satisfaction, and perceived importance of a work-life balance. These variables are likely to be important mechanisms of the relationship between other exogenous factors and the outcomes of interest. Results of these inclusions are provided in columns 4 and 8 of Table A.5 (for the high childcare involvement indicator at 9 months and 2 years respectively); column 4 of Table A.6 (for the outcome variable for taking more than two weeks leave); and column 4 of Table A.7 (for the outcome variable of high quality care).

Results show that perceived influence, parental confidence, perception of being a good parent and parental satisfaction have a strong positive association with the high childcare involvement (Table A.5)

and high quality care (Table A.7) but is not statistically significant with respect to taking more than two weeks leave (Table A.6).

It is also apparent that after controlling for these psycho-social characteristics in our estimation models, the magnitudes of several other determinants of paternal involvement reduce. This suggests a potential mediating role of these characteristics. This is the case, for example, for fathers' rating of helpfulness of family and relationship quality between mother and father with respect to the outcome variables of high childcare involvement and high quality care.

Also, ethnicity (i.e. being Māori or Pasifika) has a smaller or no significant effect on paternal involvement and / or quality of care if these psycho-social characteristics are controlled for, suggesting an important difference in paternal confidence and satisfaction between Māori and Pasifika fathers and their NZ European counterparts, which positively affects their involvement.

### 6.2.6 Heterogeneity analysis by ethnicity

Given the interesting findings with respect to differences in paternal involvement among ethnic groups, we took a closer look into the heterogeneity of all other factors by ethnicity. However, due to the very small samples in all ethnic groups except NZ European, the estimation of the models above in stratified samples by ethnic groups is not feasible. Therefore, in order to check the association of the different characteristics and paternal involvement for heterogeneity with respect to the father's ethnicity, we conduct a mean equality test separately for NZ Europeans, Māori and Pasifika. The results of this descriptive analysis and the t-tests are reported in Table A.8 (paternal involvement at 9 months), Table A.9 (quality of care) and Table A.10 (parental leave taking) in the Appendix. Note that due to the small sample sizes even these descriptive results have to be interpreted with care.

With respect to paternal involvement we can, for example, see that the negative association between already having children and paternal involvement is mainly driven by NZ European fathers whereas the association seems to be much less clear for Māori and Pasifika fathers. In contrast, having additional children in the household has a negative association with quality of care provided by the father for all three ethnicity groups portrayed in Table A.9.

We can also see that although having multiple adults in the household had no significant effect on paternal involvement on average, having additional adults in the household is positively associated with the involvement of Māori fathers and children's health seems to be positively associated with the involvement of Pasifika fathers. Also, while marriage negatively affects involvement of NZ European and Māori fathers, this isn't the case for Pasifika fathers, where the association is positive. Additionally, the positive association of relationship quality and paternal involvement can only be observed for NZ European fathers while the association between relationship quality and high quality of care is similar for all ethnic groups.

In terms of the role of financial and occupational factors, variables such as income difference and being employed fulltime are positively associated with high paternal involvement at 9 months, regardless of the father's ethnicity. If the mother is not employed, there is lower involvement from the father – but this is only true for NZ European fathers. Further, higher education level is associated with a higher quality of care, again only for NZ European fathers.

As opposed to the differences in factors associated with paternal involvement and quality of care between ethnic groups, the variables associated with paternal leave taking are quite homogenous between NZ European, Māori and Pasifika fathers. Only the question of internal substitution ability (i.e. the lower likelihood of leave taking for self-employed fathers and fathers in managerial positions) seems to be less pronounced for Māori and Pasifika fathers. This, again, underlines the important differences between determinants of involvement on the one hand and determinants of leave taking on the other.

# 7. RESEARCH AIM 3

The first two research aims of this study helped us get a picture of the involvement of NZ fathers as well as determinants of this involvement. To fully assess the relevance of these findings, the consequences of different levels of fathers' involvement on children's outcomes have to be analysed in detail. The importance of parent's quantity and quality of time for children's cognitive and psychological development has been the key justification for the political and academic interest in increased paternal involvement (see for example Cools, Fiva, and Kirkebøen 2015; Mangiavacchi, Piccoli, and Pieroni 2021; El Nokali, Bachman, and Votruba-Drzal 2010).

In this section, we present a descriptive analysis of the association between paternal involvement and different measures of children's cognitive, physical, and psychological development at later time points. This analysis should be seen as exploratory evidence on this topic, giving an indication for the importance of further research in this area.

# 7.1 Variables and empirical approach

The information on children's cognitive, physical and psychological development is observed in DCW3 (31-month interview), DCW4 (45-month interview) and DCW5 (54-month interview). Table 6 gives an overview of the variables we are using as well as a description of the variable construction and the DCW it was measured in. First, cognitive development is measured using information on language development based on the number of words used by the child (on feelings, shapes, colours) reported by the mother at ages 31 and 45 months. Secondly, physical development is based on motor skills, such as bending over and walking straight lines, reported by the mother at 54 months.

The third domain of interest is psychological development. It is evaluated using two established questionnaires: the Strengths and Difficulties Questionnaire (SDQ) based on the work of Goodman (1997) and the Temperament / Child Behaviour Questionnaire (CBQ) based on Putnam and Rothbart (2006). The SDQ is an internationally used 25-item scale that usually measures five aspects of child behaviour: emotional symptoms, peer relationship problems, hyperactivity/inattention, conduct problems and prosocial behaviour. In line with the guidelines given in the data user guide (Growing Up in New Zealand, 2018), we drop 'conduct problems' from the list due to a missing item in the questionnaire, and thus concentrate on the four remaining aspects of behaviour. Additionally, to make signs comparable between concepts, we reverse all items referring to prosocial behaviour and thus discuss antisocial behaviour instead. The CBQ is a similarly widely used scale which measures three aspects of children's temperament: negative affect, effortful control, surgency. Although research described in the data user guide (Growing Up in New Zealand 2018) suggests that a six factor structure potentially is a better fit for the data, we concentrate on the original three factors for reasons of simplicity. In line with the rescaling done for the SDQ items, we rescale all items corresponding to effortful control and surgency such that high values correspond to 'negative' psychological development (i.e. low surgency and low effortful control). All outcome variables are used in a standardised from in order to simplify coefficient interpretation and comparability. A list of all items corresponding to the SDQ and CBQ factors can be found in Tables A.11 and A.12 in the Appendix.

Variable		Descriptions	DCW
Cognitive of	<u>levelopment</u>	· · · · ·	
Language development		<ul> <li>Continuous variable: Total number of words out of a list of 32 the child is using in English or any other language (reported by mother).</li> <li>12 words from month 31 (DCW3) questionnaire (e.g. happy, sad, angry, wish etc.)</li> <li>20 words (colours, shapes, own name) from the month 45 (DCW4) questionnaire</li> </ul>	3 4
Physical de	evelopment		
Motor skill	S	<ul> <li>Continuous variable: Average frequency of whether child is able to perform a list of 11 actions rated from 1 (Never) to 4 (Always)</li> <li>Actions: bend over, walk straight line, walk backwards etc.</li> </ul>	5
Psychologi	cal development		
e (SDQ)	Emotional problems	Ordinal variable ranging from 0 to 10: Sum of corresponding 5 items of the SDQ rated on the scale 0 (Normal), 1 (Borderline) and 2 (Abnormal) • <u>Items:</u> e.g. worried, unhappy, nervous, scared	5
uestionnaire	Peer problems	Ordinal variable from 0 to 10: Sum of corresponding 5 items of the SDQ rated on the scale 0 (Normal), 1 (Borderline) and 2 (Abnormal) • <u>Items:</u> e.g. solitary, good friends, liked, bullied	5
d Difficulties Q	Hyperactivity- Inattention	Ordinal variable from 0 to 10: Sum of corresponding 5 items of the SDQ rated on the scale 0 (Normal), 1 (Borderline) and 2 (Abnormal) • <u>Items:</u> e.g. restless, squirming, easily distracted	5
Strengths an	Antisocial behaviour	Ordinal variable from 0 to 10: Sum of corresponding 5 items of the SDQ rated on the scale 0 (Normal), 1 (Borderline) and 2 (Abnormal), with all items being reversed as 'abnormal' here refers to high levels of prosociality • Items: e.g. considerate, shares, helpful, kind	5
e (CBQ)	Negative affect	Continuous variable: Average rating of corresponding 12 items of the CBQ rated on a scale from 1 (Extremely untrue) to 7 (Extremely true), with items corresponding to a low negative affect being reversed (e.g. hardly ever complains) • <u>Items:</u> e.g. upset, angry, difficult to sooth, afraid, frustrated	5
emperament / iour Questionnair	Low effortful control	Continuous variable: Average rating of corresponding 12 items of the CBQ rated on a scale from 1 (Extremely untrue) to 7 (Extremely true), with items corresponding to a high effortful control being reversed (e.g. follow instructions) <ul> <li>Items: e.g. follow instructions, becoming involved, prepares</li> </ul>	5
T Child Behavi	Low surgency	Continuous variable: Average rating of corresponding 12 items of the CBQ rated on a scale from 1 (Extremely untrue) to 7 (Extremely true) with items corresponding to a high surgency being reversed (e.g. adventurous)  • Items: e.g. shy, rush into new situations, rough and rowdy, adventurous	5

Figure 11 gives an overview of the distributions of the standardised versions of the outcome variables. The graphs show the kernel densities of the continuous outcome variables and histograms for the categorial outcome variables separately by whether the father participated in GUINZ. It can be seen that in the absence of controlling for other factors, we observe a better language development as well as lower levels of negative affect, effortful control, and emotional problems for children in families in which the father participates in GUINZ, relative to the sample where the father did not participate.





#### Child Behaviour Questionnaire (CBQ)



#### Strengths and Difficulties Questionnaire (SDQ)



Source: GUINZ DCW3, DCW4, DCW5, own calculations and illustrations.

Next, we employ multivariate linear analysis to explore the association between the developmental outcomes D<sub>i</sub> of child i with respect to the paternal involvement variables Y<sub>i</sub>:

- 1. Father participates in GUINZ (binary)
- 2. Involvement in day-to-day care at 9 months (ordinal)<sup>15</sup>
- 3. Paternal leave taking (ordinal)<sup>16</sup>
- 4. Quality care (continuous in points on the constructed quality of care scale)<sup>17</sup>

All outcome variables are assumed to be continuous for the multivariate linear analysis. We thus estimate the following estimation equation:

$$D_i = P(\alpha_i + \beta_1 Y_i + \beta_2 C_i + \beta_3 H H_i + \beta_4 M_i + \beta_5 F_i + \varepsilon_i)$$
<sup>(2)</sup>

With the groups of control variables referring to the following list:

- Child's characteristics (C<sub>i</sub>): twin birth; gender; planned pregnancy; number of siblings; indicators for young siblings (under 5 and under 2); subjective health, developmental problems.
- Household characteristics (HH<sub>i</sub>): household type; household income.
- Mother's socio-economic characteristics (M<sub>i</sub>): marital status; age, migration status; education (ethnicity; employment status / occupation).<sup>18</sup>
- Father's socio-economic characteristics (F<sub>i</sub>): biological father; age; criminal history (DCW1); migration status; ethnicity; education; employment status / occupation; income differences; shiftwork (DCW1); self-employment (DCW2); working hours (DCW1).

# 7.2 Results

Table 7 summarises the estimation results of equation (2) for all nine developmental outcome variables. For the sake of brevity, the table is restricted to the key coefficients of interest, i.e. the paternal involvement variables Y<sub>i</sub> (the full estimation details can be obtained from the authors upon request). At first glance, it is apparent that there are just a few involvement variables that seem to have a significant effect on some of the developmental outcome variables. We can see, for example, that in households where the father participates in GUINZ, the child is observed to have a slightly better language development (speaking on average, ceteris paribus, 0.076 more words) as well as higher levels of effortful control.

<sup>&</sup>lt;sup>15</sup> Ordinal categories of involvement in day-to-day care are: 1 = Not much of the time; 2 = Some of the time; 3 = Most of the time; 4 = All the time.

<sup>&</sup>lt;sup>16</sup> Ordinal categories of paternal leave taking are: 0 = No Leave; 1 = 1 week; 2 = 2 weeks; 3 = between 2 weeks and 1 month; 4 = between 1 month and 2 months; 5 = between 2 months and 3 months; 6 = more than 3 months (incl. still on leave at 9 months).

<sup>&</sup>lt;sup>17</sup> The quality care scale corresponds to the variable constructed for the analysis in research aim 2 (Section 6), i.e. the sum of all activity categories (playing games, playing with toys, talking, telling stories in DCW 1 and DCW 2 and reading books in DCW 1 and DCW 2).

<sup>&</sup>lt;sup>18</sup> Employment status/occupation and ethnicity of the mother are only used as control variables in the estimation of father's participation in GUINZ, where father's characteristics are not available.

With respect to the involvement in the day-to-day care we see a strong positive association of very frequent involvement (all the time) with motor skills and temperament (negative affect and low effortful control) as well as a positive association of involvement in general (some, most or all of the time) with lower levels of behavioural problems (especially peer problems and antisocial behaviour). Although some coefficients for different levels of paternal leave are significant, no clear pattern can be observed.

The clearest pattern evident in Table 7 is in terms of the quality of care scale indicator. We find that higher quality of care positively affects language development and motor skills as well as being associated with a lower risk of emotional problems, peer problems, hyperactivity, antisocial behaviour, and low effortful control. Thus, we can conclude that besides the quantity of care (measured by involvement in day-to-day care) the quality of involvement is highly important for developmental outcomes.

							Child Bel	haviour Ques	tionnaire
			Strengths and Difficulties Questionnaire (SDQ)				(CBQ)		
	Languago	Motor					Nog	Low	Low
	dev.	skills	Emotional	Peer	Hyperact.	Antisoc.	affect	control	surgency
	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se
Father in GUiNZ	0.076**	0.000	-0.026	-0.004	-0.043	-0.006	-0.055	-0.092***	0.006
	(0.032)	(0.036)	(0.033)	(0.032)	(0.034)	(0.035)	(0.036)	(0.035)	(0.037)
Observations	4,443	4,253	4,443	4,443	4,443	4,443	4,253	4,253	4,252
Paternal leave (Ref: None)									
1 week	0.051	-0.054	-0.042	0.005	-0.001	0.165***	-0.160**	0.052	0.027
	(0.054)	(0.063)	(0.059)	(0.058)	(0.060)	(0.064)	(0.066)	(0.063)	(0.068)
2 weeks	0.029	-0.090	-0.016	0.045	-0.057	0.166**	-0.127*	0.047	0.068
	(0.055)	(0.064)	(0.060)	(0.059)	(0.062)	(0.066)	(0.068)	(0.064)	(0.069)
> 2 weeks – 1 month	-0.054	-0.145*	-0.059	-0.031	-0.101	0.116	-0.111	0.075	0.101
	(0.063)	(0.074)	(0.069)	(0.068)	(0.071)	(0.075)	(0.078)	(0.074)	(0.080)
> 1 month – 2 months	0.026	0.088	0.040	0.055	-0.073	0.055	-0.080	0.071	0.169
	(0.091)	(0.105)	(0.099)	(0.097)	(0.102)	(0.108)	(0.111)	(0.105)	(0.113)
> 2 months – 3 months	-0.003	-0.348**	-0.125	0.152	-0.080	0.310*	-0.256	0.517***	0.054
	(0.146)	(0.171)	(0.160)	(0.157)	(0.164)	(0.174)	(0.181)	(0.170)	(0.184)
> 3 months	0.060	0.058	-0.077	-0.030	-0.100	-0.025	-0.337**	0.048	0.052
	(0.136)	(0.159)	(0.149)	(0.145)	(0.152)	(0.162)	(0.167)	(0.158)	(0.170)
Observations	2,539	2,479	2,539	2,539	2,539	2,539	2,480	2,480	2,479
Involvement in day-to-day car	e at DCW1 (F	Ref: Not muc	h)						
Some of the time	-0.022	0.063	-0.088	-0.160**	-0.063	-0.142*	-0.111	-0.124*	-0.059
	(0.063)	(0.071)	(0.068)	(0.067)	(0.070)	(0.073)	(0.076)	(0.071)	(0.077)
Most of the time	0.005	0.102	-0.049	-0.148**	-0.139*	-0.229***	-0.127	-0.114	-0.024
	(0.067)	(0.076)	(0.072)	(0.072)	(0.075)	(0.078)	(0.081)	(0.076)	(0.082)
All the time	-0.098	0.230**	-0.188**	-0.215**	-0.074	-0.234**	-0.251**	-0.233**	-0.072
	(0.083)	(0.095)	(0.090)	(0.089)	(0.093)	(0.097)	(0.101)	(0.094)	(0.102)
Observations	3,098	3,013	3,098	3,098	3,098	3,098	3,014	3,014	3,013
Father's quality care	0.168***	0.071**	-0.057*	-0.103***	-0.121***	-0.065*	-0.047	-0.131***	0.001
	(0.031)	(0.035)	(0.033)	(0.033)	(0.034)	(0.036)	(0.037)	(0.035)	(0.038)
Observations	3,098	3,013	3,098	3,098	3,098	3,098	3,014	3,014	3,013

#### Table 7 - Estimation results – outcome: child's development

*Source:* GUINZ DCW0, DCW1, DCW2, DCW3, DCW4 and DCW5, own calculations and illustrations. *Notes:* \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

As with all the regressions presented, these results show correlation rather than causation. It is, therefore, possible that the positive link between paternal involvement and children's outcomes could be driven by other factors. For example, it may be the case that the level of involvement of mothers and fathers within families is positively correlated (that is, a child with a highly involved father is more likely to also have a highly involved mother). If this is the case, then the measures of paternal involvement may also be indirectly capturing the degree of the mother's involvement, and thus contributing to the positive observed association. Another possibility is that a child who has more carers has better outcomes. This could be due to greater socialisation or simply because these children have a higher total amount of time when an adult is engaged with them. In summary, we cannot rule out the possibility that higher paternal involvement could be proxying for overall higher quantity and quality involvement from other sources.

# 8. LIMITATIONS

Given the breadth of results presented in Sections 5 through to 7, it is important to acknowledge any relevant limitations that must be taken into account, before concluding with key findings in the following section.

First, the findings presented are based on a survey of a selected sample of fathers whose socio-economic characteristics may not be representative of the whole of NZ, for example, GUINZ mothers are on average older than the NZ population and non-European mothers are oversampled (Morton et al., 2013).

Second, and very much in line with the first point, is the possibility that survey data contains measurement errors driven, for example, by conscious and unconscious misreporting. The findings about the fairness of the division of tasks presented in Section 5, for example, show that especially reported levels of involvement are at risk of being biased by overestimation due to the subjective nature of the rating.

Third, as mentioned earlier, the sub-group analysis comparing characteristics of determinants associated with varying levels of paternal involvement and engagement with their children is hampered by small sample size for Māori and Pasifika. This small sample size is partly due to a reasonable share of GUINZ households who did not have a father who participated in GUINZ, with the share of non-participating fathers being higher among Māori, Pasifika and Asian households than NZ European ones. As discussed in Section 6.2.6, this small sample means that our more detailed investigation into ethnic differences is limited to descriptive results, and that even these descriptive results should be interpreted with caution. Moreover, it should be kept in mind that the results on paternal involvement, engagement and children's outcomes relate to fathers who participated in GUINZ rather than being representative of the overall population of NZ families.

The final and most important limitation concerns the causal nature of the estimated associations. Due to a lack of clear exogenous variation in the factors influencing involvement in research aim 2 as well as development outcomes in research aim 3, these results *do not* represent causal relationships. Rather, these findings represent the relationship between 1) family and father characteristics and paternal involvement and 2) paternal involvement and developmental outcomes. Nevertheless, our findings do hopefully pave the way for future research to exploit exogenous variation in involvement in order to identify causal relationships.

# 9. CONCLUSIONS

This study used rich data from the Growing Up in New Zealand study to draw a very detailed picture of the involvement of NZ fathers in their children's upbringing in the first years after birth. The information available allowed us to get an overview of different levels and forms of engagement observed with respect to parental leave taking, direct involvement in day-to-day care and activity indicators that proxied for quality of care. We then explored the internal and external factors driving differences between paternal engagement based on these measures; and by virtue of the longitudinal nature of the data, assessed how different levels of engagement were associated with children's developmental outcomes at later time points.

Prior to undertaking the empirical analysis in this study, we summarised the relevant policy background and current evidence on the gender care gap in NZ. For example, we know that while parental leave entitlements sit with the mother, she can transfer all or part of her entitlements to her spouse or de facto partner – however the uptake of paid parental leave (PPL) by fathers in NZ is less than 1%. Further, while a number of OECD countries have begun to try and incentivise paternal leave taking with the introduction of earmarked leave for fathers (daddy months), NZ is one of just six OECD countries with PPL that has not moved in this direction. Available data on time spent in paid and unpaid work also illustrated that NZ has a negative gender work gap, driven largely by fathers spending more time in paid work. This may be a signal of unequal labour market patterns driving the gender care gap, more so than underlying intra-household gender norms.

To gain a snapshot of fathers' engagement during the early years of their children's lives we first compared indicators of direct involvement relative to the mothers in the sample. Fathers were distinctly less involved than mothers, although involvement levels did increase when the child was aged between 9 months and 2 years. Interestingly, fathers strongly overestimate their anticipated involvement and direct responsibility in childcare based on antenatal interviews. A similar pattern between fathers, mothers and over time is evident for a range of activities intended to capture quality of care (e.g. playing games, reading books, etc). In terms of leave taking we find that for more than half of the employed sample, anticipated leave is shorter than the preferred amount; and the actual leave taken is less than the anticipated level. The most common reason for taking no / shorter leave was professional or work commitments, followed by financial reasons. Additionally, we find that parental leave taking has no significant link to direct involvement within the household and is only weakly linked to quality of care, indicating an important difference between parental leave taking and quantity and quality of direct involvement.

In our second research aim we analysed the internal and external determinants of the above-discussed levels of paternal involvement. Unsurprisingly, high childcare involvement (defined as father being involved at least most of the time in day-to-day care of the child) is negatively associated with several work-related factors reflecting more time spent in paid employment such as self-employment; full-time employment; and overtime work. Further, the likelihood of high childcare involvement is reduced the higher the income difference between the father and mother. This is a potential indication of differences in the distribution of bargaining power in driving the division in unpaid labour, even when work and education related factors are controlled for.

We capture quality of care with a binary indicator defined as above-median amount of activities with the child, and find a range of factors are associated with this outcome. For example, more siblings means a lower likelihood of high quality of care; while the helpfulness rating of the family and relationship quality between father and mother are associated with a greater likelihood of fathers interacting with their children in ways that reflect high quality of care.

A novel contribution of this research is to delve into the psycho-social characteristics and personality traits as mechanisms behind NZ fathers' involvement with their children. We find that conscientiousness is an important determinant of day-to-day care; while extraversion and openness of the father is associated with a greater likelihood of high quality of care. Perceived influence, parental confidence and parental satisfaction are also strongly positively associated with high childcare involvement in general

There are clear ethnic differences apparent in this research. Māori and Pasifika fathers in general undertake higher childcare involvement and Pasifika fathers are found to be more likely to undertake high quality of care activities, all relative to their NZ European counterparts.

Previous studies often do not have good data on paternal involvement and often have to proxy this with the amount of leave taken. Importantly, we find that paternal leave taking is far from being a good proxy for actual involvement of fathers especially if one is interested in determinants driving the behaviour. Paternal leave taking is strongly driven by occupational factors such as internal substitutionability in the firm (e.g. managers and self-employed fathers) as well as negatively associated with existing children in the household. While there are no ethnic differences in leave taking, there were clear differences by ethnicity in day-to-day care and involvement in general.

In the final research aim of this study we undertook exploratory analysis of the association between paternal involvement in the child's early years and children's cognitive, physical and psychological development at later time points. The findings show that no clear associations between paternal leave taking and developmental outcomes can be found. As opposed to this, we do see an important positive relationship between involvement in day-to-day care and especially psychological outcomes such as lower levels of peer and emotional problems and antisocial behaviour as well as higher levels of effortful control. Nevertheless, the clearest pattern evident in Section 7 is in terms of the quality of care scale indicator. We find that higher quality of care positively affects language development, motor skills and most psychological outcomes (e.g. lower risk of emotional problems, peer problems, hyperactivity / inattention, antisocial behaviour, and low effortful control). Thus, we concluded that besides the quantity of direct involvement, the quality of involvement is highly important for developmental outcomes.

# REFERENCES

- Akerlof, George A., and Rachel E. Kranton. 2000. "Economics and Identity." *The Quarterly Journal of Economics* 115 (3): 715–53.
- Allen, Sarah, and Alan J. Hawkins. 1999. "Maternal Gatekeeping: Mothers' Beliefs and Behaviors That Inhibit Greater Father Involvement in Family Work." *Journal of Marriage and the Family* 61 (1): 199. https://doi.org/10.2307/353894.
- Allen, Tammy D. 2012. "The Work-Family Role Interface: A Synthesis of the Research from Industrial and Organizational Psychology." In *Handbook of Psychology, Second Edition*, edited by Irving B. Weiner. John Wiley & Sons, Inc. https://doi.org/10.1002/9781118133880.hop212026.
- Anderson, Deborah J., Melissa Binder, and Kate Krause. 2002. "The Motherhood Wage Penalty: Which Mothers Pay It and Why?" *The American Economic Review* 92 (2): 354–58.
- Barrett, Jennifer, and Alison S. Fleming. 2011. "Annual Research Review: All Mothers Are Not Created Equal: Neural and Psychobiological Perspectives on Mothering and the Importance of Individual Differences." *Journal of Child Psychology and Psychiatry* 52 (4): 368–97.
- Bartel, Ann P., Maya Rossin-Slater, Christopher J. Ruhm, Jenna Stearns, and Jane Waldfogel. 2018. "Paid Family Leave, Fathers' Leave-Taking, and Leave-Sharing in Dual-Earner Households: Paid Family Leave and Fathers' Leave-Taking." *Journal of Policy Analysis and Management* 37 (1): 10–37. https://doi.org/10.1002/pam.22030.
- Becker, Gary S. 1981. *A Treatise on the Family*. Cambridge, MA: Harvard University Press. https://doi.org/10.2307/351313.
- Bertrand, Marianne, Claudia Goldin, and Lawrence F Katz. 2010. "Dynamics of the Gender Gap for Young Professionals in the Financial and Corporate Sectors." *American Economic Journal: Applied Economics* 2 (3): 228–55. https://doi.org/10.1257/app.2.3.228.
- Bertrand, Marianne, Jessica Pan, and Emir Kamenica. 2015. "Gender Identity and Relative Income within Households." *The Quarterly Journal of Economics* 130 (2): 571–614.
- Bianchi, Suzanne M. 2000. "Maternal Employment and Time with Children: Dramatic Change or Surprising Continuity?" *Demography* 37 (4): 401–14.
- Birkett, Holly, and Sarah Forbes. 2019. "Where's Dad? Exploring the Low Take-up of Inclusive ParentingPoliciesintheUK."PolicyStudies40(2):205–24.https://doi.org/10.1080/01442872.2019.1581160.
- Blau, Francine D., and Lawrence M. Kahn. 2007. "Changes in the Labor Supply Behavior of Married Women: 1980–2000." *Journal of Labor Economics* 25 (3): 393–438.
- ———. 2017. "The Gender Wage Gap: Extent, Trends, and Explanations." *Journal of Economic Literature* 55 (3): 789–865.

Brady, Michelle, Emily Stevens, Laetitia Coles, Maria Zadoroznyj, and Bill Martin. 2017. "'You Can Spend

Time. . .But Not Necessarily Be Bonding with Them': Australian Fathers' Constructions and Enactments of Infant Bonding." *Journal of Social Policy* 46 (1): 69–90. https://doi.org/10.1017/S0047279416000374.

- Brandth, Berit, and Elin Kvande. 2019. "Workplace Support of Fathers' Parental Leave Use in Norway." *Community, Work & Family* 22 (1): 43–57. https://doi.org/10.1080/13668803.2018.1472067.
- Budig, Michelle J., and Paula England. 2001. "The Wage Penalty for Motherhood." *American Sociological Review* 66 (2): 204–25. https://doi.org/10.2307/2657415.
- Bütikofer, Aline, Sissel Jensen, and Kjell G. Salvanes. 2018. "The Role of Parenthood on the Gender Gap among Top Earners." *European Economic Review* 109 (October): 103–23. https://doi.org/10.1016/j.euroecorev.2018.05.008.
- Bygren, Magnus, and Ann-Zofie Duvander. 2006. "Parents' Workplace Situation and Fathers' Parental Leave Use." *Journal of Marriage and Family* 68 (2): 363–72. https://doi.org/10.1111/j.1741-3737.2006.00258.x.
- Callister, Dr Paul. 2005. "New Zealand Fathers: Overworked, Undervalued, and Overseas?" In .
- Cools, Sara, Jon H. Fiva, and Lars J. Kirkebøen. 2015. "Causal Effects of Paternity Leave on Children and Parents." *The Scandinavian Journal of Economics* 117 (3): 801–28. https://doi.org/10.1111/sjoe.12113.
- Cunningham, Mick. 2007. "Influences of Women's Employment on the Gendered Division of Household Labor Over the Life Course: Evidence From a 31-Year Panel Study." *Journal of Family Issues* 28 (3): 422–44. https://doi.org/10.1177/0192513X06295198.
- Dahl, Gordon B., Katrine V. Løken, and Magne Mogstad. 2014. "Peer Effects in Program Participation." *American Economic Review* 104 (7): 2049–74. https://doi.org/10.1257/aer.104.7.2049.
- Dixon, Sylvia. 2000. "Pay Inequality between Men and Women in New Zealand." New Zealand Department of Labour Occasional Paper.
- Druedahl, Jeppe, Mette Ejrnæs, and Thomas H. Jørgensen. 2019. "Earmarked Paternity Leave and the Relative Income within Couples." *Economics Letters* 180 (July): 85–88. https://doi.org/10.1016/j.econlet.2019.04.018.
- Ekberg, John, Rickard Eriksson, and Guido Friebel. 2013. "Parental Leave A Policy Evaluation of the Swedish 'Daddy-Month' Reform." *Journal of Public Economics* 97 (January): 131–43. https://doi.org/10.1016/j.jpubeco.2012.09.001.
- El Nokali, Nermeen E., Heather J. Bachman, and Elizabeth Votruba-Drzal. 2010. "Parent Involvement and Children's Academic and Social Development in Elementary School." *Child Development* 81 (3): 988–1005. https://doi.org/10.1111/j.1467-8624.2010.01447.x.
- Eriksson, Rickard. 2005. Parental Leave in Sweden: The Effects of the Second Daddy Month. http://urn.kb.se/resolve?urn=urn:nbn:se:su:diva-72367.

- Farré, Lídia, and Libertad González. 2019. "Does Paternity Leave Reduce Fertility?" *Journal of Public Economics* 172: 52–66.
- Gangl, Markus, and Andrea Ziefle. 2009. "Motherhood, Labor Force Behavior, and Women's Careers: An Empirical Assessment of the Wage Penalty for Motherhood in Britain, Germany, and the United States." *Demography* 46 (2): 341–69.
- Geisler, Esther, and Michaela Kreyenfeld. 2019. "Policy Reform and Fathers' Use of Parental Leave in Germany: The Role of Education and Workplace Characteristics." *Journal of European Social Policy* 29 (2): 273–91. https://doi.org/10.1177/0958928718765638.
- Goodman, Robert. 1997. "The Strengths and Difficulties Questionnaire: A Research Note." *Journal of Child Psychology and Psychiatry* 38 (5): 581–86.
- Gough, Margaret, and Mary Noonan. 2013. "A Review of the Motherhood Wage Penalty in the United StatesMargaret Gough, Mary Noonan" 7 (4): 328–42.
- Growing Up in New Zealand. 2017. "Who Are Today's Dads? Fathers and Co-Parents of Children in the Growing Up in New Zealand Study." Centre for Longitudinal Research, University of Auckland.

----. 2018. "External Data Release - Data User Guide, DCW6."

- Haas, Linda, and Tine Rostgaard. 2011. "Fathers' Rights to Paid Parental Leave in the Nordic Countries: Consequences for the Gendered Division of Leave." *Community, Work & Family* 14 (2): 177–95.
- Heim, Bradley T. 2007. "The Incredible Shrinking Elasticities: Married Female Labor Supply, 1978 2002." J. Human Resources XLII (January 2007): 881–918. https://doi.org/10.3368/jhr.XLII.4.881.
- Hennecke, Juliane, and Astrid Pape. 2020. "Suddenly a Stay-at-Home Dad? Short- and Long-Term Consequences of Fathers' Job Loss on Time Investment in the Household." *IZA Discussion Paper* 13866. www.iza.org.
- Hook, Jennifer L. 2010. "Gender Inequality in the Welfare State: Sex Segregation in Housework, 1965–2003." *American Journal of Sociology* 115 (5): 1480–1523. https://doi.org/10.1086/651384.
- Huebener, Mathias, Jonas Jessen, Daniel Kühnle, and Michael Oberfichtner. 2021. "A Firm-Side Perspective on Parental Leave." *SSRN Electronic Journal*. https://doi.org/10.2139/ssrn.3874350.
- Juhn, Chinhui, and Kevin M. Murphy. 1997. "Wage Inequality and Family Labor Supply." *Journal of Labor Economics* 15 (1): 72–97. https://doi.org/10.1086/209847.
- Kotsadam, Andreas, and Henning Finseraas. 2011. "The State Intervenes in the Battle of the Sexes: Causal Effects of Paternity Leave." *Social Science Research* 40 (6): 1611–22.

Luketina, Francis, Carl Davidson, and Penny Palmer. 2009. "Supporting Kiwi Dads: Roles and Needs of

New Zealand Fathers." 5/09. Families Commission.

- Mangiavacchi, Lucia, Luca Piccoli, and Luca Pieroni. 2021. "Fathers Matter: Intrahousehold Responsibilities and Children's Wellbeing during the COVID-19 Lockdown in Italy." *Economics & Human Biology* 42 (August): 101016. https://doi.org/10.1016/j.ehb.2021.101016.
- Meehan, Lisa. forthcoming. "Paid Parental Leave, Maternal Labour Supply and Children's Outcomes in New Zealand."
- Mercante, Joseph, and Penny Mok. 2014. "Working for Families Changes: The Effect on Labour Supply in New Zealand." 14/18. Treasury Working Paper. https://www.treasury.govt.nz/publications/wp/working-families-changes-effect-labour-supplynew-zealand-wp-14-18-html.
- Moberg, Ylva, and Maaike Van der Vleuten. 2021. "Why Do Gendered Divisions of Labour Persist? Parental Leave Take-up among Adoptive and Biological Parents." Presentation given on the workshop "Frontiers in parental leave research: Evidence from the Nordic countries." Presentation given on the workshop "Frontiers in parental leave research: Evidence from the Nordic countries" at Stockholm University.
- Morrissey, Suzy. 2020. "Questioning policy-making as problem-solving. A Bacchian examination of how paid parental leave was problematized in New Zealand and Norway." Doctoral Thesis, Victoria University of Wellington.
- Morton, Susan M B, Polly Atatoa Carr, Dinusha K Bandara, Cameron C Grant, Vivienne C Ivory, Te K. R Kingi, Renee Liang, et al. 2010. *Before We Are Born*. https://cdn.auckland.ac.nz/assets/growingup/research-findings-impact/report01.pdf.
- Morton, Susan M B, Polly Atatoa Carr, Cameron C Grant, Elizabeth M Robinson, Dinusha K Bandara, Amy Bird, Vivienne C Ivory, et al. 2013. "Cohort Profile: Growing Up in New Zealand." *International Journal of Epidemiology* 42 (1): 65–75. https://doi.org/10.1093/ije/dyr206.
- Morton, Susan M B, Caroline G Walker, Sarah Gerriten, Ashley Smith, Jane Cha, Amy Bird, Pat Bullen, et al. 2020. "Now We Are Eight: Key Findings from Growing Up in New Zealand."
- Noy, Shakked, and Isabelle Sin. 2021. "The Drivers of Mothers' Parental Leave Decisions Evidence from the Growing Up in New Zealand Longitudinal Survey."

OECD. 2020. "Net Childcare Costs." 2020. https://data.oecd.org/benwage/net-childcare-costs.htm.

----. 2021. "OECD.Stats - Employment." 2021. https://stats.oecd.org/.

- OECD Family Database. 2021. "Parental Leave Systems." 2021. https://www.oecd.org/els/soc/PF2\_1\_Parental\_leave\_systems.pdf.
- Patnaik, Ankita. 2019. "Reserving Time for Daddy: The Consequences of Fathers' Quotas." *Journal of Labor Economics* 37 (4): 1009–59. https://doi.org/10.1086/703115.

Pryor, Jan, Susan Morton, Dinusha Bandara, Elizabeth Robinson, and Cameron Grant. 2014. "Pregnant

Partners: Fathers of the Growing Up in New Zealand Children." Journal of Family Studies 20 (1): 15.

- Putnam, Samuel P., and Mary K. Rothbart. 2006. "Development of Short and Very Short Forms of the Children's Behavior Questionnaire." *Journal of Personality Assessment* 87 (1): 102–12.
- Reilly, Amanda, and Suzy Morrissey. 2017. "Why New Zealand Should Introduce Paid "Dad and Partner Leave"." *New Zealand Journal of Employment Relations* 41 (3): 66–75.
- Samtleben, Bringmann, Bünning, and Hipp. 2019. "What Helps and What Hinders? Exploring the Role of Workplace Characteristics for Parental Leave Use and Its Career Consequences." Social Sciences 8 (10): 270. https://doi.org/10.3390/socsci8100270.
- Samtleben, Claire. 2019. "Also on Sundays, Women Perform Most of the Housework and Child Care." *DIW Weekly Report*. https://doi.org/10.18723/DIW\_DWR:2019-10-2.
- Samtleben, Claire, Clara Schaeper, and Katharina Wrohlich. 2019. "Elterngeld Und Elterngeld Plus: Nutzung Durch Väter Gestiegen, Aufteilung Zwischen Müttern Und Vätern Aber Noch Sehr Ungleich." *DIW Wochenbericht* 86 (35): 607–13.
- Sanchez, Laura, and Elizabeth Thomas. 1997. "Becoming Mothers and Fathers Parenthood, Gender and Division of Labor." *Gender and Society* 11 (6): 747–72.
- Sigurdardottir, Heida Maria, and Ólöf Garðarsdóttir. 2018. "Backlash in Gender Equality? Fathers' Parental Leave during a Time of Economic Crisis." *Journal of European Social Policy* 28 (4): 342–56.
- Sin, Isabelle, Kabir Dasgupta, and Gail Pacheco. 2018. "Parenthood and Labour Market Outcomes." Ministry of Women, New Zealand.
- Stats NZ. 2017. "Effect of Motherhood on Pay Methodology and Full Results." Stats NZ. https://www.stats.govt.nz/reports/effect-of-motherhood-on-pay-methodology-and-full-results.
- Tamm, Marcus. 2019. "Fathers' Parental Leave-Taking, Childcare Involvement and Labor Market Participation." *Labour Economics*, Special Issue on "European Association of Labour Economists, 30th annual conference, Lyon, France, 13-15 September 2018, 59 (August): 184–97. https://doi.org/10.1016/j.labeco.2019.04.007.
- Tanaka, Sakiko, and Jane Waldfogel. 2007. "Effects of Parental Leave and Work Hours on Fathers' Involvement with Their Babies." *Community, Work & Family* 10 (4): 409–26. https://doi.org/10.1080/13668800701575069.
- Wilner, Lionel. 2016. "Worker-Firm Matching and the Parenthood Pay Gap: Evidence from Linked Employer-Employee Data." *Journal of Population Economics* 29 (4): 991–1023. https://doi.org/10.1007/s00148-016-0597-9.

# APPENDIX

### Table A.1 - Descriptive statistics by father availability

	Father in GUINZ			
	No	Yes		
Observations	1,345	3,098		
CHILDREN AND PREGNANCY CHARACTERISTICS				
Is twin	0.02	0.03		
Is girl	0.49	0.49		
Pregnancy planned	0.52	0.75	***	
Number of siblings				
None	0.31	0.40	***	
One	0.35	0.37		
Two	0.17	0.15	**	
Three	0.09	0.05	* * *	
Four or more	0.08	0.03	***	
Youngest sibling under 5	0.51	0.45	***	
Youngest sibling under 2	0.15	0.12	***	
Child's subjective health				
Excellent	0.57	0.62	***	
Very good	0.30	0.27	**	
Good	0.11	0.08	***	
Fair or Poor	0.03	0.03		
Child has developmental problem	0.09	0.10		
External care (at age 2)				
None	0.52	0.41	***	
ECE (incl. Kindergarten Kohanga Reo and PI ECE)	0.32	0.38	***	
Home-base care	0.04	0.07	***	
Nanny	0.02	0.04	***	
Grandparents or other relatives	0.09	0.08		
Other (friends, neighbours, community center)	0.02	0.02		
HOUSEHOLD CHARACTERISTICS				
HH type – Extended family	0.37	0.19	***	
Household income				
Less than 20K	0.06	0.01	***	
20-30K	0.08	0.03	***	
30-50K	0.18	0.10	***	
50-70К	0.19	0.15	***	
70-100К	0.23	0.25		
100-150K	0.17	0.27	***	
More than 150K	0.10	0.19	***	
MOTHER CHARACTERISTICS				
Mother is married	0.57	0.75	***	

Mo	ther is cohabiting	0.94	0.99	***
Mo	ther's age			
	<25 years	0.24	0.09	***
	25-29 years	0.25	0.24	
	30-34 years	0.29	0.37	***
	35-39 years	0.18	0.25	***
	40+ years	0.04	0.04	
Mo	ther born in NZ	0.65	0.68	**
Mo	ther's prioritised ethnicity			
	NZ European	0.41	0.66	***
	Māori	0.22	0.12	***
	Pasifika	0.20	0.07	***
	Asian	0.16	0.13	**
	MELAA or Other	0.02	0.03	
Mo	ther's education			
	No sec school qualification	0.08	0.03	***
	Secondary school/NCEA 1-4	0.25	0.19	***
	Diploma/Trade cert/NCEA 5-6	0.37	0.27	***
	Bachelors degree	0.18	0.29	***
	Higher degree	0.11	0.22	***
Mo	ther's occupation			
	Does not apply (not employed)	0.51	0.34	***
	Managers	0.05	0.08	***
	Professionals	0.19	0.34	***
	Clerical and administrative workers	0.03	0.02	
	Technicians and trades workers	0.05	0.04	*
	Machinery operators and drivers	0.10	0.12	*
	Community and personal service workers	0.04	0.04	
	Sales workers	0.01	0.00	***
	Labourers	0.03	0.02	*

Source: GUINZ DCW0, DCW1 and DCW2, own calculations and illustrations.

*Notes:* Significant stats in column 3 refer to the significance levels of a t-test for mean equality between columns 2 and 3 with \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

Variable	Description	Categories	Mean
CHILDREN AND PREG	NANCY CHARACTERISTICS		
ls twin	Dummy variable: 1 if child is a twin, 0 if not		0.02
Is girl	Dummy variable: 1 if child is a girl, 0 if a boy		0.49
Pregnancy planned	Dummy variable: 1 if pregnancy was planned, 0 if		0.68
	not.		
Number of siblings	Categorial variable: Calculated number of siblings at	Non	0.37
	birth based on the reported years of birth in the	One	0.36
	phone interview at 16 months, i.e. reported	Two	0.16
	number of siblings minus number of siblings born	Three	0.06
	after the child.	Four or more	0.05
Youngest sibling	Dummy variable: 1 if child had a sibling under 5 at		0.47
under 5	time of birth, 0 if not.		
Youngest sibling	Dummy variable: 1 if child had a sibling under 2 at		0.13
under 2	time of birth, 0 if not.		
Developmental	Dummy variable: 1 if child has any reported		0.10
problems (DCW1)	developmental problems, 0 if not		
Childs subjective	Categorial variable: Parental report about their child	Excellent	0.60
health (DCW1)	health. 1 if child is in excellent health, 2 if child is in	Very Good	0.28
	very good health, 3 if child is in good health, 4 if	Good	0.09
	child is in fair or poor health.	Fair or Poor	0.03
HOUSEHOLD CHARAC	TERISTICS		
HH type – Extended	Dummy variable: 1 if reported household included		0.25
family	other adults other than parents, 0 if not.		
Household income	Categorial variable: Reported household income in	Less than 20K	0.03
	bands.	20-30K	0.04
		30-50K	0.12
		50-70K	0.16
		70-100K	0.24
		100-150K	0.24
		More than 150K	0.16
Partnership quality <sup>a</sup>	Latent continuous variable (standardised): Quality		0.03
	of the relationship based on factor analysis of a list		
	of 15 situations in the partnership rated on a scale		
	from 1 (All the time) to 6 (Almost never).		
	Examples: Get angry with each other, Let each		
	other know you really care about each other		
	(reversed).		
Family helpfulness <sup>a</sup>	Continuous variable (standardised): Average rating		0.01
	of the helpfulness of own parents, partner's parents		

Table A.2 - Variable overview and summary statistics – children, household and mothers (all observations)

as well as own and partners extended family on a

scale from 1 (Not available) to 6 (Extremely helpful).

MOTHER'S CHARACTE	RISTICS		
Mother is married	Dummy variable: 1 if mother is married or in a civil		0.70
	union, 0 if not (missing if no partner)		
Mother is	Dummy variable: 1 if mother reports cohabiting		0.98
cohabiting with	with her partner, 0 if not (missing if no partner)		
partner			
Mother's age	Categorial variable: Reported age grouped into 5-	<25 years	0.14
	year bands.	25-29 years	0.24
		30-34 years	0.35
		35-39 years	0.23
		40+ years	0.04
Mother born in NZ	Dummy variable: 1 if mother reports being born in		0.67
	New Zealand, 0 if not.		
Mother's prioritised	Categorial variable: - Respondents are allocated a		
ethnicity	single ethnicity where the order of priority is Māori,	NZ European	0.58
	Pacific Peoples, Asian, MELAA and Other, NZ	Māori	0.15
	European.	Pasifika	0.10
		Asian	0.14
		MELAA	0.02
Mother's education	Categorial variable: Reported highest level of	No sec school qualification	0.04
	education of the mother.	Sec school/NCEA 1-4	0.21
		Diploma/Trade cert/NCEA 5-6	0.30
		Bachelor's degree	0.26
		Higher degree	0.19
Mother's	Categorial variable: Reported occupation of the	Does not apply (not employed)	0.39
occupation	mother if she reports to be employed.	Managers	0.07
		Professionals	0.30
		Clerical and administrative	0.02
		Workers	
		Technicians and trades workers	0.04
		Machinery operators and drivers	0.12
		Community and personal service	0.04
		Workers	
		Sales workers	0.00
		Labourers	0.02

Source: GUINZ DCW0, DCW1 and DCW2, own calculations and illustrations.

*Notes:* <sup>a</sup> Variables on partnership quality and helpfulness of family come from partner questionnaire and are thus only available for the sample of permanent fathers.

Variable	Description	Categories	Mean
SOCIO-ECONOMIC CH	ARACTERISTICS		
Partner is biological	Dummy variable: 1 if fathers reports to be the		0.99
father	biological father of the baby, 0 if not.		
Father's age	Categorial variable: Reported age grouped into 5-	<25 years	0.05
	year bands.	25-29 years	0.19
		30-34 years	0.33
		35-39 years	0.28
		40+ years	0.15
Father has criminal	Dummy variable: 1 if father reports having a		0.02
history (DCW1)	criminal history, 0 if not.		
Father born in NZ	Dummy variable: 1 if father reports being born in		0.69
	New Zealand, 0 if not.		
Father migrated as a	Dummy variable: 1 if migration year < birth year +		0.02
kid (if migrant)	18, 0 if not, missing if father born in NZ.		
Father's prioritised	Categorial variable: Respondents are allocated a	NZ European	0.65
ethnicity	single ethnicity where the order of priority is Māori,	Māori	0.13
	Pacific Peoples, Asian, MELAA, Other, NZ European.	Pasifika	0.08
		Asian	0.11
		MELAA	0.02
		Other	0.65
Father's education	Categorial variable: Reported highest level of	No sec school qualification	0.05
	education of the father.	Sec school/NCEA 1-4	0.18
		Diploma/Trade cert/NCEA 5-6	0.37
		Bachelors degree	0.21
		Higher degree	0.19
Father's occupation	Categorial variable: Reported occupation of the	Does not apply / not employed	0.17
	father if he reports to be employed.	Managers	0.18
		Professionals	0.29
		Clerical and administrative	0.16
		Workers	
		Technicians and trades workers	0.03
		Machinery operators and	0.04
		drivers	
		Community and personal	0.04
		service workers	
		Sales workers	0.05
		Labourers	0.05
Income difference	Continuous variable: Difference between father's		2.96
in \$1000	income and mother's income in \$1000. Calculated		
	by setting income equal to the average income		
	within an income band, e.g. 10K for <20K, 25K for		

Table A 🤅	3 - Variable	overview and	summarv	statistics -	fathers	(sample)	permanent f	<sup>:</sup> athers'
			Juilling	Julia	rutiters	Junpic.	permanent	utituty

	20-30K, 40K for 30-50K, 60K for 50-70K, 85K for 70-		
	100K, 125K for 100-150K and 175K for >150K.		
Shiftwork (DCW1)	Dummy variable: 0 if father reports working a		0.20
	regular daily schedule, 1 if not.		
Self-employed	Dummy variable: 1 if father reports being self-		0.20
(DCW2)	employed, 0 if not.		
Fulltime (DCW1)	Dummy variable: 1 if father reports weekly actual		0.84
	working hours of 40 hours or more, 0 if not.		
Overtime (DCW1)	Dummy variable: 1 if father reports weekly working		0.60
	hours of more than 40 hours, 0 if not.		
HEALTH			
Mental health	Dummy variable: 1 if father reports having		0.11
	depression and/or anxiety/panic attacks diagnosed		
	ever at any time in his life, 0 if not.		
Physical health	Dummy variable: 1 if father reports having a		0.32
	disability, asthma, a heart disease/high blood		
	pressure or diabetes diagnosed ever at any time in		
	his life, 0 if not.		
Subjective health	Dummy variable: 1 if father reports having a good,		0.88
	very good or excellent health, 0 if not.		
PERSONALITY			
Big Five personality	Five continuous variables: Average rating of a	Extraversion (E)	3.54
inventory	number of 8 statements/items per variable on a	Agreeableness (A)	4.00
	scale from 1 (Disagree strongly) to 5 (Agree	Conscientiousness (C)	3.99
	strongly).	Neuroticism (N)	2.21
	Example: I see myself as someone who 1. Is talkative (E)	Openness (O)	3.98
	2. Tends to find fault with others (A)		
	3. Does things carefully and completely (C)		
	4. Is depressed, down (N)		
	5. Is original, comes up with new ideas (O)		
Perceived stress	Latent continuous variable: Predicted from factor		-0.05
scale	analysis based on the father's evaluation of a list of		
	10 situations on a scale from 1 (Never) to 5 (Very		
	often).		
	Example: In the last four weeks, how often have you		
	been upset because of something that happened		
	unexpectedly?		
PSYCHO-SOCIAL CHA	RACTERISTICS (PARENTAL IDENTITY)		
Work-life balance	Dummy variable: 1 if father has a higher than mean		0.49
(DCW1)	self-assessed importance of a work-life balance (as		
	a latent continuous measure predicted from factor		
	analysis based on father's evaluation of		

	8 statements on a scale from 1 (Strongly disagree)	
	to 7 (Strongly agree).	
	Example: Having both work and family	
	responsibilities makes me a more well-rounded	
	person.	
Perceived influence	Dummy variable: 1 if father thinks he positively	0.84
(DCW1)	affects the baby's development based a reported 5	
	or 6 on a scale from 1 (Not at all) to 6 (A great deal),	
	0 if not.	
Parental identity	Dummy variable: 1 if father reports "being a	0.98
(DCW2)	parent" is an important part of who he is, 0 if not.	
Parental confidence	Dummy variable: 1 if father reports he is confident	0.86
(DCW1)	when caring for his baby based a reported 5 or 6 on	
	a scale from 1 (Not at all confident) to 6	
	(Completely confident), 0 if not.	
Good parent	Dummy variable: 1 if father reports he is a better	0.81
(DCW2)	than average or very good parent, 0 if not.	
Parental satisfaction	Continuous variable: Average rating of 3 statements	5.22
(DCW1)	on a scale from 1 (Strongly disagree) to 5 (Strongly	
	agree).	
	Statements: On the whole, I enjoy being a parent,	
	Being a parent is very satisfying, On the whole, it's	
	good to be a parent.	

Source: GUINZ DCW0, DCW1 and DCW2, own calculations and illustrations.

	(1)	(2)
	b/se	b/se
Is twin	0.045	0.015
	(0.044)	(0.042)
ls girl	-0.009	-0.006
	(0.013)	(0.013)
Pregnancy planned	0.113***	0.048***
	(0.016)	(0.016)
Number of siblings (Ref: None)		
One	-0.057***	-0.062***
	(0.020)	(0.020)
Two	-0.067***	-0.081***
	(0.024)	(0.024)
Three	-0.128***	-0.123***
	(0.032)	(0.033)
Four or more	-0.162***	-0.141***
	(0.037)	(0.039)
Youngest sibling under 5	0.014	-0.005
	(0.019)	(0.019)
Youngest sibling under 2	0.009	0.014
	(0.021)	(0.021)
Child's subjective health (Ref: Excellent)		
Very good	-0.024	-0.018
	(0.015)	(0.015)
Good	-0.033	-0.022
	(0.024)	(0.023)
Fair	0.022	0.038
	(0.037)	(0.035)
Child has developmental problems	0.000	-0.013
	(0.023)	(0.022)
HH type = Extended family	-0.119***	-0.040**
	(0.014)	(0.016)
Household income (Ref: Less than 20K)		
20-30K	0.077	0.046
	(0.057)	(0.055)
30-50K	0.180***	0.111**
	(0.049)	(0.048)
50-70K	0.229***	0.136***
	(0.048)	(0.047)
70-100К	0.278***	0.143***
	(0.047)	(0.047)
100-150K	0.329***	0.173***

# Table A.4 - Full estimation results - outcome: GUINZ participation

	(0.047)	(0.048)
More than 150K	0.350***	0.180***
	(0.048)	(0.050)
Mother is married		0.085***
		(0.015)
Mother is cohabiting with partner		0.222***
		(0.047)
Mother's age (Ref: <25 years)		ref.
25-29 years		0.070***
		(0.024)
30-34 years		0.068***
		(0.025)
35-39 years		0.094***
		(0.027)
40+ years		0.089**
		(0.039)
Mother born in NZ		0.013
		(0.018)
Mother's prioritised ethnicity (Ref: NZ European)		
Māori		-0.067***
		(0.021)
Pasifika		-0.173***
		(0.027)
Asian		-0.101***
		(0.026)
MELAA or other		-0.004
		(0.043)
Mother's education (Ref: No sec qualification)		
Sec school/NCEA 1-4		0.064*
		(0.035)
Diploma/Trade cert/NCEA 5-6		0.031
		(0.035)
Bachelors degree		0.085**
		(0.037)
Higher degree		0.099**
		(0.039)
Mother's occupation (Ref: Professional or admin. worker)		
Not employed		-0.036**
		(0.016)
Managers		-0.027
		(0.029)
Technicians/ trades workers/ machinery operators		-0.097**
		(0.042)

Other		-0.002
		(0.022)
Observations	4,443	4,443
Pseudo R-squared	0.089	0.126

Source: GUINZ DCW0, DCW1 and DCW2, own calculations and illustrations.

*Notes:* \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

Table A.5 - Full	estimation	results -	outcome:	involve	ement in	day-to-day care
------------------	------------	-----------	----------	---------	----------	-----------------

			DCW1		DCW2				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se	
Is twin	-0.029	-0.015	-0.019	-0.009	0.156***	0.179***	0.179***	0.186***	
	(0.054)	(0.053)	(0.053)	(0.053)	(0.056)	(0.056)	(0.056)	(0.055)	
Is girl	-0.013	-0.016	-0.013	-0.013	-0.001	-0.000	0.003	0.003	
	(0.017)	(0.016)	(0.016)	(0.016)	(0.018)	(0.017)	(0.017)	(0.017)	
Pregnancy	-0.048**	-0.027	-0.027	-0.036*	-0.028	-0.009	-0.007	-0.015	
planned	(0.022)	(0.021)	(0.021)	(0.021)	(0.022)	(0.022)	(0.022)	(0.022)	
Number of siblings	(Ref: None)								
One	0.001	-0.005	-0.001	-0.001	-0.004	0.001	0.010	0.011	
	(0.028)	(0.027)	(0.027)	(0.027)	(0.030)	(0.029)	(0.029)	(0.029)	
Two	-0.034	-0.031	-0.024	-0.025	-0.031	-0.019	-0.004	-0.003	
	(0.033)	(0.033)	(0.033)	(0.032)	(0.035)	(0.035)	(0.035)	(0.035)	
Three	0.098**	0.050	0.051	0.028	-0.021	-0.047	-0.038	-0.048	
	(0.047)	(0.046)	(0.046)	(0.044)	(0.048)	(0.048)	(0.048)	(0.047)	
Four or more	-0.000	-0.067	-0.052	-0.069	0.084	0.046	0.065	0.051	
	(0.056)	(0.052)	(0.053)	(0.051)	(0.059)	(0.061)	(0.061)	(0.061)	
Youngest sibling	-0.093***	-0.036	-0.030	-0.016	-0.084***	-0.029	-0.026	-0.020	
under 5	(0.027)	(0.027)	(0.026)	(0.026)	(0.028)	(0.028)	(0.028)	(0.028)	
Youngest sibling	0.018	0.005	0.009	0.014	0.004	0.004	0.008	0.011	
under 2	(0.029)	(0.029)	(0.029)	(0.028)	(0.030)	(0.030)	(0.030)	(0.030)	
Child's subjective h	ealth (Ref: Exc	ellent)							
Very good	0.008	0.002	0.009	0.014	-0.023	-0.028	-0.026	-0.023	
	(0.020)	(0.019)	(0.019)	(0.019)	(0.021)	(0.020)	(0.020)	(0.020)	
Good	-0.027	-0.026	-0.017	-0.014	0.023	0.024	0.034	0.036	
	(0.032)	(0.031)	(0.031)	(0.031)	(0.034)	(0.033)	(0.033)	(0.033)	
Fair	0.032	0.024	0.025	0.023	-0.017	-0.012	-0.011	-0.012	
	(0.052)	(0.050)	(0.050)	(0.049)	(0.053)	(0.052)	(0.052)	(0.052)	
Develpmental	-0.000	-0.002	0.004	0.008	-0.024	-0.025	-0.021	-0.019	
problems	(0.029)	(0.028)	(0.028)	(0.028)	(0.031)	(0.030)	(0.030)	(0.030)	
HH type =	0.033	-0.009	-0.017	-0.024	0.034	0.001	-0.006	-0.009	
Extended family	(0.022)	(0.022)	(0.022)	(0.022)	(0.024)	(0.024)	(0.024)	(0.024)	
Household income	(Ref: Less thar	1 20K)							
20-30K	-0.002	0.031	0.011	0.019	-0.077	-0.031	-0.049	-0.047	
	(0.092)	(0.078)	(0.080)	(0.077)	(0.093)	(0.094)	(0.093)	(0.093)	

30-50K	-0.050	0.024	0.000	0.013	-0.134*	-0.079	-0.095	-0.089
	(0.079)	(0.066)	(0.068)	(0.066)	(0.079)	(0.081)	(0.081)	(0.080)
50-70K	-0.098	0.021	-0.000	0.019	-0.194**	-0.109	-0.123	-0.112
	(0.077)	(0.066)	(0.068)	(0.065)	(0.078)	(0.080)	(0.080)	(0.079)
70-100K	-0.095	0.035	0.016	0.029	-0.126	-0.035	-0.045	-0.040
	(0.077)	(0.065)	(0.067)	(0.065)	(0.077)	(0.079)	(0.079)	(0.079)
100-150K	-0.106	0.052	0.031	0.045	-0.160**	-0.049	-0.062	-0.057
	(0.077)	(0.066)	(0.068)	(0.066)	(0.077)	(0.080)	(0.080)	(0.080)
More than	0.112	0.072	0.055	0.064	0 177**	0.047	0.050	0.057
150K	-0.112	0.072	0.055	0.064	-0.177**	-0.047	-0.059	-0.057
	(0.078)	(0.068)	(0.070)	(0.067)	(0.078)	(0.082)	(0.082)	(0.081)
Married	-0.043**	-0.015	-0.020	-0.019	-0.024	-0.009	-0.012	-0.011
	(0.021)	(0.020)	(0.020)	(0.020)	(0.022)	(0.022)	(0.022)	(0.022)
Cohabiting	-0.186*	-0.172*	-0.163*	-0.176*	-0.190	-0.210*	-0.200*	-0.205*
	(0.101)	(0.097)	(0.096)	(0.095)	(0.119)	(0.117)	(0.117)	(0.117)
Mother's age (Ref: •	<25 years)	0.003	0.006	0.012	0 0 2 9	0.072*	0.072*	0.060*
25-29 years	(0.000	-0.005	(0.024)	(0.024)	-0.036	-0.075	-0.072	-0.009
20.24	(0.055)	(0.055)	(0.034)	(0.034)	(0.030)	(0.058)	(0.058)	(0.056)
30-34 years	0.021	0.021	0.029	0.032	-0.011	-0.057	-0.052	-0.054
25.20	(0.034)	(0.038)	(0.038)	(0.037)	(0.037)	(0.042)	(0.042)	(0.041)
35-39 years	0.029	0.023	0.033	0.034	-0.001	-0.054	-0.050	-0.051
	(0.036)	(0.041)	(0.041)	(0.040)	(0.039)	(0.045)	(0.045)	(0.045)
40+ years	0.036	0.002	0.015	0.009	-0.038	-0.109*	-0.100*	-0.107*
	(0.052)	(0.055)	(0.054)	(0.053)	(0.055)	(0.059)	(0.059)	(0.059)
Mother born in	-0.013	-0.018	-0.033	-0.025	-0.067***	-0.063***	-0.073***	-0.070***
INZ	(0.019)	(0.021)	(0.021)	(0.021)	(0.020)	(0.022)	(0.022)	(0.022)
Mother's education	(Ref: No sec	qualification)	0.000	0.002	0 110**	0 125**	0 104**	0 101**
NCEA 1-4	0.039	0.060	0.060	0.062	(0.050)	0.135**	0.134**	0.131**
	(0.056)	(0.052)	(0.051)	(0.050)	(0.058)	(0.056)	(0.056)	(0.056)
Diploma/Trade cert/NCEA 5-6	0.012	0.038	0.041	0.045	0.097*	0.119**	0.119**	0.11/**
	(0.055)	(0.051)	(0.050)	(0.050)	(0.057)	(0.055)	(0.055)	(0.055)
Bachelors	-0.019	0.027	0.024	0.031	0.072	0.096*	0.094*	0.094*
ucgree	(0.056)	(0.052)	(0.052)	(0.051)	(0.058)	(0.057)	(0.056)	(0.056)
Higher degree	0.009	0.043	0.048	0.058	0.085	0.087	0.089	0.090
	(0.057)	(0.054)	(0.053)	(0.052)	(0.059)	(0.058)	(0.058)	(0.058)
Partner is		-0.080	-0.062	-0.040		0.114	0.132	0.141
		(0.096)	(0.095)	(0.092)		(0.108)	(0.107)	(0.106)
Father's age (Ref: <	25 years)							
25-29 years		0.005	0.001	0.020		0.071	0.062	0.072
		(0.044)	(0.045)	(0.043)		(0.047)	(0.047)	(0.047)
30-34 years		-0.000	-0.009	0.010		0.076	0.062	0.073
		(0.046)	(0.047)	(0.045)		(0.049)	(0.050)	(0.049)
35-39 years		-0.007	-0.017	-0.001		0.098*	0.084	0.096*

	(0.049)	(0.049)	(0.047)	(0.051)	(0.052)	(0.051)
40+ years	0.070	0.066	0.084	0.141***	0.131**	0.145***
	(0.052)	(0.053)	(0.051)	(0.054)	(0.055)	(0.054)
Father criminal	-0.022	-0.029	-0.013	0.009	0.005	0.015
history	(0.065)	(0.064)	(0.063)	(0.072)	(0.072)	(0.071)
Father born in NZ	-0.056**	-0.062***	-0.049**	-0.016	-0.017	-0.009
	(0.022)	(0.022)	(0.022)	(0.024)	(0.024)	(0.024)
Father's prioritised ethnicit	y (Ref: NZ European)	)				
Māori	0.093***	0.087***	0.074***	0.043	0.040	0.035
	(0.027)	(0.027)	(0.027)	(0.028)	(0.028)	(0.028)
Pasifika	0.112***	0.081**	0.057	0.118***	0.106***	0.093**
	(0.035)	(0.036)	(0.035)	(0.037)	(0.038)	(0.038)
Asian	-0.032	-0.050	-0.052*	0.009	-0.006	-0.007
	(0.032)	(0.031)	(0.031)	(0.036)	(0.036)	(0.036)
MELAA and	0.044	0.037	0.016	0.068	0.058	0.044
Other	(0.059)	(0.058)	(0.056)	(0.064)	(0.063)	(0.063)
Father's education (Ref: No	sec qualification)					
Sec school/NCEA	-0.036	-0.040	-0.046	-0.006	-0.011	-0.014
1-4	(0.043)	(0.043)	(0.042)	(0.045)	(0.045)	(0.045)
Diploma/Trade cert/NCEA 5-6	-0.026	-0.037	-0.043	-0.016	-0.027	-0.030
	(0.041)	(0.041)	(0.040)	(0.043)	(0.043)	(0.043)
Bachelors degree	-0.081*	-0.091**	-0.087*	-0.030	-0.038	-0.037
	(0.045)	(0.045)	(0.045)	(0.048)	(0.048)	(0.047)
Higher degree	-0.035	-0.052	-0.047	0.023	0.012	0.012
	(0.047)	(0.047)	(0.047)	(0.049)	(0.049)	(0.049)
Father's occupation (Ref: P	rofessionals and adm	nin. workers)				
Father's occupation (Ref: Pro Not employed	0.028	0.025	0.022	-0.005	-0.011	-0.013
	(0.027)	(0.026)	(0.026)	(0.028)	(0.028)	(0.028)
Managers	-0.022	-0.025	-0.029	-0.023	-0.031	-0.034
	(0.025)	(0.025)	(0.025)	(0.027)	(0.027)	(0.026)
Technicians/	0.069**	0.065**	0.056**	0.010	0.006	0.002
Trades workers/ machinery	(0.027)	(0.027)	(0.026)	(0.028)	(0.028)	(0.028)
Other	0.056*	0.057*	0.047	0.037	0.037	0.032
	(0.030)	(0.030)	(0.030)	(0.032)	(0.032)	(0.032)
Income difference	-0.010***	-0.009***	-0.008***	-0.016***	-0.016***	-0.015***
in K\$	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Father shiftwork	0.042**	0.040*	0.034*	0.013	0.006	0.003
(DCW1)	(0.020)	(0.020)	(0.020)	(0.022)	(0.022)	(0.022)
Father self-	-0.021	-0.028	-0.029	-0.050**	-0.058**	-0.060***
employed (DCW2)	(0.022)	(0.022)	(0.021)	(0.023)	(0.023)	(0.023)
Father fulltime	-0 171***	-0.131***	-0.128***	-0 023	-0.032	-0 030
(DCW1)	(0.026)	(0.026)	(0.026)	(0.029)	(0.029)	(0 029)
	(0.020)	(0.020)	(0.020)	(0.025)	(0.02)	(0.020)

Father overtime	-0.076***	-0.079***	-0.075***		-0.078***	-0.083***	-0.078***
(DCW1)	(0.020)	(0.020)	(0.019)		(0.021)	(0.021)	(0.021)
Mental health		-0.023	-0.012			-0.070**	-0.065**
		(0.028)	(0.027)			(0.029)	(0.029)
Physical health		-0.017	-0.015			0.004	0.005
		(0.018)	(0.018)			(0.019)	(0.019)
Subjective health		0.009	-0.006			0.012	0.001
		(0.026)	(0.026)			(0.028)	(0.028)
Extraversion		-0.016	-0.029**			0.026*	0.018
		(0.013)	(0.012)			(0.013)	(0.013)
Agreeableness		0.024	0.006			0.005	-0.005
		(0.017)	(0.016)			(0.018)	(0.018)
Conscientiousness		0.041***	0.029**			0.024	0.016
		(0.015)	(0.015)			(0.016)	(0.016)
Neuroticism		0.008	0.007			0.035**	0.036**
		(0.016)	(0.016)			(0.017)	(0.017)
Openness		0.026*	0.014			0.021	0.014
		(0.016)	(0.016)			(0.017)	(0.017)
Perceived stress		0.009	0.014			0.008	0.011
scale		(0.011)	(0.011)			(0.012)	(0.012)
Relationship		0.018*	0.009			0.031***	0.026**
quality		(0.010)	(0.010)			(0.010)	(0.010)
Average		0.039***	0.027***			0.022**	0.014
helpfulness of		(0.009)	(0.009)			(0.009)	(0.010)
Perceived		( )	0.123***			( )	0.055**
influence (DCW1)			(0.026)				(0.026)
Parental identity			-0.012				-0.052
(DCW2)			(0.056)				(0.060)
Parental			. ,				. ,
confidence			0 100***				0.05.2*
(DCW1)			(0.027)				(0.027)
Coodmonst			(0.027)				(0.027)
(DCW2)			(0.038*				(0.055***
Darontal			(0.022)				(0.025)
satisfaction			0.009				0.030
(DCW1)			(0.014)				(0.014)
Work-life balance			-0.016				-0.013
			(0.017)				(0.018)
Observations 3,098	3,098	3,098	3,098	3,098	3,098	3,098	3,098
Pseudo R-squared 0.022	0.076	0.088	0.114	0.022	0.057	0.066	0.074

Source: GUINZ DCW0, DCW1 and DCW2, own calculations and illustrations.

*Notes:* \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

	Marginal effects of logit estimation – Dependent				Linear estimation – Dependent variable: Weeks of leave			
	variable:	Father took m	ore than 2 we	eks of leave				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se
Is twin	0.170***	0.184***	0.174***	0.175***	0.686*	0.804**	0.686*	0.694*
	(0.043)	(0.043)	(0.044)	(0.044)	(0.399)	(0.400)	(0.405)	(0.406)
Is girl	-0.002	-0.007	-0.009	-0.008	0.083	0.053	0.054	0.060
	(0.016)	(0.016)	(0.016)	(0.016)	(0.126)	(0.126)	(0.126)	(0.127)
Pregnancy	0.006	0.008	0.005	0.004	-0.206	-0.182	-0.188	-0.180
planned	(0.021)	(0.021)	(0.021)	(0.021)	(0.160)	(0.161)	(0.161)	(0.162)
Number of siblings	(Ref: None)							
One	-0.067**	-0.066**	-0.066**	-0.067**	-0.610***	-0.580***	-0.543**	-0.553***
	(0.030)	(0.029)	(0.030)	(0.030)	(0.211)	(0.213)	(0.214)	(0.214)
Тwo	-0.115***	-0.104***	-0.107***	-0.109***	-0.897***	-0.810***	-0.784***	-0.793***
	(0.031)	(0.032)	(0.032)	(0.032)	(0.249)	(0.254)	(0.255)	(0.257)
Three	-0.086*	-0.092**	-0.094**	-0.098**	-0.816**	-0.831**	-0.814**	-0.853**
	(0.044)	(0.043)	(0.043)	(0.043)	(0.353)	(0.359)	(0.360)	(0.361)
Four or more	-0.142***	-0.131***	-0.133***	-0.137***	-1.205***	-1.165***	-1.176***	-1.186***
	(0.047)	(0.049)	(0.049)	(0.048)	(0.429)	(0.443)	(0.444)	(0.445)
Youngest sibling	0.022	0.028	0.032	0.033	0.380*	0.483**	0.503**	0.519**
under 5	(0.028)	(0.028)	(0.028)	(0.028)	(0.203)	(0.207)	(0.207)	(0.207)
Youngest sibling	0.028	0.020	0.020	0.019	-0.084	-0.096	-0.113	-0.112
under 2	(0.028)	(0.028)	(0.028)	(0.028)	(0.216)	(0.216)	(0.216)	(0.217)
Child's subjective h	nealth (Ref: Exc	cellent)						
Very good	0.021	0.015	0.010	0.011	0.047	0.015	-0.003	0.009
	(0.019)	(0.019)	(0.019)	(0.019)	(0.148)	(0.148)	(0.148)	(0.148)
Good	0.009	0.005	-0.005	-0.005	-0.077	-0.082	-0.138	-0.130
	(0.032)	(0.031)	(0.031)	(0.031)	(0.241)	(0.241)	(0.242)	(0.242)
Fair	-0.110***	-0.110***	-0.114***	-0.113***	-0.448	-0.504	-0.527	-0.505
	(0.037)	(0.037)	(0.037)	(0.037)	(0.376)	(0.376)	(0.376)	(0.377)
Develpmental	0.025	0.024	0.026	0.027	0.017	0.006	0.010	0.022
problems	(0.027)	(0.027)	(0.027)	(0.027)	(0.216)	(0.216)	(0.216)	(0.216)
HH type =	-0.010	-0.019	-0.015	-0.016	0.066	0.022	0.044	0.046
Extended family	(0.023)	(0.023)	(0.023)	(0.023)	(0.173)	(0.181)	(0.182)	(0.182)
Household Income	(Ref: Less thar	n 20K)						
20-30K	0.044	0.032	0.016	0.020	0.053	0.099	-0.074	0.005
	(0.099)	(0.091)	(0.090)	(0.089)	(0.818)	(0.818)	(0.817)	(0.817)
30-50K	0.018	0.015	0.016	0.019	0.311	0.458	0.388	0.444
	(0.085)	(0.078)	(0.079)	(0.078)	(0.711)	(0.710)	(0.710)	(0.710)
50-70K	0.069	0.075	0.073	0.078	0.885	1.119	1.035	1.095
	(0.083)	(0.077)	(0.079)	(0.077)	(0.700)	(0.702)	(0.702)	(0.702)
70-100K	0.059	0.070	0.065	0.068	0.751	1.032	0.914	0.977
	(0.083)	(0.076)	(0.078)	(0.077)	(0.696)	(0.699)	(0.699)	(0.699)

# Table A.6 - Full estimation results - outcome: leave taking

100-150K	0.094	0.109	0.106	0.110	0.830	1.172*	1.099	1.168*
	(0.083)	(0.077)	(0.078)	(0.077)	(0.698)	(0.703)	(0.703)	(0.704)
More than		0.074	0.000					4 4 9 5
150K	0.046	0.071	0.066	0.067	0.709	1.142	1.081	1.135
	(0.083)	(0.078)	(0.079)	(0.078)	(0.705)	(0.712)	(0.712)	(0.713)
Married	-0.027	-0.033	-0.027	-0.027	-0.302*	-0.272*	-0.222	-0.211
	(0.020)	(0.020)	(0.020)	(0.020)	(0.157)	(0.160)	(0.161)	(0.161)
Cohabiting	-0.009	0.014	0.018	0.017	0.357	0.482	0.525	0.512
	(0.100)	(0.098)	(0.097)	(0.097)	(0.768)	(0.779)	(0.779)	(0.779)
Mother's age (Ref: ·	<25 years)	0.015	0.000	0.010	0.407	0.4.40	0.450	0.445
25-29 years	-0.011	-0.015	-0.020	-0.018	-0.107	-0.142	-0.152	-0.145
	(0.034)	(0.036)	(0.036)	(0.036)	(0.274)	(0.289)	(0.289)	(0.289)
30-34 years	0.019	0.009	0.006	0.007	0.241	0.188	0.176	0.183
	(0.034)	(0.039)	(0.040)	(0.039)	(0.277)	(0.313)	(0.312)	(0.313)
35-39 years	0.058	0.048	0.044	0.046	0.527*	0.448	0.423	0.431
	(0.037)	(0.043)	(0.043)	(0.043)	(0.292)	(0.336)	(0.335)	(0.335)
40+ years	0.114**	0.082	0.081	0.082	0.654	0.452	0.460	0.471
	(0.055)	(0.059)	(0.060)	(0.060)	(0.398)	(0.440)	(0.439)	(0.440)
Mother born in	-0.021	-0.012	-0.001	-0.001	-0.144	-0.104	-0.045	-0.031
NZ	(0.018)	(0.020)	(0.020)	(0.020)	(0.141)	(0.158)	(0.160)	(0.160)
Mother's education	n (Ref: No sec	qualification)						
Sec school/	0.058	0.060	0.058	0.059	0.435	0.470	0.427	0.432
NCEA 1-4	(0.056)	(0.057)	(0.058)	(0.058)	(0.452)	(0.453)	(0.452)	(0.452)
Diploma/Trade	0.046	0.041	0.038	0.039	0.370	0.369	0.303	0.311
cert/NCEA 5-6	(0.055)	(0.056)	(0.057)	(0.057)	(0.445)	(0.446)	(0.446)	(0.446)
Bachelors	0.059	0.054	0.055	0.056	0.329	0.346	0.297	0.299
degree	(0.055)	(0.057)	(0.057)	(0.057)	(0.450)	(0.454)	(0.454)	(0.454)
Higher degree	0.048	0.035	0.029	0.030	0.346	0.316	0.208	0.216
	(0.056)	(0.058)	(0.058)	(0.058)	(0.459)	(0.465)	(0.466)	(0.466)
Partner is		-0.171*	-0.188**	-0.185**		-1.141	-1.216	-1.223
biological father		(0.089)	(0.088)	(0.088)		(0.807)	(0.806)	(0.806)
Father's age (Ref: <	25 vears)	( )	( )	( )			( )	( )
25-29 vears	, ,	0.013	0.013	0.015		0.208	0.197	0.260
,		(0.048)	(0.048)	(0.048)		(0.387)	(0 387)	(0 388)
30-34 years		0.025	0.029	0.031		0 181	0 185	0.236
So Sa years		(0.050)	(0.049)	(0.049)		(0 300)	(0 300)	(0.400)
25 20 years		0.022	0.022	0.022		0.355)	0.555	0.202
55-55 years		(0.051)	(0.022	0.023		(0.412)	(0.414)	(0.414)
401		(U.U.)	(1001)	(0.01)		(0.413)	(0.414)	(0.414)
40+ years		0.047	0.044	0.046		0.414	0.3/6	0.406
		(0.055)	(0.054)	(0.054)		(0.433)	(0.435)	(0.435)
Father criminal		-0.154	-0.163	-0.162		-0.165	-0.164	-0.180
mstory		(0.101)	(0.101)	(0.101)		(0.566)	(0.565)	(0.565)
Father born in NZ		0.018	0.021	0.024		-0.190	-0.158	-0.137

Faither's prioritised ethnicity (Her Vertice0,0060,0060,0070,0210,2230,225Marin0,0070,0070,0170,1010,1030,0080,0070,0080,0070,0080,01
Mori0.0060.0060.0060.4140.2370.226Pasifika0.0270.02700.02700.02700.0287Pasifika0.0430.0170.0370.0280.0270.027Asian0.0470.0640.0370.02180.0270.027MELA and other0.0370.0360.0310.13140.1310.1315Father's education (Ref. No sector other0.0320.0270.0270.1430.1430.1431-40.0400.0400.0410.4340.1440.1440.1441-40.0400.0240.0270.1410.1440.1441-40.0400.0410.0410.1440.1440.1441-40.0400.0410.0410.1440.1440.1441-40.0400.0240.0270.1410.1420.1411-40.0400.0240.0210.1410.1420.1411-40.0400.0240.0210.1410.1410.1231-40.0400.0240.0210.1410.1410.1411-40.0400.0240.0210.1410.1410.1411-50.0380.0210.0210.1410.1410.1411-40.0390.0240.0210.1410.1410.1411-50.0390.0410.0410.1410.1410.1411-60.0230.0240.0210.141
Image[0.027][0.026][0.026][0.027][0.208][0.208][0.208]Pafifia0.040.0140.0170.1450.0800.087Asian[0.030](0.030)(0.031)(0.250)(0.250)(0.258)MELAA and other0.0600.0630.0310.1310.3150.315Father's education (Ref: Nota other0.0270.0500.3140.3190.318Father's education (Ref: Nota other0.0280.0210.0430.0430.4310.341Set school/NCEA0.0200.0200.0410.3440.3440.3440.344Diploma/Trade cert/NCEA 5-60.0320.0390.0390.3280.3280.3280.328Pather's education (Ref: Protection cert/NCEA 5-60.0580.0550.0560.5430.3280.328Bachelors degree0.0580.0590.0490.0490.0490.3140.3280.328Ingle red action (Ref: Protection cert/NCEA 5-60.0580.0590.0580.5680.5880.5780.3780.378Ingle red action (Ref: Protection cert/NCEA 5-60.0580.0590.0510.0510.0510.0510.0510.0510.0510.051Ingle red action (Ref: Protection cert/NCEA0.0590.0510.0210.0110.0210.0120.0120.0120.0120.0120.0120.0120.0120.0120.0120.0120.0120.0120
Pasifika0.0040.0190.017-0.145-0.0800.087(0.035)(0.037)(0.037)(0.279)(0.279)(0.278)Asian0.0470.060-0.0180.0510.0510.0510.0510.019MELA and0.0070.0600.0100.1310.1310.131other0.0700.0630.0310.1430.1310.131Fatter's education (Ref: No sec-truct0.0200.0210.0210.0310.0310.1311-40.0400.0400.0410.0410.1410.1410.141Diplom/Trade0.0580.0560.5450.5270.428crt/NCFA 5-60.0580.0560.03200.3200.321Bachelors degree0.0580.0490.0490.0490.3410.321Anagers0.099**0.0450.0450.131*0.2580.58Anagers0.099**0.0410.0410.1410.2310.274Anagers0.099**0.0410.011*0.1510.2740.214Technicians/0.099**0.0210.011*0.1510.2140.214Crture0.0210.0210.0210.011*0.1510.214Technicians/0.0210.0210.0210.1140.1240.214Technicians/0.020.0210.0210.1140.1240.124Crture0.0210.0210.0210.1140.1240.124<
Image: big
<table-cell><math> </math></table-cell>
indexindexindexindexindexindexindexindexMELA and otherindexindexindexindexindexindexindexHELA and otherindexindexindexindexindexindexindexFather selucation (Ref. was selected)indexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexindexindex <t< td=""></t<>
MELA and other0.0070.0050.0140.0170.015Ather0.063)0.063)0.043)0.043)0.043)0.043)Father's education (Ref: No sec:Extonol/NCEA0.0220.0270.2420.1910.1751-40.0400.0400.0410.3430.3440.3400.341Diploma/Trade0.076*0.7670.545*0.5270.496cert/NCEA 5-60.0580.5580.5580.5280.3620.328Bachelors degree0.0580.0500.0420.3400.3280.362Higher degree0.99**0.0540.074*0.361*0.375*0.375*Higher degree0.099**0.0210.0210.3740.315*0.375*Father's occupation (Ref: Professional and second se
other(0.604)(0.603)(0.603)(0.493)(0.493)(0.493)Father's education (Ref: No security)(302)0.0270.0270.2420.1910.75Sec schol/NCEA(0.020)(0.020)(0.010)(0.344)(0.344)(0.344)(0.344)1-4(0.030)(0.030)(0.320)(
NoteNoteNoteNoteNoteSe school/NCEA0.3020.0260.04710.0401
Sec school/NCEA0.0320.0260.0270.2420.1910.1751-40.0400.0400.0410.3440.3440.3440.344Diploma/Trade0.076**0.076**0.076**0.545*0.5270.496cert/NCEA 5-60.0580.0390.0390.3280.3280.3280.328Bachelors degree0.0580.5550.5650.5430.5280.5280.528Higher degree0.0490.049*0.049*0.0490.646*0.5880.585Higher degree0.049*0.049*0.049*0.616*0.5880.528Fatter's occupation (Ref: Professor0.047*0.047**0.617**0.314*0.228*0.228*Technicans/0.020*0.0210.021*0.021*0.11**0.228*0.228*0.228*Technicans/0.0240.0240.021*0.021*0.11**0.128*0.218*0.228*Technicans/0.0240.024*0.021*0.121*0.121*0.121*0.121*0.121*Technicans/0.05*0.02*0.021*0.021*0.121*0.121*0.121*0.121*0.121*Technicans/0.05*0.02*0.02*0.02*0.02*0.01*0.121*0.121*0.121*0.121*Technicans/0.05*0.02*0.02*0.02*0.02*0.14*0.14*0.14*0.14*Technicans/0.02*0.02*0.02*0.02*0.01*<
1-4(0.04)(0.04)(0.04)(0.04)(0.34)(0.34)(0.34)Diploma/Trade cert/NCFA 5-6(0.03)(0.03)(0.03)(0.32)(0.32)(0.32)Bachelors degree0.0580.0550.5640.5430.4970.489Bachelors degree(0.042)(0.042)(0.042)0.616*0.5830.552Higher degree0.09**0.09**0.09**0.616*0.5830.555Fatter's occupation (Ref: Professional and attrime vorters)0.0450.0450.037*0.037*0.037*Fatter's occupation (Ref: Professional and attrime vorters)0.0210.0210.11*0.2230.228*Technicians/0.020*0.0210.0210.11*0.125*0.125*Technicians/0.026*0.0210.0210.11*0.192*0.121*operators0.026*0.021*0.021*0.191*0.192*0.192*other s/frequence0.026*0.021*0.021*0.11*0.191*0.191*0.191*in KS0.026*0.021*0.021*0.021*0.021*0.01*0.021*0.01*0.01*0.01*in KS0.029*0.021*0.021*0.021*0.01*0.021*0.01* </td
$ end{pmp} $ Diploma/Trade 0.076** 0.077** 0.076* 0.054* 0.54* 0.52* 0.494 end(NCA 5-6)
cert/NCEA 5-6(0.038)(0.039)(0.039)(0.328)(0.328)(0.328)Bachelors degree0.0580.0560.0540.4970.489Higher degree0.099**0.099**0.061*0.5430.558Higher degree0.099**0.0450.0450.0370.3700.370Fatter's occupation (Ref: Professional and the second sec
<table-row>Bachelors degree0.580.0560.0540.4300.4970.497Higher degree0.099**0.099**0.099**0.616*0.58*0.58*Higher degree0.040*0.040*0.040*0.374*0.374*0.374*0.374*0.374*0.374*0.314*0.23*0.23*Stater's occupation (Ref: Professor and the second second</table-row>
Image and the set of
<table-cell>          Higher degree         0.099**         0.098**         0.099**         0.014         0.161*         0.581         0.581           Autor         0.040         0.040         0.0370         0.370         0.370         0.370           Father's occupation (Ref: Protexture         0.000***         0.071***         0.011***         0.111**         0.123         0.128           Managers         0.000***         0.071***         0.011***         0.123         0.128           Technicans/         0.000***         0.021***         0.011***         0.123         0.128           Technicans/         0.000         0.020         0.020***         0.128         0.128           Technicans/         0.000         0.020****         0.028         0.128         0.128           Technicans/         0.000         0.020****         0.128         0.128         0.128           Teacher workers/         0.028         0.012****         0.128         0.128         0.128           Operators         0.029         0.021****         0.128         0.128         0.128           Other         0.029         0.021****         0.021****         0.128         0.128         0.128           Income difference         &lt;</table-cell>
Interface
indexindexindexindexindexindexindexAnagersindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindexindexIndexindexindexindexindexindexindex
Managers-0.080***-0.071***-0.072***-0.311**-0.223-0.228(0.021)(0.021)(0.021)(0.017)(0.176)(0.176)(0.176)(0.176)Technicians/-0.0090.0020.0000.2630.333*0.322*trades workers/
Indext (0.021)(0.021)(0.021)(0.175)(0.176)(0.176)Technicians/ trades workers/ operators operators-0.0900.0200.0000.2630.33*0.32*Machinery operators Other(0.026)(0.026)(0.020)(0.191)(0.192)(0.192)Other0.0050.0130.0120.2220.3050.300Income difference (1 K\$)0.0110.021(0.029)(0.029)0.0110.0100.010Income difference (1 CCV1)0.091***0.0020.0020.0100.0100.0130.013Father self- (DCW1)0.031***0.0210.023****0.1620.1620.1620.162Father self- (DCW1)0.091***0.023***0.023****0.1620.1620.1620.162Father self- (DCW1)0.097***0.0230.023****0.156***0.1510.156***0.156***Father self- (DCW1)0.0070.023***0.023****0.1510.151***0.156***ODT0.027***0.023****0.024****0.151***0.136****Father self- (DCW1)0.070.03****0.024****0.151***0.151***0.156****ODT0.021****0.021*****0.023*****0.151***********************************
Technicians/ trades workers/ machinery operators-0.0090.0020.0000.2630.33*0.322*machinery operators Other(0.020)(0.020)(0.020)(0.191)(0.192)(0.192)0.0050.0130.0120.2220.3050.3030.30310.020(0.020)(0.020)(0.020)(0.216)(0.216)(0.216)10.020.0010.0020.0020.0020.0100.0100.01010.45(0.020)(0.020)(0.020)(0.020)(0.013)(0.163)(0.163)Father self- oppoled (DCW1)-0.031***(0.021)(0.021)(0.021)(0.021)(0.162)(0.162)Father fultime (DCW1)-0.037***(0.021)(0.021)(0.224)(0.224)(0.225)Father overtime (DCW1)(0.07)(0.13)(0.14)(0.153)(0.153)(0.154)
trades workers/ machinery operators         (0.026)         (0.026)         (0.026)         (0.191)         (0.192)         (0.192)           Other         0.005         0.013         0.012         0.222         0.305         0.300           (0.029)         (0.029)         (0.029)         (0.021)         (0.216)         (0.216)         (0.216)           Income difference in K\$         0.001         0.002         (0.021)         (0.015)         (0.015)         (0.015)           Father shiftwork (DCW1)         0.092***         0.091***         0.090***         0.0163         (0.163)         (0.163)           Father self- employed (DCW2)         0.021         (0.022)         (0.022)         (0.022)         (0.021)         (0.162)         (0.162)           Father fulltime (DCW1)         -0.091***         -0.077***         -0.079***         -0.498***         -0.498***         -0.491***           Pather self- employed (DCW2)         (0.021)         (0.022)         (0.021)         (0.021)         (0.021)         (0.162)         (0.162)         (0.162)           Father fulltime (DCW1)         (0.027)         (0.021)         (0.021)         (0.021)         (0.17)         (0.133)         (0.143)         (0.153)           GDCW1         (0.0
Interniery operators         (0.026)         (0.026)         (0.026)         (0.191)         (0.192)         (0.192)           Other         0.005         0.013         0.012         0.222         0.305         0.300           (0.029)         (0.029)         (0.029)         (0.021)         (0.216)         (0.216)         0.216)           Income difference         0.001         0.002         (0.020)         (0.020)         0.010         -0.010         -0.010         0.015)         0.015)         0.015)           Income difference         0.002         (0.002)         (0.020)         (0.020)         0.010         0.015)         0.015)         0.015)           Father shiftwork         0.092***         0.991***         0.909***         0.211         0.203         0.191           (DCW1)         0.020         0.020         0.020         0.015         0.163         0.163         0.163           Father self-         -0.081***         -0.07***         -0.083***         -0.498***         -0.498***         -0.491****           employed (DCW2)         (0.027)         (0.027)         0.027         0.162         0.162         0.162           fCDW1         0.007         0.023         0.027         0.
Other0.0050.0130.0120.2220.3050.30010.0290.0290.0290.2160.2160.2160.2161ncome difference0.0010.0010.0020.0100.0100.0081n K\$0.0020.0020.0020.0100.0150.0150.015Father shiftwork (DCW1)0.092***0.091***0.002**0.1630.1630.1630.163Father self- employed (DCW2)0.081***0.022**0.022**0.022**0.023***0.023***0.162***0.162***0.162***Father fulltime (DCW1)0.027**0.027**0.027***0.027***0.224***0.224***0.225***Father overtime (DCW1)0.007***0.014****0.015****0.153***0.153***0.154***(DCW1)0.020*****0.020*****0.020******0.015***********************************
Income difference in K\$(0.029)(0.029)(0.029)(0.021)(0.216)(0.216)Income difference in K\$0.0010.0010.0020.0020.0100.0100.008Tather shiftwork (DCW1)0.092***0.091***0.090***0.2110.2030.163)0.163)Father self- employed (DCW2)0.081***0.00270.02010.02010.0210.16200.16200.1621Father fulltime (DCW1)0.090***0.0230.0230.0230.0240.16210.16210.1621Father overtime (DCW1)0.0070.0130.0140.0240.22400.22510.136Father overtime (DCW1)0.0200.0200.0200.0200.1530.1430.154
Income difference in K\$0.0010.0010.0020.0020.0100.0100.003Father shiftwork (DCW1)0.092***0.091***0.090***0.2110.2030.191(DCW1)0.0200.0200.0200.1630.1630.1630.163Father self- employed (DCW2)-0.081***-0.077***-0.079***-0.498***-0.488***-0.491***(DCW1)0.0200.0200.0220.0220.1620.1620.1620.162Father fulltime (DCW1)-0.090***-0.083***-0.083***-0.564***-0.519**-0.506***Father overtime (DCW1)0.0070.0140.01530.143-0.136Father overtime (DCW1)0.0200.0200.0200.0200.1530.153
in K\$(0.002)(0.002)(0.002)(0.015)(0.015)(0.015)Father shiftwork (DCW1)0.092****0.091****0.090****0.2110.2030.191(DCW1)(0.020)(0.020)(0.020)(0.163)(0.163)(0.163)0.163)Father self- employed (DCW2)-0.081****-0.077****-0.079****-0.498****-0.488****-0.498****-0.498****Father fulltime (DCW1)-0.090****-0.083****-0.083****-0.564***-0.519***-0.506***Father overtime (DCW1)0.0070.0130.014-0.175-0.143-0.136(DCW1)(0.020)(0.020)(0.020)(0.153)(0.153)(0.154)
Father shiftwork (DCW1)         0.092***         0.091***         0.090***         0.211         0.203         0.191           Indext (DCW1)         (0.020)         (0.020)         (0.020)         (0.020)         (0.163)         (0.163)         (0.163)           Father self- employed (DCW2)         -0.081***         -0.077***         -0.079***         -0.498***         -0.488***         -0.491***           COULD         (0.022)         (0.022)         (0.022)         (0.022)         (0.022)         (0.162)         (0.162)         (0.162)           Father fulltime (DCW1)         -0.090***         -0.083***         -0.083***         -0.564**         -0.519**         -0.506**           Father overtime (DCW1)         (0.027)         (0.027)         (0.027)         (0.027)         (0.143)         -0.136           (DCW1)         (0.020)         (0.020)         (0.020)         (0.153)         (0.153)         (0.154)
(DCW1)         (0.020)         (0.020)         (0.020)         (0.020)         (0.020)         (0.163)         (0.163)         (0.163)           Father self-         -0.081***         -0.077***         -0.079***         -0.498***         -0.488***         -0.491***           employed (DCW2)         (0.022)         (0.022)         (0.022)         (0.162)         (0.162)         (0.162)           Father fulltime         -0.090***         -0.083***         -0.083***         -0.564**         -0.519**         -0.506**           (DCW1)         (0.027)         (0.027)         (0.027)         (0.224)         (0.224)         (0.225)           Father overtime         0.007         0.013         0.014         -0.175         -0.143         -0.136           (DCW1)         (0.020)         (0.020)         (0.020)         (0.153)         (0.153)         (0.154)
Father self-       -0.081***       -0.077***       -0.079***       -0.498***       -0.488***       -0.491***         employed (DCW2)       (0.022)       (0.022)       (0.022)       (0.162)       (0.162)       (0.162)         Father fulltime       -0.090***       -0.083***       -0.083***       -0.564**       -0.519**       -0.506**         (DCW1)       (0.027)       (0.027)       (0.027)       (0.027)       (0.224)       (0.224)       (0.225)         Father overtime       0.007       0.013       0.014       -0.175       -0.143       -0.136         (DCW1)       (0.020)       (0.020)       (0.020)       (0.153)       (0.153)       (0.154)
employed (DCW2)         (0.022)         (0.022)         (0.022)         (0.162)         (0.162)         (0.162)           Father fulltime         -0.090***         -0.083***         -0.0564**         -0.519**         -0.506**           (DCW1)         (0.027)         (0.027)         (0.027)         (0.027)         (0.224)         (0.224)         (0.225)           Father overtime         0.007         0.013         0.014         -0.175         -0.143         -0.136           (DCW1)         (0.020)         (0.020)         (0.020)         (0.153)         (0.153)         (0.154)
Father fulltime       -0.090***       -0.083***       -0.083***       -0.564**       -0.519**       -0.506**         (DCW1)       (0.027)       (0.027)       (0.027)       (0.224)       (0.224)       (0.225)         Father overtime       0.007       0.013       0.014       -0.175       -0.143       -0.136         (DCW1)       (0.020)       (0.020)       (0.020)       (0.153)       (0.153)       (0.154)
(DCW1)         (0.027)         (0.027)         (0.027)         (0.224)         (0.224)         (0.225)           Father overtime         0.007         0.013         0.014         -0.175         -0.143         -0.136           (DCW1)         (0.020)         (0.020)         (0.020)         (0.153)         (0.153)         (0.154)
Father overtime0.0070.0130.014-0.175-0.143-0.136(DCW1)(0.020)(0.020)(0.020)(0.153)(0.153)(0.154)
(DCW1) (0.020) (0.020) (0.153) (0.153) (0.154)
Mental health 0.024 0.024 0.012 0.027
(0.027) (0.027) (0.218) (0.218)
Physical health         0.011         0.011         0.120         0.123
(0.017) (0.018) (0.137) (0.138)
Subjective health         -0.061**         -0.063**         -0.324         -0.318

Extraversion			-0.011	-0.012			-0.141	-0.148
			(0.012)	(0.012)			(0.095)	(0.096)
Agreeableness			-0.014	-0.015			-0.189	-0.206
			(0.016)	(0.016)			(0.128)	(0.129)
Conscientiousness			-0.040***	-0.042***			-0.237**	-0.248**
			(0.014)	(0.014)			(0.114)	(0.114)
Neuroticism			-0.020	-0.020			-0.087	-0.095
			(0.016)	(0.016)			(0.123)	(0.123)
Openness			0.014	0.012			0.228*	0.207*
			(0.016)	(0.016)			(0.122)	(0.123)
Perceived stress			-0.011	-0.010			0.045	0.052
scale			(0.011)	(0.011)			(0.086)	(0.086)
Relationship			0.019**	0.019*			0.160**	0.164**
quality			(0.010)	(0.010)			(0.076)	(0.077)
Average			-0.020**	-0.022**			-0.133*	-0.147**
helpfulness of			(0,009)	(0.009)			(0.069)	(0.070)
family Perceived			(0.000)	0.008			(0.000)	0.245
influence (DCW1)				(0.024)				(0.182)
Parental identity				-0.008				0.184
(DCW2)				(0.053)				(0.427)
Parental				, , ,				( )
confidence								
(DCW1)				0.033				0.312
				(0.025)				(0.194)
Good parent (DCW2)				0.000				-0.352**
				(0.021)				(0.167)
Parental satisfaction				0.006				0.035
(DCW1)				(0.013)				(0.100)
Work-life balance				-0.005				-0.064
				(0.016)				(0.129)
Constant					1.137	1.854	3.594**	3.159*
					(1.065)	(1.347)	(1.627)	(1.710)
Observations	2,539	2,539	2,539	2,539	2,539	2,539	2,539	2,539
Pseudo R-squared	0.026	0.059	0.070	0.071	0.019	0.039	0.048	0.052

Source: GUINZ DCW0, DCW1 and DCW2, own calculations and illustrations.

*Notes:* \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

	Marginal	effects of logi	t estimation –	Dependent	Linear estimation – Dependent variable: Points				
	variable: Fa	ther provides	better than av	erage quality		quality (std)			
		с	are						
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
	b/se	b/se	b/se	b/se	b/se	b/se	b/se	b/se	
Is twin	-0.013	-0.018	-0.031	-0.017	0.042	0.039	0.028	0.044	
	(0.054)	(0.054)	(0.053)	(0.052)	(0.056)	(0.056)	(0.055)	(0.054)	
ls girl	-0.018	-0.019	-0.017	-0.018	-0.016	-0.016	-0.013	-0.015	
	(0.017)	(0.017)	(0.017)	(0.017)	(0.018)	(0.018)	(0.017)	(0.017)	
Pregnancy	0.015	0.023	0.028	0.019	0.020	0.027	0.031	0.018	
planned	(0.022)	(0.022)	(0.022)	(0.021)	(0.023)	(0.023)	(0.022)	(0.022)	
Number of siblings	(Ref: None)								
One	-0.134***	-0.124***	-0.112***	-0.114***	-0.149***	-0.141***	-0.126***	-0.127***	
	(0.029)	(0.030)	(0.029)	(0.029)	(0.030)	(0.030)	(0.029)	(0.029)	
Two	-0.228***	-0.218***	-0.193***	-0.195***	-0.237***	-0.228***	-0.199***	-0.200***	
	(0.035)	(0.035)	(0.035)	(0.035)	(0.036)	(0.036)	(0.035)	(0.035)	
Three	-0.196***	-0.195***	-0.190***	-0.209***	-0.207***	-0.213***	-0.205***	-0.229***	
	(0.048)	(0.049)	(0.048)	(0.047)	(0.049)	(0.049)	(0.048)	(0.047)	
Four or more	-0.184***	-0.184***	-0.153**	-0.171***	-0.247***	-0.252***	-0.212***	-0.232***	
	(0.060)	(0.061)	(0.061)	(0.060)	(0.060)	(0.062)	(0.060)	(0.059)	
Youngest sibling	-0.088***	-0.070**	-0.060**	-0.046*	-0.133***	-0.110***	-0.097***	-0.080***	
under 5	(0.027)	(0.027)	(0.027)	(0.027)	(0.029)	(0.029)	(0.029)	(0.028)	
Youngest sibling	0.001	-0.005	0.003	0.013	0.045	0.035	0.045	0.055*	
under 2	(0.029)	(0.029)	(0.028)	(0.028)	(0.031)	(0.030)	(0.030)	(0.029)	
Child's subjective h	nealth (Ref: Exc	ellent)							
Very good	-0.016	-0.017	-0.013	-0.008	-0.010	-0.011	-0.005	0.001	
	(0.020)	(0.020)	(0.020)	(0.020)	(0.021)	(0.021)	(0.020)	(0.020)	
Good	-0.106***	-0.102***	-0.090***	-0.087***	-0.154***	-0.149***	-0.134***	-0.128***	
	(0.033)	(0.032)	(0.032)	(0.032)	(0.034)	(0.034)	(0.033)	(0.033)	
Fair	-0.016	-0.009	0.003	0.009	-0.080	-0.071	-0.059	-0.053	
	(0.051)	(0.051)	(0.050)	(0.050)	(0.054)	(0.053)	(0.052)	(0.051)	
Develpmental	-0.011	-0.015	-0.007	-0.002	-0.007	-0.013	-0.002	0.004	
problems	(0.030)	(0.030)	(0.029)	(0.029)	(0.031)	(0.031)	(0.030)	(0.030)	
HH type =	-0.018	-0.032	-0.040*	-0.044*	-0.020	-0.033	-0.045*	-0.052**	
Extended family	(0.023)	(0.024)	(0.024)	(0.023)	(0.024)	(0.025)	(0.024)	(0.024)	
Household income	(Ref: Less thar	n 20K)							
20-30K	-0.008	0.007	-0.017	-0.007	0.060	0.085	0.050	0.060	
	(0.090)	(0.089)	(0.088)	(0.086)	(0.094)	(0.093)	(0.091)	(0.089)	
30-50K	-0.010	0.029	0.017	0.029	0.001	0.062	0.042	0.054	
	(0.077)	(0.076)	(0.075)	(0.074)	(0.080)	(0.080)	(0.078)	(0.077)	
50-70K	-0.019	0.040	0.028	0.044	-0.001	0.081	0.060	0.080	
	(0.075)	(0.075)	(0.074)	(0.073)	(0.079)	(0.079)	(0.077)	(0.076)	
70-100K	-0.008	0.048	0.043	0.052	-0.011	0.064	0.052	0.064	

# Table A.7 - Full estimation results - outcome: quality of care
	(0.074)	(0.074)	(0.073)	(0.072)	(0.078)	(0.078)	(0.076)	(0.075)
100-150K	0.009	0.069	0.065	0.076	0.015	0.090	0.078	0.092
	(0.075)	(0.075)	(0.074)	(0.073)	(0.078)	(0.079)	(0.077)	(0.076)
More than	0.015	0.074	0.000	0.070	0.020	0.104	0.000	0.000
150K	0.015	0.074	0.069	0.076	0.039	0.104	0.088	0.096
	(0.076)	(0.077)	(0.076)	(0.075)	(0.080)	(0.080)	(0.079)	(0.077)
Married	(0.021)	0.026	0.029	0.031	0.033	0.044*	0.046**	0.049**
	(0.021)	(0.022)	(0.022)	(0.021)	(0.022)	(0.023)	(0.022)	(0.022)
Cohabiting	-0.056	-0.045	-0.029	-0.042	-0.014	0.010	0.025	0.008
	(0.104)	(0.104)	(0.102)	(0.100)	(0.108)	(0.108)	(0.105)	(0.103)
Mother's age (Ref: •	<25 years)	0.024	0.040	0.046	0.062*	0.047	0.059	0.064*
23-29 years	0.045	(0.034)	(0.027)	(0.027)	(0.027)	(0.047	(0.020)	(0.028)
20.24	(0.035)	(0.038)	(0.037)	(0.037)	(0.037)	(0.040)	(0.039)	(0.038)
30-34 years	0.052	0.051	0.059	0.061	0.062**	0.039	0.053	0.053
	(0.036)	(0.041)	(0.040)	(0.040)	(0.037)	(0.043)	(0.042)	(0.041)
35-39 years	0.002	0.004	0.008	0.008	0.042	0.019	0.026	0.025
	(0.038)	(0.044)	(0.044)	(0.043)	(0.039)	(0.046)	(0.045)	(0.044)
40+ years	0.026	0.015	0.031	0.022	0.009	-0.036	-0.010	-0.023
	(0.053)	(0.059)	(0.058)	(0.057)	(0.055)	(0.061)	(0.060)	(0.059)
Mother born in	0.103***	0.078***	0.067***	0.071***	0.110***	0.083***	0.066***	0.071***
INZ	(0.019)	(0.022)	(0.022)	(0.021)	(0.020)	(0.023)	(0.022)	(0.022)
Mother's education	(Ref: No sec	qualification)	0.000	0.005*	0 1 4 0 * *	0 105 **	0 100**	0 1 2 4 * *
NCEA 1-4	0.109	(0.058)	0.093	(0.055)	(0.000)	(0.050)	(0.058)	(0.057)
Dialana /Taada	(0.058)	(0.007	(0.057)	(0.056)	(0.060)	(0.059)	(0.058)	(0.057)
cert/NCEA 5-6	0.091	(0.057)	(0.051)	0.086	(0.050)	(0.050)	(0.057)	0.134**
	(0.057)	(0.057)	(0.056)	(0.056)	(0.059)	(0.059)	(0.057)	(0.056)
Bachelors	0.092	0.074	0.061	0.068	0.168***	0.138**	0.122**	0.128**
	(0.058)	(0.058)	(0.058)	(0.057)	(0.060)	(0.060)	(0.059)	(0.058)
Higher degree	0.179***	0.146**	0.139**	0.150**	0.241***	0.187***	0.179***	0.190***
	(0.059)	(0.060)	(0.059)	(0.059)	(0.061)	(0.062)	(0.060)	(0.059)
Partner is		-0.030	0.014	0.033		-0.187*	-0.126	-0.105
biological lattici		(0.108)	(0.108)	(0.105)		(0.109)	(0.107)	(0.105)
Father's age (Ref: <	25 years)							
25-29 years		0.042	0.021	0.042		0.055	0.024	0.048
		(0.048)	(0.048)	(0.047)		(0.050)	(0.049)	(0.048)
30-34 years		0.022	-0.006	0.015		0.058	0.019	0.043
		(0.050)	(0.050)	(0.049)		(0.053)	(0.052)	(0.051)
35-39 years		-0.000	-0.030	-0.011		0.028	-0.013	0.012
		(0.053)	(0.052)	(0.051)		(0.055)	(0.054)	(0.053)
40+ years		0.020	-0.008	0.014		0.062	0.025	0.054
		(0.056)	(0.055)	(0.054)		(0.058)	(0.057)	(0.056)
Father criminal		-0.079	-0.096	-0.084		-0.127*	-0.148**	-0.132*
history		(0.071)	(0.070)	(0.069)		(0.074)	(0.072)	(0.071)

Father born in NZ	0.003	0.004	0.016	-0.002	-0.004	0.013
	(0.023)	(0.023)	(0.023)	(0.024)	(0.024)	(0.024)
Father's prioritised ethnicity (	Ref: NZ European)	)				
Māori	0.046*	0.041	0.026	0.036	0.026	0.010
	(0.027)	(0.027)	(0.027)	(0.028)	(0.028)	(0.028)
Pasifika	0.074**	0.071**	0.045	0.079**	0.068*	0.038
	(0.035)	(0.035)	(0.035)	(0.037)	(0.037)	(0.037)
Asian	-0.048	-0.047	-0.057*	-0.086**	-0.088**	-0.098***
	(0.035)	(0.035)	(0.034)	(0.037)	(0.036)	(0.036)
MELAA and	-0.102*	-0.111*	-0.133**	-0.097	-0.114*	-0.143**
other	(0.060)	(0.059)	(0.057)	(0.064)	(0.063)	(0.061)
Father's education (Ref: No se	ec qualification)					
Sec school/NCEA	0.123***	0.106**	0.104**	0.132***	0.105**	0.100**
1-4	(0.044)	(0.044)	(0.043)	(0.046)	(0.045)	(0.044)
Diploma/Trade	0.106**	0.080*	0.078*	0.085*	0.047	0.044
cert/NCEA 5-6	(0.042)	(0.041)	(0.041)	(0.044)	(0.043)	(0.042)
Bachelors degree	0.123***	0.096**	0.100**	0.127***	0.086*	0.090*
	(0.046)	(0.046)	(0.045)	(0.048)	(0.048)	(0.047)
Higher degree	0.156***	0.127***	0.130***	0.167***	0.124**	0.125**
	(0.048)	(0.048)	(0.047)	(0.050)	(0.049)	(0.049)
Father's occupation (Ref: Prof	fessionals and adm	nin. workers)				
	0.005	-0.002	-0.007	0.009	0.002	-0.003
	(0.028)	(0.028)	(0.027)	(0.029)	(0.028)	(0.028)
Managers	-0.035	-0.037	-0.040	-0.027	-0.031	-0.037
	(0.026)	(0.026)	(0.025)	(0.027)	(0.026)	(0.026)
Technicians/	-0.034	-0.029	-0.037	-0.084***	-0.076***	-0.084***
trades workers/						
operators	(0.028)	(0.027)	(0.027)	(0.029)	(0.028)	(0.028)
Other	0.020	0.026	0.016	-0.001	0.010	-0.002
	(0.031)	(0.031)	(0.031)	(0.033)	(0.032)	(0.031)
Income difference	-0.007***	-0.006***	-0.006***	-0.009***	-0.008***	-0.007***
in K\$	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Father shiftwork	-0.011	-0.018	-0.021	0.013	0.002	-0.002
(DCW1)	(0.022)	(0.022)	(0.021)	(0.023)	(0.022)	(0.022)
Father self-	0.030	0.017	0.014	0.018	-0.001	-0.005
employed (DCW2)	(0.022)	(0.022)	(0.022)	(0.023)	(0.023)	(0.022)
Father fulltime	-0.121***	-0.131***	-0.131***	-0.144***	-0.158***	-0.159***
(DCW1)	(0.029)	(0.029)	(0.028)	(0.030)	(0.029)	(0.029)
Father overtime	0.004	0.005	0.009	-0.007	-0.006	0.001
(DCW1)	(0.021)	(0.021)	(0.021)	(0.022)	(0.022)	(0.021)
Mental health	. ,	0.027	0.035	· · · /	-0.003	0.007
		(0.029)	(0.028)		(0.029)	(0.029)
Physical health		0.011	0.013		0.008	0.011
·						

			(0.018)	(0.018)			(0.019)	(0.019)
Subjective health			0.045*	0.034			0.047*	0.030
			(0.027)	(0.027)			(0.028)	(0.027)
Extraversion			0.036***	0.021			0.039***	0.021
			(0.013)	(0.013)			(0.013)	(0.013)
Agreeableness			0.028	0.010			0.029*	0.007
			(0.017)	(0.017)			(0.018)	(0.017)
Conscientiousness			0.007	-0.001			0.010	-0.002
			(0.015)	(0.015)			(0.016)	(0.015)
Neuroticism			0.020	0.017			0.017	0.015
			(0.016)	(0.016)			(0.017)	(0.017)
Openness			0.072***	0.062***			0.107***	0.095***
			(0.016)	(0.016)			(0.017)	(0.017)
Perceived stress			0.018	0.022*			0.012	0.017
scale			(0.011)	(0.011)			(0.012)	(0.012)
Relationship			0.045***	0.035***			0.055***	0.044***
quality			(0.010)	(0.010)			(0.010)	(0.010)
Average			0.025***	0.013			0.039***	0.024**
helpfulness of			(0.009)	(0.009)			(0.010)	(0.009)
family Perceived			(0.000)	0.085***			(0.010)	0.100***
influence (DCW1)				(0.024)				(0.025)
Parental identity				0.104*				0.095
(DCW2)				(0.060)				(0.059)
Parental				( )				( )
confidence				0.044				0.0514
(DCW1)				0.044*				0.051*
				(0.026)				(0.026)
Good parent (DCW2)				0.044**				0.088***
				(0.022)				(0.023)
Parental satisfaction				0.073***				0.089***
(DCW1)				(0.013)				(0.014)
Work-life balance				-0.014				-0.024
				(0.017)				(0.017)
Constant					0.559***	0.700***	-0.091	-0.618***
					(0.134)	(0.174)	(0.211)	(0.220)
Observations	3,098	3,098	3,098	3,098	3,098	3,098	3,098	3,098
Pseudo R-squared	0.065	0.083	0.105	0.124	0.111	0.146	0.188	0.221

*Notes:* \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

	NZ European		Māori		<u>Pasifika</u>	
	Involvemen	t at 9m	Involvemer	nt at 9m	Involveme	ent at 9m
	Not much or	Most or	Not much or	Most or	Not much	Most or
	Some of the	All of the	Some of the	All of the	or Some of	All of the
	time	time	time	time	the time	time
Observations	1,390	622	237	180	118	128
	69.09%	30.91%	56.83%	43.17%	47.97%	52.03%
Is twin	0.03	0.03	0.03	0.03	0.03	0.00**
Is girl	0.50	0.48	0.52	0.50	0.51	0.41
Pregnancy planned	0.83	0.80	0.65	0.58	0.53	0.47
Number of siblings						
None	0.38	0.46***	0.32	0.38	0.31	0.34
One	0.40	0.35**	0.33	0.29	0.29	0.29
Two	0.16	0.14	0.23	0.13***	0.21	0.17
Three	0.04	0.04	0.05	0.12***	0.08	0.11
Four or more	0.02	0.01	0.07	0.08	0.11	0.09
Youngest sibling under 5	0.51	0.40***	0.49	0.38***	0.47	0.48
Youngest sibling under 2	0.13	0.11	0.14	0.13	0.16	0.17
Child's subjective health						
Excellent	0.64	0.65	0.60	0.53	0.52	0.64**
Very good	0.25	0.25	0.25	0.32	0.33	0.26
Good	0.08	0.07	0.10	0.11	0.15	0.07**
Fair or poor	0.03	0.03	0.05	0.04	0.00	0.03*
Child has developmental problem	0.12	0.12	0.09	0.11	0.09	0.09
HH type = Extended family	0.10	0.12	0.21	0.32**	0.50	0.50
Household income						
Less than 20K	0.00	0.01	0.03	0.02	0.02	0.05
20-30K	0.02	0.02	0.02	0.03	0.05	0.05
30-50K	0.06	0.06	0.13	0.16	0.14	0.20
50-70K	0.13	0.11	0.21	0.18	0.17	0.23
70-100K	0.24	0.25	0.27	0.29	0.31	0.16**
100-150K	0.31	0.29	0.26	0.24	0.19	0.22
More than 150K	0.24	0.25	0.09	0.08	0.11	0.09
Mother is married	0.78	0.71***	0.62	0.51**	0.63	0.73*
Mother is cohabiting	1.00	0.99*	0.98	0.97	0.99	0.97
Mother's age						
<25 years	0.07	0.08	0.18	0.19	0.13	0.19
25-29 years	0.19	0.19	0.28	0.32	0.40	0.25**
30-34 years	0.39	0.37	0.36	0.28	0.27	0.35
35-39 years	0.30	0.30	0.16	0.17	0.17	0.19
40+ years	0.05	0.06	0.02	0.04	0.03	0.02
			•		•	

#### Table A.8 - Heterogeneity analysis by ethnicity - outcome: high involvement (9m)

Mother born in NZ	0.76	0.75	0.88	0.89	0.59	0.55
Mother's prioritised ethnicity						
NZ European	0.86	0.83**	0.57	0.46**	0.24	0.16
Māori	0.07	0.09	0.37	0.46*	0.17	0.16
Pasifika	0.02	0.02	0.04	0.06	0.55	0.64
Asian	0.04	0.04	0.02	0.02	0.03	0.03
MELAA	0.01	0.03***	0.00	0.00	0.01	0.01
Mother's education						
Sec school/NCEA 1-4	0.15	0.18	0.24	0.24	0.32	0.33
Diploma/Trade cert/NCEA 5-6	0.26	0.23	0.28	0.32	0.28	0.36
Bachelors degree	0.31	0.30	0.26	0.26	0.21	0.12**
Higher degree	0.26	0.27	0.14	0.14	0.12	0.13
Mother's occupation						
Does not apply / not employed	0.33	0.22***	0.41	0.40	0.40	0.48
Managers	0.08	0.13***	0.04	0.08**	0.07	0.05
Professionals	0.38	0.41	0.30	0.27	0.20	0.21
Technicians and trades workers	0.03	0.02	0.02	0.01	0.01	0.02
Community and personal service workers	0.03	0.04	0.05	0.05	0.05	0.07
Clerical and administrative workers	0.12	0.12	0.12	0.13	0.16	0.10
Sales workers	0.03	0.04	0.02	0.04	0.06	0.03
Labourers	0.01	0.01	0.04	0.02	0.03	0.04
Partner is biological father	0.99	0.99*	1.00	0.99	1.00	0.99
Father's age						
<25 years	0.04	0.05	0.09	0.12	0.09	0.15
25-29 years	0.15	0.16	0.26	0.28	0.31	0.23
30-34 years	0.33	0.32	0.36	0.31	0.28	0.23
35-39 years	0.32	0.28**	0.24	0.16*	0.20	0.26
40+ years	0.16	0.20**	0.05	0.13***	0.12	0.14
Father has criminal history	0.01	0.01	0.05	0.04	0.01	0.04
Father born in NZ	0.81	0.75***	0.96	0.96	0.53	0.47
Father's education						
No sec school qualification	0.03	0.04	0.09	0.16*	0.09	0.15
Sec school/NCEA 1-4	0.15	0.18**	0.21	0.24	0.33	0.36
Diploma/Trade cert/NCEA 5-6	0.37	0.38	0.43	0.43	0.38	0.37
Bachelors degree	0.24	0.18***	0.16	0.12	0.09	0.08
Higher degree	0.22	0.23	0.11	0.06	0.10	0.05
Father's occupation						
Does not apply / not employed	0.14	0.16	0.22	0.26	0.19	0.25
Managers	0.24	0.17***	0.14	0.10	0.08	0.05
Professionals	0.33	0.32	0.22	0.16	0.20	0.08***
Technicians and trades workers	0.14	0.18**	0.17	0.16	0.11	0.17
Community and personal service workers	0.02	0.04***	0.04	0.04	0.03	0.05
Clerical and administrative workers	0.04	0.03	0.06	0.04	0.10	0.03**

Sales workers	0.04	0.04	0.02	0.06**	0.04	0.05
Machinery operators and drivers	0.02	0.03	0.06	0.08	0.14	0.20
Labourers	0.04	0.03	0.08	0.09	0.10	0.13
Income difference in K\$	4.02	2.05***	2.67	1.23***	1.65	0.89**
Father works in shifts	0.17	0.21**	0.24	0.23	0.26	0.29
Father is self-employed	0.23	0.23	0.15	0.12	0.10	0.08
Farther is fulltime employed	0.92	0.77***	0.86	0.68***	0.85	0.73**
Father works more than 40 hours per week	0.70	0.52***	0.67	0.48***	0.51	0.44
Father has any mental health diagnosis	0.13	0.14	0.10	0.12	0.06	0.03
Father has any physical health diagnosis	0.33	0.32	0.40	0.43	0.25	0.30
In good (subjective) health	0.91	0.91	0.84	0.78	0.73	0.83*
Big five personality traits						
Extraversion	3.55	3.51	3.60	3.54	3.49	3.44
Agreeableness	3.93	4.01***	4.01	4.00	4.14	4.15
Conscientiousness	3.98	4.00	3.93	3.91	3.92	4.00
Neuroticism	2.21	2.20	2.17	2.23	2.13	2.28*
Openness	3.97	4.02*	4.05	4.00	3.92	3.87
Perceived stress scale	-0.12	-0.16	-0.08	0.13**	0.28	0.44
High importance of work-life balance (> median)						
(9m)	0.47	0.42*	0.55	0.52	0.60	0.55
Thinks he has influence on his child's						
development (9m)	0.78	0.91***	0.80	0.91***	0.84	0.94**
Being a parent is important part of identity (2y)	0.97	0.97	0.99	0.98	0.99	1.00
Feels confident parenting (9m)	0.84	0.91***	0.82	0.95***	0.88	0.97***
Thinks he is a better-than-average parent (2y)	0.78	0.84***	0.76	0.80	0.81	0.86
Satisfaction with being parent (9m)	5.04	5.30***	5.22	5.48***	5.39	5.64***
Relationship quality	0.01	0.17***	-0.18	-0.16	-0.22	-0.18
Average helpfulness of family	-0.08	0.05***	0.21	0.31	0.36	0.43

*Notes:* Significant starts in columns 2,4 and 6 refer to the significance levels of a t-test for mean equality between fathers with low and high involvement with \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

Table A 9 – Heterogeneity	analysis hy et	hnicity - outcome	nuality of care
Table A.J Therefore the first	analysis by Cu	milling outcome.	quality of care

	NZ Euro	NZ European		Māori		Pasifika	
	Quality	r care	Quality	/ care	Qualit	ty care	
	< Mean	>Mean	< Mean	>Mean	< Mean	>Mean	
Observations	961	1,051	194	223	117	129	
	47.76%	52.24%	46.52%	53.48%	47.56%	52.44%	
Is twin	0.04	0.02	0.03	0.04	0.00	0.03*	
ls girl	0.50	0.49	0.56	0.47*	0.50	0.43	
Pregnancy planned	0.78	0.85***	0.57	0.66*	0.46	0.53	
Number of siblings							
None	0.27	0.53***	0.23	0.45*	0.21	0.42***	
One	0.45	0.33***	0.33	0.30	0.29	0.29	
Тwo	0.21	0.10***	0.24	0.14**	0.28	0.11***	
Three	0.05	0.03**	0.10	0.05*	0.10	0.09	
Four or more	0.03	0.01***	0.10	0.04**	0.11	0.09	
Youngest sibling under 5	0.61	0.36***	0.56	0.34***	0.56	0.40**	
Youngest sibling under 2	0.16	0.09***	0.17	0.10**	0.21	0.13	
Child's subjective health							
Excellent	0.61	0.68***	0.57	0.58	0.55	0.61	
Very good	0.26	0.24	0.28	0.27	0.28	0.30	
Good	0.09	0.06***	0.12	0.09	0.16	0.06**	
Fair or poor	0.04	0.02***	0.03	0.06	0.01	0.02	
Child has developmental problem	0.12	0.11	0.10	0.10	0.10	0.08	
HH type = Extended family	0.11	0.10	0.26	0.26	0.53	0.47	
Household income							
Less than 20K	0.01	0.00	0.03	0.02	0.06	0.02*	
20-30K	0.02	0.01	0.03	0.02	0.06	0.04	
30-50K	0.07	0.06*	0.13	0.14	0.21	0.14	
50-70K	0.15	0.11***	0.23	0.17	0.24	0.17	
70-100K	0.26	0.22*	0.28	0.28	0.21	0.26	
100-150K	0.28	0.32**	0.24	0.26	0.16	0.25*	
More than 150K	0.21	0.27***	0.07	0.10	0.07	0.13	
Mother is married	0.75	0.77	0.55	0.59	0.69	0.67	
Mother is cohabiting	1.00	1.00	0.99	0.96*	0.97	0.99	
Mother's age							
<25 years	0.07	0.07	0.17	0.19	0.21	0.12*	
25-29 years	0.18	0.20	0.27	0.32	0.35	0.29	
30-34 years	0.35	0.42***	0.35	0.30	0.27	0.35	
35-39 years	0.34	0.26***	0.18	0.16	0.15	0.20	
40+ years	0.06	0.05	0.03	0.03	0.02	0.04	
Mother born in NZ	0.75	0.77	0.89	0.89	0.47	0.67***	
Mother's prioritised ethnicity							
NZ European	0.83	0.87**	0.51	0.53	0.13	0.26*	

Māori	0.09	0.06***	0.42	0.40	0.10	0.22*
Pasifika	0.01	0.02	0.05	0.05	0.72	0.49***
Asian	0.04	0.04	0.02	0.02	0.05	0.02
MELAA	0.02	0.01*	0.00	0.00	0.00	0.02
Mother's education						
Sec school/NCEA 1-4	0.17	0.15	0.22	0.26	0.37	0.29
Diploma/Trade cert/NCEA 5-6	0.28	0.22***	0.34	0.26	0.32	0.32
Bachelors degree	0.30	0.31	0.26	0.26	0.12	0.20*
Higher degree	0.22	0.30***	0.11	0.17*	0.09	0.16
Mother's occupation						
Does not apply / not employed	0.35	0.24***	0.44	0.38	0.49	0.40
Managers	0.06	0.12***	0.04	0.07	0.03	0.08
Professionals	0.34	0.43***	0.28	0.29	0.21	0.21
Technicians and trades workers	0.03	0.03	0.02	0.01	0.01	0.02
Community and personal service workers	0.04	0.03	0.06	0.04	0.08	0.05
Clerical and administrative workers	0.13	0.10**	0.11	0.13	0.08	0.18**
Sales workers	0.03	0.03	0.01	0.04**	0.04	0.05
Machinery operators and drivers	0.00	0.00	0.00	0.00	0.02	0.00
Labourers	0.01	0.01	0.03	0.03	0.05	0.02
Partner is biological father	1.00	0.99	0.99	1.00	1.00	0.99
Father's age						
<25 years	0.04	0.04	0.08	0.13	0.15	0.10
25-29 years	0.14	0.17**	0.27	0.26	0.26	0.26
30-34 years	0.29	0.36***	0.35	0.34	0.25	0.26
35-39 years	0.34	0.28***	0.21	0.20	0.22	0.24
40+ years	0.19	0.15**	0.10	0.08	0.12	0.14
Father has criminal history	0.01	0.01*	0.07	0.02**	0.01	0.04
Father born in NZ	0.80	0.78	0.97	0.95	0.43	0.57**
Father's education						
No sec school qualification	0.05	0.02***	0.16	0.09**	0.10	0.14
Sec school/NCEA 1-4	0.15	0.17	0.21	0.24	0.40	0.29*
Diploma/Trade cert/NCEA 5-6	0.40	0.34***	0.44	0.43	0.40	0.35
Bachelors degree	0.21	0.23	0.11	0.16	0.04	0.12**
Higher degree	0.19	0.24***	0.08	0.09	0.05	0.09
Father's occupation						
Does not apply / not employed	0.13	0.15	0.23	0.25	0.21	0.23
Managers	0.26	0.18***	0.10	0.14	0.03	0.09*
Professionals	0.29	0.36***	0.18	0.21	0.15	0.12
Technicians and trades workers	0.15	0.15	0.18	0.15	0.18	0.11
Community and personal service workers	0.02	0.03**	0.03	0.05	0.02	0.07**
Clerical and administrative workers	0.04	0.03	0.07	0.03*	0.04	0.09*
Sales workers	0.04	0.04	0.04	0.03	0.04	0.05
Machinery operators and drivers	0.03	0.02***	0.10	0.05*	0.16	0.17
		I		I		

Labourers	0.04	0.03	0.08	0.09	0.15	0.08*
Income difference in K\$	4.34	2.56***	2.35	1.79	1.90	0.67***
Father works in shifts	0.19	0.18	0.24	0.24	0.28	0.27
Father is self-employed	0.22	0.24	0.15	0.12	0.05	0.12**
Farther is fulltime employed	0.91	0.84***	0.78	0.78	0.81	0.77
Father works more than 40 hours per week	0.69	0.60***	0.56	0.61	0.46	0.48
Father has any mental health diagnosis	0.12	0.14	0.09	0.13	0.03	0.06
Father has any physical health diagnosis	0.32	0.34	0.40	0.43	0.30	0.26
In good (subjective) health	0.90	0.92**	0.78	0.83	0.76	0.80
Big five personality traits						
Extraversion	3.48	3.59***	3.52	3.63*	3.40	3.53*
Agreeableness	3.92	3.99***	3.97	4.04	4.10	4.19
Conscientiousness	3.99	3.99	3.88	3.96	3.90	4.02
Neuroticism	2.21	2.21	2.23	2.17	2.25	2.16
Openness	3.90	4.06***	3.98	4.07	3.82	3.96**
Perceived stress scale	-0.11	-0.15	0.03	-0.00	0.42	0.31
High importance of work-life balance (> median)						
(9m)	0.48	0.43**	0.56	0.51	0.63	0.53*
Thinks he has influence on his child's						
development (9m)	0.76	0.88***	0.79	0.90***	0.84	0.94**
Being a parent is important part of identity (2y)	0.96	0.98**	0.99	0.98	0.99	1.00
Feels confident parenting (9m)	0.84	0.87*	0.85	0.90*	0.91	0.95
Thinks he is a better-than-average parent (2y)	0.75	0.84***	0.78	0.78	0.79	0.88*
Satisfaction with being parent (9m)	4.97	5.26***	5.23	5.41***	5.46	5.57
Relationship quality	-0.08	0.19***	-0.32	-0.05***	-0.49	0.06***
Average helpfulness of family	-0.11	0.02***	0.10	0.39***	0.35	0.45

*Notes:* Significant starts in columns 2,4 and 6 refer to the significance levels of a t-test for mean equality between fathers with low and high involvement with \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

#### Table A.10 - Heterogeneity analysis by ethnicity - outcome: leave taking

	<u>NZ Eur</u>	NZ European		Māori		<u>Pasifika</u>	
	Parenta	al leave	Parent	al leave	Parent	al leave	
	≤ 2 weeks	> 2 weeks	≤ 2 weeks	> 2 weeks	≤ 2 weeks	>2 weeks	
Observations	961	1,051	194	223	117	129	
	47.76%	52.24%	46.52%	53.48%	47.56%	52.44%	
ls twin	0.02	0.05***	0.02	0.09**	0.01	0.06	
Is girl	0.49	0.48	0.54	0.49	0.41	0.54	
Pregnancy planned	0.82	0.85	0.64	0.74	0.53	0.57	
Number of siblings							
None	0.38	0.44**	0.32	0.46**	0.32	0.29	
One	0.39	0.40	0.34	0.32	0.26	0.29	
Тwo	0.17	0.11***	0.20	0.15	0.20	0.26	
Three	0.04	0.04	0.07	0.03	0.09	0.09	
Four or more	0.02	0.01	0.07	0.03	0.13	0.09	
Youngest sibling under 5	0.50	0.46	0.45	0.35	0.49	0.43	
Youngest sibling under 2	0.12	0.13	0.13	0.09	0.16	0.17	
Child's subjective health							
Excellent	0.65	0.63	0.57	0.57	0.55	0.60	
Very good	0.24	0.27	0.27	0.34	0.32	0.29	
Good	0.07	0.09	0.11	0.06	0.12	0.11	
Fair or poor	0.04	0.02*	0.06	0.03	0.01	0.00	
Child has developmental problem	0.11	0.13	0.11	0.06	0.07	0.14	
HH type = Extended family Household income	0.10	0.10	0.26	0.17	0.51	0.43	
Less than 20K	0.00	0.00	0.02	0.00	0.03	0.00	
20-30K	0.02	0.01	0.01	0.00	0.05	0.06	
30-50K	0.06	0.03**	0.11	0.05	0.16	0.11	
50-70K	0.12	0.13	0.22	0.15	0.22	0.14	
70-100K	0.24	0.21	0.29	0.25	0.18	0.37**	
100-150K	0.30	0.36**	0.26	0.43***	0.26	0.17	
More than 150K	0.26	0.26	0.10	0.12	0.10	0.14	
Mother is married	0.78	0.74	0.58	0.65	0.72	0.66	
Mother is cohabiting	1.00	0.99	0.97	1.00	0.99	0.94**	
Mother's age							
<25 years	0.06	0.05	0.14	0.17	0.12	0.11	
25-29 years	0.20	0.17	0.32	0.20*	0.35	0.26	
30-34 years	0.40	0.36	0.33	0.40	0.29	0.43	
35-39 years	0.29	0.35*	0.18	0.18	0.22	0.11	
40+ years	0.05	0.07*	0.02	0.05	0.02	0.09**	
Mother born in NZ	0.75	0.73	0.88	0.88	0.54	0.66	
Mother's prioritised ethnicity							
			I		I		

	NZ European	0.85	0.87	0.54	0.69**	0.17	0.29
	Māori	0.08	0.05**	0.38	0.29	0.18	0.20
	Pasifika	0.02	0.02	0.06	0.00**	0.62	0.49
	Asian	0.03	0.05*	0.02	0.02	0.03	0.03
	MELAA	0.02	0.01	0.00	0.00	0.00	0.00
N	lother's education						
	Sec school/NCEA 1-4	0.16	0.15	0.23	0.26	0.33	0.29
	Diploma/Trade cert/NCEA 5-6	0.26	0.24	0.30	0.22	0.33	0.29
	Bachelors degree	0.30	0.32	0.28	0.31	0.16	0.26
	Higher degree	0.26	0.28	0.14	0.15	0.12	0.14
N	lother's occupation						
	Does not apply / not employed	0.30	0.28	0.36	0.31	0.46	0.34
	Managers	0.10	0.10	0.07	0.08	0.05	0.11
	Professionals	0.38	0.42	0.30	0.40	0.22	0.20
	Technicians and trades workers	0.03	0.02	0.02	0.00	0.01	0.00
	Community and personal service workers	0.03	0.04	0.07	0.00**	0.05	0.06
	Clerical and administrative workers	0.12	0.10	0.12	0.14	0.10	0.26**
	Sales workers	0.03	0.03	0.02	0.05	0.05	0.03
	Machinery operators and drivers	0.00	0.00	0.00	0.00	0.01	0.00
	Labourers	0.01	0.01	0.04	0.03	0.04	0.00
Pa	artner is biological father	1.00	0.98***	1.00	1.00	1.00	1.00
Fa	ather's age						
	<25 years	0.03	0.03	0.07	0.03	0.10	0.09
	25-29 years	0.15	0.13	0.26	0.23	0.28	0.26
	30-34 years	0.33	0.32	0.32	0.46**	0.25	0.29
	35-39 years	0.32	0.31	0.25	0.22	0.26	0.20
	40+ years	0.17	0.20*	0.11	0.06	0.12	0.17
Fa	ather has criminal history	0.01	0.00	0.05	0.00*	0.02	0.03
Fa	ather born in NZ	0.79	0.78	0.97	0.91**	0.47	0.69**
Fa	ather's education						
	No sec school qualification	0.04	0.01**	0.10	0.09	0.16	0.03**
	Sec school/NCEA 1-4	0.17	0.12**	0.25	0.17	0.34	0.31
	Diploma/Trade cert/NCEA 5-6	0.36	0.39	0.42	0.46	0.36	0.40
	Bachelors degree	0.23	0.20	0.15	0.18	0.06	0.20***
	Higher degree	0.21	0.27**	0.08	0.09	0.08	0.06
Fa	ather's occupation						
	Managers	0.28	0.16***	0.16	0.18	0.08	0.09
	Professionals	0.36	0.44***	0.25	0.26	0.17	0.23
	Technicians and trades workers	0.18	0.16	0.21	0.22	0.18	0.17
	Community and personal service workers	0.02	0.08***	0.06	0.05	0.05	0.09
	Clerical and administrative workers	0.04	0.05	0.06	0.11	0.08	0.11
	Sales workers	0.05	0.05	0.05	0.02	0.06	0.06
	Machinery operators and drivers	0.03	0.03	0.09	0.11	0.21	0.17
				•		•	

Labourers	0.04	0.03	0.12	0.06	0.16	0.09
Income difference in K\$	3.71	3.51	2.51	2.94	1.55	1.83
Father works in shifts	0.15	0.26***	0.25	0.34	0.27	0.20
Father is self-employed	0.25	0.18**	0.16	0.12	0.08	0.11
Farther is fulltime employed	0.90	0.83***	0.86	0.88	0.85	0.86
Father works more than 40 hours per week	0.68	0.61**	0.64	0.68	0.55	0.46
Father has any mental health diagnosis	0.12	0.13	0.10	0.09	0.04	0.11*
Father has any physical health diagnosis	0.33	0.31	0.38	0.43	0.28	0.34
In good (subjective) health	0.92	0.90	0.84	0.85	0.80	0.60**
Big five personality traits						
Extraversion	3.55	3.51	3.57	3.54	3.43	3.46
Agreeableness	3.97	3.97	4.04	3.89*	4.15	4.12
Conscientiousness	4.02	3.94**	3.95	3.86	3.99	3.88
Neuroticism	2.21	2.13*	2.19	2.17	2.20	2.28
Openness	3.96	4.00	4.00	4.09	3.84	4.01
Perceived stress scale	-0.15	-0.22	-0.01	-0.09	0.34	0.58
High importance of work-life balance (> median)	0.46	0.44	0.55	0.49	0.58	0.54
(9m)						
Thinks he has influence on his child's	0.01	0.02	0.94	0.00	0.80	0.00
development (9m)	0.81	0.85	0.84	0.00	0.89	0.89
Being a parent is important part of identity (2y)	0.97	0.98	0.99	0.97	0.99	1.00
Feels confident parenting (9m)	0.86	0.88	0.88	0.86	0.93	0.89
Thinks he is a better-than-average parent (2y)	0.80	0.80	0.78	0.82	0.84	0.80
Satisfaction with being parent (9m)	5.12	5.16	5.32	5.22	5.53	5.33*
Relationship quality	0.04	0.15**	-0.16	0.09*	-0.26	0.10
Average helpfulness of family	-0.02	-0.13**	0.29	0.17	0.32	0.40

*Notes:* Significant starts in columns 2,4 and 6 refer to the significance levels of a t-test for mean equality between fathers with low and high involvement with \*\*\* for p<0.01, \*\* for p<0.05 and \* for p<0.1.

### Table A.11 – Items list for CBQ variables

		Negative effect	Low effortful control	Low surgency
1	{name} seems always in a big hurry to get from one place to another {name} gets quite frustrated when prevented from doing something {he/she} wants	x		x(r)
2	When drawing or colouring in a book {name} shows strong concentration	X	x(r)	
4	{name} likes going down high slides or other adventurous activities			x(r)
5	{name} is quite upset by a little cut or bruise	x		
6	{name} prepares for trips and outings by planning things {he/she} will need	~	x(r)	
7	{name} often rushes into new situations			x(r)
, 8	{name} tends to become sad if the family's plans don't work out	x		,(1)
9	{name} likes being sung to	X	x(r)	
10	{name} seems to be at ease with almost any person		X(1)	x(r)
11	[name] is afraid of hurglars or the "hoogie man"	Y		A(1)
12	(name) is alraid of bulgious of the boogle man	A	$\mathbf{x}(\mathbf{r})$	
12			~(1)	×
10	When angry about something, {name} tends to stay upset for ten minutes			^
14	or longer	Х		
	When building or putting something together, becomes very involved in		( )	
15	what {name} is doing		x(r)	
16	{name} likes to go high and fast when pushed on a swing			x(r)
1/	{name} seems to feel depressed when unable to accomplish some tasks	Х	<i>(</i> )	
18	{name} is good at following instructions		x(r)	
19	{name} takes a long time in approaching new situations			Х
20	{name} hardly ever complains when ill with a cold	x(r)		
21 22	{name} likes the sound of words, such as nursery rhymes {name} is sometimes shy even around people {he/she} has known for a long time		x(r)	X
23	{name} is very difficult to soothe when {he/she} has become upset	х		
24	{name} is guickly aware of some new item in the living room		x(r)	
25	{name} is full of energy, even in the evening			x(r)
26	{name} is not afraid of the dark	x(r)		( )
	{name} sometimes becomes absorbed in a picture book and looks at it for			
27	a long time		x(r)	
28	{name} likes rough and rowdy games			x(r)
29	{name} is not very upset at minor cuts or bruises	x(r)		
20	{name} approaches places {he/she} has been told are dangerous slowly or			
30	caulously		x(r)	
31	{name} is slow and unnurried in deciding what to do next {name} gets angry when {he/she} can't find something {he/she} wants to			X
32	play with	х		
33	{name} enjoys gentle rhythmic activities such as rocking or swaying		x(r)	
34	{name} sometimes turns away shyly from new acquaintances			х
	{name} becomes upset when loved relatives or friends are getting ready to			
35	leave	Х		
36	{name} comments when a parent has changed his/her appearance		x(r)	

Source: GUINZ DCW5, own calculations and illustrations.

*Notes:* Items marked with a (r) are reversed.

## Table A.12 – Items list for SDQ variables

		Emotional problems	Peer problems	Hyper- activity	Antisocial behaviour
1	{name} Considerate of other people's feelings at 54 months				X(r)
2	{name} is restless, overactive, cannot stay still for long at 54 month {name} often complains of headaches, stomach-aches, or sickness at 54			Х	
3	months	х			
4	{name} shares readily with other children (treats, toys, pencils, etc.) at 54 months				X(r)
6	{name} rather solitary, prefers to play alone at 54 months		х		
8	{name} worries or often seems worried at 54 months	х			
9	{name} helpful if someone is hurt, upset or feeling ill at 54 months				X(r)
10	{name} is constantly fidgeting or squirming at 54 months			х	
11	{name} has at least one good friend at 54 months		х		
13	{name} is often unhappy, down-hearted or tearful at 54 months	х			
14	{name} is generally liked by other children at 54 months		х		
15	{name} is easily distracted, concentration wanders at 54 months {name} is nervous or clingy in new situations, easily loses confidence at 54			х	
16	months	Х			
17	{name} is kind to younger children				X(r)
19	{name} is picked on or bullied by other children at 54 months {name} often volunteers to help others (parents, teachers, other children)		Х		
20	at 54 months				X(r)
23	{name} gets on better with adults than with other children at 54 months		х		
24	{name} has many fears, is easily scared at 54 months	х			
29	{name} thinks things out before acting at 54 months {name} has good attention span, sees chores or work through to the end			Х	
31	at 54months			Х	

Source: GUINZ DCW5, own calculations and illustrations.

Notes: Items marked with a (r) are reversed



# NEW ZEALAND WORK RESEARCH INSTITUTE

Auckland University of Technology, Auckland, New Zealand work.research@aut.ac.nz | www.workresearch.aut.ac.nz

